RANDOM DYNAMICS ON REAL AND COMPLEX PROJECTIVE SURFACES

SERGE CANTAT AND ROMAIN DUJARDIN

ABSTRACT. We initiate the study of random iteration of automorphisms of real and complex projective surfaces, as well as compact Kähler surfaces, focusing on the classification of stationary measures. We show that, in a number of cases, such stationary measures are invariant, and provide criteria for uniqueness, smoothness and rigidity of invariant probability measures. This involves a variety of tools from complex and algebraic geometry, random products of matrices, non-uniform hyperbolicity, as well as recent results of Brown and Rodriguez Hertz on random iteration of surface diffeomorphisms.

1. INTRODUCTION

1.1. Random dynamical systems. Consider a compact manifold M and a probability measure ν on Diff(M); to simplify the exposition we assume throughout this introduction that the support $\text{Supp}(\nu)$ is finite. The data (M, ν) defines a random dynamical system, obtained by randomly composing independent diffeomorphisms with distribution ν . In this paper, these random dynamical systems are studied from the point of view of *ergodic theory*, that is, we are mostly interested in understanding the *asymptotic distribution* of orbits.

A probability measure μ on M is ν -invariant if $f_*\mu = \mu$ for ν -almost every $f \in \text{Diff}(M)$, and it is ν -stationary if it is invariant on average: $\int f_*\mu d\nu(f) = \mu$. A simple fixed point argument shows that stationary measures always exist. On the other hand, the existence of an invariant measure should hold only under special circumstances, for instance when the group Γ_{ν} generated by $\text{Supp}(\nu)$ is amenable, or has a finite orbit, or preserves an invariant volume form.

According to Breiman's law of large numbers, for every $x \in M$ and $\nu^{\mathbf{N}}$ -almost every $(f_j) \in \text{Diff}(M)^{\mathbf{N}}$, every cluster value of the sequence of empirical measures

(1.1)
$$\frac{1}{n} \sum_{j=0}^{n-1} \delta_{f_j \circ \cdots \circ f_0(x)}$$

is a stationary measure. Thus, a classification of stationary measures gives an essentially complete understanding of the distribution of random orbits as n goes to $+\infty$. Our goal is to combine algebraic and holomorphic dynamics with recent results in random dynamics to study the case when M is a real or complex projective surface and the action is by algebraic diffeomorphisms.

1.2. **Stiffness.** Let us present a few landmark results about stationary measures (for consistency with the rest of the paper, most of the discussion is restricted to real dimension 2).

Let Γ be a subgroup of $\operatorname{GL}_m(\mathbf{C})$. We say that he action of Γ on \mathbf{C}^m is strongly irreducible if the orbit of any subspace $V \subset \mathbf{C}^m$ with $0 < \dim(V) < m$ is infinite; it is **proximal** if there is an element $\gamma \in \Gamma$ with a unique eigenvalue of maximum modulus (the corresponding eigenline provides an attracting fixed point in $\mathbb{P}^{m-1}(\mathbf{C})$). This said, suppose that ν is a finitely supported probability measure on $SL_2(\mathbf{C})$, and consider the action of $SL_2(\mathbf{C})$ on $M = \mathbb{P}^1(\mathbf{C})$. Suppose that the group Γ_{ν} generated by the support of ν is **non-elementary**, that is, Γ_{ν} is proximal and strongly irreducible. Then, *there is a unique* ν -stationary (probability) measure μ on $\mathbb{P}^1(\mathbf{C})$, and this measure is not invariant. This is one instance of a more general result due to Furstenberg [50]. The non-invariance of μ is due to the existence of proximal elements in Γ_{ν} .

Now, let ν be a finitely supported measure on $SL_2(\mathbf{Z})$, and consider the action of $SL_2(\mathbf{Z})$ on the torus $M = \mathbf{R}^2/\mathbf{Z}^2$. In that case, the Haar measure of $\mathbf{R}^2/\mathbf{Z}^2$, as well as the atomic measures equidistributed on finite orbits $\Gamma_{\nu}(x, y)$, for $(x, y) \in \mathbf{Q}^2/\mathbf{Z}^2$, are examples of Γ_{ν} -invariant measures. By using Fourier analysis and additive combinatorics techniques, Bourgain, Furman, Lindenstrauss and Mozes [15] proved that if Γ_{ν} is non-elementary, then every stationary measure μ on $\mathbf{R}^2/\mathbf{Z}^2$ is Γ_{ν} -invariant and is a convex combination of the above mentioned invariant measures. This property of automatic invariance of stationary measures was called **stiffness** (or ν -stiffness) by Furstenberg [51], who conjectured it to hold in this setting. Soon after, Benoist and Quint [7] gave an ergodic theoretic proof of this result and extended it to certain actions of discrete groups on homogeneous spaces. They also derived the following equidistribution result: for every $(x, y) \notin \mathbf{Q}^2/\mathbf{Z}^2$, the random trajectories of (x, y) determined by ν almost surely equidistribute towards the Haar measure on $\mathbf{R}^2/\mathbf{Z}^2$.

Finally, Brown and Rodriguez-Hertz [16], building on the work of Eskin and Mirzakhani [45], managed to recast these measure rigidity results in terms of Pesin theory to obtain a version of the stiffness theorem of [15] for general C^2 diffeomorphisms of compact surfaces. We shall describe their results in due time; for the moment we content ourselves with one illustrative consequence of [16]. Let $\nu = \sum \alpha_j \delta_{f_j}$ be a finitely supported probability measure on $SL_2(\mathbf{Z})$ generating a non-elementary subgroup. Consider perturbations $\{f_{i,\varepsilon}\}$ of the f_i in the group $\text{Diff}_{vol}^2(\mathbf{R}^2/\mathbf{Z}^2)$ of C^2 diffeomorphisms of $\mathbf{R}^2/\mathbf{Z}^2$ preserving the Haar measure. Set $\nu_{\varepsilon} = \sum \alpha_j \delta_{f_{j,\varepsilon}}$. Then, for sufficiently small perturbations, any ν_{ε} -stationary measure on $\mathbf{R}^2/\mathbf{Z}^2$ is invariant and is a combination of the Haar measure and measures supported on finite $\Gamma_{\nu_{\varepsilon}}$ -orbits.

In this paper, we prove a stiffness theorem for groups of algebraic diffeomorphisms of real algebraic surfaces. The work of Brown and Rodriguez-Hertz is our main source of inspiration and a key ingredient for some of our main results.

1.3. Sample results: stiffness, classification, and rigidity. Let X be a smooth complex projective surface, or more generally a compact Kähler surface. Denote by $\operatorname{Aut}(X)$ its group of holomorphic diffeomorphisms, referred to in this paper as **automorphisms**. When $X \subset \mathbb{P}^N(\mathbb{C})$ is defined by polynomial equations with real coefficients, the complex conjugation induces an anti-holomorphic involution $s: X \to X$, whose fixed point set is the real part $X(\mathbb{R})$ of X. We denote by $X_{\mathbb{R}}$ the surface X viewed as an algebraic variety defined over \mathbb{R} , and by $\operatorname{Aut}(X_{\mathbb{R}})$ the group of automorphisms defined over \mathbb{R} ; $\operatorname{Aut}(X_{\mathbb{R}})$ is the subgroup of $\operatorname{Aut}(X)$ centralizing s. When $X(\mathbb{R}) \neq \emptyset$, the elements of $\operatorname{Aut}(X_{\mathbb{R}})$ are the real-analytic diffeomorphisms of $X(\mathbb{R})$ admitting a holomorphic extension to X. Note that in stark contrast with groups of smooth diffeomorphisms, the groups $\operatorname{Aut}(X_{\mathbb{R}})$ and $\operatorname{Aut}(X)$ are typically discrete and at most countable.

The group $\operatorname{Aut}(X)$ acts on the cohomology $H^*(X; \mathbb{Z})$. By definition, a subgroup $\Gamma \subset \operatorname{Aut}(X)$ is **non-elementary** if its image $\Gamma^* \subset \operatorname{GL}(H^*(X; \mathbb{C}))$ contains a non-Abelian free group; equivalently, Γ^* is not virtually Abelian. By Yomdin's theorem, when Γ is non-elementary, there exists a pair $(f, g) \in \Gamma^2$ generating a free group of rank 2 such that the topological entropy of every element in that group is positive (see [27]).

3

1.3.1. Stiffness. As before, if ν is a finitely supported probability measure on Aut(X), we denote by Γ_{ν} the subgroup generated by Supp(ν).

Theorem A. Let $X_{\mathbf{R}}$ be a real projective surface and ν be a finitely supported symmetric probability measure on Aut $(X_{\mathbf{R}})$. If Γ_{ν} preserves an area form on $X(\mathbf{R})$, then every ergodic ν stationary measure μ on $X(\mathbf{R})$ is either invariant or supported on a proper Γ_{ν} -invariant subvariety. In particular if there is no Γ_{ν} -invariant algebraic curve, the random dynamical system (X, ν) is stiff.

This theorem is mostly interesting when Γ_{ν} is non-elementary and we focus on this case in the remainder of this introduction. Stationary measures supported on invariant curves are easily analysed (see §10.4). Moreover, if Γ_{ν} is non-elementary, it is always possible to contract all Γ_{ν} invariant curves, creating a complex analytic surface X_0 with finitely many singularities. Then on $X_0(\mathbf{R})$, stiffness holds unconditionally.

This result applies to many interesting examples, because Abelian, K3, and Enriques surfaces, which concentrate most of the dynamically interesting automorphisms on compact complex surfaces, admit a canonical Aut(X)-invariant 2-form.

1.3.2. Invariant measures. Once stiffness is established, the next step is to classify invariant measures. A **parabolic** automorphism of a compact Kähler surface is an automorphism g such that the norm of $(g^n)^*$ on $H^2(X; \mathbf{R})$ grows quadratically (i.e. like αn^2 for some $\alpha > 0$); such an automorphism automatically preserves a genus 1 fibration on X (see e.g. [26]). When Γ_{ν} contains a parabolic automorphism, Γ_{ν} -invariant measures are classified in [19, 26]. A nice consequence is that for a non-elementary group of $\operatorname{Aut}(X_{\mathbf{R}})$ containing parabolic elements and preserving an area form, any invariant ergodic measure is either atomic, or concentrated on a Γ_{ν} -invariant algebraic curve, or is the restriction of the area form on some open subset of $X(\mathbf{R})$ bounded by a piecewise smooth curve.

Thus, if Γ_{ν} contains a parabolic element, we get a fairly complete answer to the equidistribution problem raised in §1.1. A widely studied example is the family of **Wehler surfaces** that is, smooth surfaces $X \subset \mathbb{P}^1 \times \mathbb{P}^1 \times \mathbb{P}^1$ defined by an equation of degree (2, 2, 2). Then for each $i \in \{1, 2, 3\}$, the projection $\pi_i \colon X \to \mathbb{P}^1 \times \mathbb{P}^1$ which "forgets the variable x_i " has degree 2; thus, there is an involution σ_i of X that permutes the two points in the generic fiber of π_i .

Corollary. Let $X_{\mathbf{R}} \subset \mathbb{P}^1 \times \mathbb{P}^1 \times \mathbb{P}^1$ be a real Wehler surface such that $X(\mathbf{R})$ is non empty. If $X_{\mathbf{R}}$ is generic, then:

- (1) the surface X is a K3 surface and there is a unique (up to choosing an orientation of $X(\mathbf{R})$) algebraic 2-form $\operatorname{vol}_{X_{\mathbf{R}}}$ on $X(\mathbf{R})$ such that $\int_{X(\mathbf{R})} \operatorname{vol}_{X_{\mathbf{R}}} = 1$;
- (2) the group $\operatorname{Aut}(X_{\mathbf{R}})$ is generated by the three involutions σ_i and coincides with $\operatorname{Aut}(X)$; furthermore it preserves the probability measure defined by $\operatorname{vol}_{X_{\mathbf{R}}}$;
- (3) if ν is finitely supported and Γ_{ν} has finite index in Aut $(X_{\mathbf{R}})$ then $(X(\mathbf{R}), \nu)$ is stiff; moreover the only ν -stationary measures on $X(\mathbf{R})$ are convex combinations of the probability measures defined by vol_{X_R} on the connected components of $X(\mathbf{R})$.

Here by generic we mean that the equation of X belongs to the complement of at most countably many hypersurfaces in the set of polynomial equations of degree (2, 2, 2) (see §3.2 for details). This result follows from Theorem A, Proposition 3.7, Corollary B of [26], and the generic non-existence of finite orbits established in [25]. If we do not assume X to be generic but assume only that X does not contain any fiber of the three projections π_i , then the set of stationary measures supported in $X(\mathbf{R})$ is a finite dimensional simplex (see [25]); the equidistribution problem is further studied in [23].

The techniques of [19, 26] do not apply in the absence of parabolic automorphisms. In this context, we establish the following rigidity result.

Theorem B. Let $X_{\mathbf{R}}$ be a real projective surface. Let Γ be a non-elementary subgroup of $\operatorname{Aut}(X_{\mathbf{R}})$. If μ is a Γ -invariant probability measure on $X(\mathbf{R})$ and if μ is ergodic and of positive entropy for some $f \in \Gamma$, then μ is absolutely continuous with respect to any area form on $X(\mathbf{R})$.

In particular if Γ is a group of area preserving automorphisms, then up to normalization μ will be the restriction of the area form on some Γ -invariant set.

1.4. Some ingredients of the proofs. The proofs of Theorems A and B rely on the deep results of Brown and Rodriguez-Hertz [16]. To be more precise, recall that an ergodic stationary measure μ on X admits two Lyapunov exponents $\lambda^+(\mu) \ge \lambda^-(\mu)$, and that μ is called **hyperbolic** if $\lambda^+(\mu) > 0 > \lambda^-(\mu)$. In this case the (random) Oseledets theorem shows that for μ -almost every x and $\nu^{\mathbf{N}}$ -almost every $\omega = (f_j)_{j \in \mathbf{N}}$ in $\operatorname{Aut}(X)^{\mathbf{N}}$, there exists a stable direction $E^s_{\omega}(x) \subset T_x X_{\mathbf{R}}$. In [16], stiffness is established under the condition that $E^s_{\omega}(x) \subset T_x X_{\mathbf{R}}$ depends non-trivially on the random itinerary $\omega = (f_j)_{j \in \mathbf{N}}$, or equivalently that stable directions do not induce a measurable Γ_{ν} -invariant line field. One of our main contributions is to take care of this possibility in our setting: for this we study the dynamics on the *complex* surface X.

Theorem C. Let X be a complex projective surface and ν be a finitely supported probability measure on Aut(X). If Γ_{ν} is non-elementary, then any hyperbolic ergodic ν -stationary measure μ on X satisfies the following alternative:

- (a) either μ is invariant, and its fiber entropy $h_{\mu}(X; \nu)$ vanishes;
- (b) or μ is supported on a Γ_{ν} -invariant algebraic curve;
- (c) or the field of Oseledets stable directions of μ is not Γ_{ν} -invariant; in other words, it genuinely depends on the itinerary $\omega = (f_j)_{j \ge 0} \in \operatorname{Aut}(X)^{\mathbb{N}}$.

As opposed to Theorems A and B, this result applies to the dynamics on the *complex* manifold X, without assuming the existence of an invariant volume form or an invariant real structure. When μ is not invariant, nor supported by a proper Zariski closed subset, Assertion (c) precisely says that the condition on stable directions used in [16] is satisfied. This is our key input towards Theorems A and B. The arguments leading to Theorem C involve an interesting blend of Hodge theory, pluripotential analysis, and Pesin theory. They rely on the following well-known principle in higher dimensional holomorphic dynamics. If μ is ergodic and hyperbolic, almost every point (ω, x) provides a stable manifold $W^s_{\omega}(x)$ biholomorphic to C. Then, according to a construction going back to Ahlfors and Nevanlinna, to any entire curve $\phi : \mathbf{C} \to X$ is associated a (family of) closed positive (1, 1)-current(s) describing the asymptotic distribution of $\phi(\mathbf{C})$ in X, hence also a (family of) cohomology class(es) in $H^2(X, \mathbf{R})$. These classes relate the stable manifolds of μ to the action of Γ_{ν} on $H^2(X; \mathbf{R})$, which itself can be analyzed by combining complex algebraic geometry with Furstenberg's theory of random products of matrices.

Theorem D. Let X be a complex projective surface. Let ν be a finitely supported probability measure on Aut(X) such that Γ_{ν} is non-elementary. Let κ_0 be a fixed Kähler form on X.

(1) If κ is any Kähler form on X, then for $\nu^{\mathbf{N}}$ -almost every $\omega := (f_j)_{j \ge 0} \in \operatorname{Aut}(X)^{\mathbf{N}}$ the limit

$$T_{\omega}^{s} := \lim_{n \to +\infty} \frac{1}{\int_{X} \kappa_{0} \wedge (f_{n} \circ \dots \circ f_{0})^{*} \kappa} (f_{n} \circ \dots \circ f_{0})^{*} \kappa$$

exists as a closed positive (1,1)-current. Moreover this current T^s_{ω} does not depend on κ and has Hölder continuous potentials.

(2) If the ν -stationary measure μ is ergodic, hyperbolic (or more generally if $\lambda^{-}(\mu) < 0 \leq \lambda^{+}(\mu)$) and not supported on a Γ_{ν} -invariant proper Zariski closed set, then for μ -almost every x and $\nu^{\mathbf{N}}$ -almost every ω , the only Ahlfors-Nevanlinna current of mass 1 (with respect to κ_{0}) associated to the stable manifold $W_{\omega}^{s}(x)$ coincides with T_{ω}^{s} .

One might consider that the right setting for such a statement would be that of a compact Kähler surface. We actually show in the companion paper [27] that any compact surface supporting a non-elementary group of automorphisms is projective. The algebraicity of X is, in fact, a crucial technical ingredient in the proof of assertion (2), because we use techniques of laminar currents which are available only on projective surfaces. Theorem D enters the proof of Theorem C as follows: since Γ_{ν} is non-elementary, Furstenberg's description of the random action on $H^2(X, \mathbf{R})$ implies that the cohomology class $[T^s_{\omega}]$ depends non-trivially on ω ; therefore for μ -almost every x, $W^s_{\omega}(x)$ also depends non-trivially on ω . Then, taking advantage of the complex structure again, we show in Section 9, that $E^s_{\omega}(x)$ depends non-trivially on ω as well.

Remark 1.1. Beyond finitely supported measures, Theorem A, B, C, and D hold under optimal moment conditions on ν (this adds several technicalities, notably in Sections 5 and 6).

1.5. Organization of the article. Let X be a compact Kähler surface and ν be a probability measure on Aut(X).

– In Section 2 we describe the action of $\operatorname{Aut}(X)$ on $H^*(X; \mathbb{Z})$, in particular on $H^{1,1}(X; \mathbb{R})$. The Hodge index theorem endows it with a Minkowski structure, which is essential in our understanding of the dynamics of Γ_{ν} on the cohomology. This section 2 prepares the ground for the analysis of random products of matrices done in Section 5 (and it is also used in [26, 25]). A delicate point to keep in mind is that the action of a non-elementary subgroup of $\operatorname{Aut}(X)$ on $H^{1,1}(X; \mathbb{R})$ may be reducible.

- In Section 3 we briefly recall some constraints on X coming from the Enriques-Kodaira classification of surfaces, and gives some details on the example of Wehler surfaces.

– After a short Section 4 introducing the vocabulary of random products of diffeomorphisms, Furstenberg's theory of random products of matrices is applied in Section 5 to the study of the action on $H^{1,1}(X; \mathbf{R})$. This, combined with the theory of closed positive currents, leads to the proof of the first assertion of Theorem D in Section 6 (see Corollary 6.12 and Theorems 6.14 and 6.16). The continuity of the potentials of the currents T^s_{ω} , which plays a key role in Section 8, relies on a recent result of Gouëzel and Karlsson [53].

– Pesin theory enters into play in Section 7, in which the basics of the smooth ergodic theory of random dynamical systems are described in some detail for complex surfaces. This is used in Section 8 to connect the stable manifolds to the currents T^s_{ω} , using techniques of laminar currents (Theorem 8.2 gives the second part of Theorem D).

– Theorem C is proven in Section 9 by combining ideas of [16] with Theorem D and an elementary fact from local complex geometry inspired by a lemma from [5]. – Theorem A is finally established in Section 10. When Γ_{ν} is non-elementary (Theorem 10.7) it follows rather directly from [16], Theorem C, and the *invariance principle* of Crauel [37] and Avila-Viana [1]. Elementary groups are handled separately by using the classification of automorphism groups of compact Kähler surfaces (see Section 10.3); note that the symmetry assumption on ν is used only in the elementary case.

- Theorem B is established in Section 11, in a slightly more precise form (see Theorem 11.1).

1.6. Further comments.

- This article is part of a series of papers dedicated to the dynamics of groups of automorphisms of compact Kähler surfaces. In [27] we discuss further examples and sharpen the classification of surfaces admitting non-elementary groups of automorphisms. The article [26] classifies invariant measures in presence of parabolic elements. In [23], we study uniform expansion for random complex dynamics and apply it to equidistribution. In [25], finite orbits are analyzed with tools from algebraic and arithmetic dynamics. Note that some results originally contained in the preprint version of this paper are now in other papers of the series.

- After the first version of this paper and [25] were released, Filip and Tosatti [47] gave an alternate approach of some of the results of Section 6.

- In Theorem A, one may wonder how the invariant measure μ relates to the dynamics of individual elements of Γ_{ν} , in particular if it might coincide with the maximal entropy measure μ_f of some loxodromic element f of Γ_{ν} . For simplicity, assume that X is a real Wehler surface and Γ_{ν} has finite index in Aut $(X_{\mathbf{R}})$ (see § 1.3.2). Then, according to [25, Thm. 5.12], Γ_{ν} contains a loxodromic element h with $\mu_h \neq \mu$. Moreover, if $X(\mathbf{R})$ is connected, the coincidence $\mu = \mu_f$ for some $f \in \Gamma_{\nu}$ is equivalent to the existence of a loxodromic element $f \in Aut(X_{\mathbf{R}})$ such that μ_f is the canonical area form on $X(\mathbf{R})$. We conjecture that such an example does not exist. This is reminiscent of, but different from, the Kummer rigidity results of [28, 48] (see [22, §3.5.2]).

– One may wonder what remains of our results in the real-analytic category. The proofs of Theorems D and C rely on global complex geometric arguments (via Ahlfors-Nevanlinna currents and the Hodge index theorem) to show that stable manifolds depend on random itineraries; in particular Zariski dense (complex) stable and unstable manifolds always admit a transverse intersection in $X(\mathbf{C})$. Such a global geometric argument does not carry over to the real-analytic setting; indeed, there are real analytic diffeomorphisms of closed surfaces with two saddle fixed points xand x' such that their stable and unstable manifolds are Zariski dense but $W_f^s(x) \cap W_f^u(x') = \emptyset$. Theorems D and C also rely on local properties of complex analytic disks to go from stable directions to stable manifolds (see § 9). While some of the results of § 9.2 might persist in the real-analytic category, the key Lemma 9.7 does not (see Remark 9.8).

– Some of our techniques should be transposable to automorphism groups of certain affine surfaces, for example polynomial automorphisms of \mathbb{C}^2 , a main issue in this case being to deal with the lack of compactness. Another example is provided by the character variety of representations of the free group $F_2 = \langle a, b | \emptyset \rangle$ into $SL_2(\mathbb{C})$, with a fixed trace of the commutator $aba^{-1}b^{-1}$; this variety is a surface, and the outer automorphism group $GL_2(\mathbb{Z})$ of F_2 acts by automorphisms on it (see [20, 52, 36] for instance). As shown by Rebelo and Roeder [75], several dynamical regimes coexist on the complex surface, which presumably makes a classification of stationary measure quite elusive. However, looking at the real part (corresponding to representations in the compact group SU_2), we expect stiffness to hold.

7

1.7. **Conventions.** Throughout the paper C stands for a "constant" which may change from line to line, independently of some asymptotic quantity that should be clear from the context (typically an integer n corresponding to the number of iterations of a dynamical system). We write $a \leq b$ if $a \leq Cb$ and $a \approx b$ if $a \leq b \leq a$. Complex manifolds are considered to be connected, so from now on "complex manifold" stands for "connected complex manifold". For a random dynamical system on a disconnected complex manifold, there is a finite index sugbroup Γ' of Γ_{ν} fixing each connected component, and an induced measure ν' on Γ' with properties qualitatively similar to those of ν (see §10.2), so the problem is reduced to the connected case.

Acknowledgments. We are grateful to Sébastien Gouëzel, François Ledrappier, and François Maucourant for interesting discussions and insightful comments. We thank the referees for constructive suggestions and criticisms. The first named author was partially supported by the French Academy of Sciences (Del Duca foundation) and the European Research Council (ERC) under the European Union's Horizon 2020 programme (grant agreement No 101053021 - GOAT), and the second named author by a grant from the Institut Universitaire de France.

2. HODGE INDEX THEOREM AND MINKOWSKI SPACES

In this section we define the notion of a non-elementary group of automorphisms of a compact Kähler surface X. We study the action of such a group on the cohomology of X, and in particular the question of (ir)reducibilty. We work in the Kähler setting because these results are eventually useful to prove that a compact Kähler surface carrying a non-elementary action must be projective (see [27], which also includes a discussion of the non-Kähler case).

2.1. Cohomology.

2.1.1. Hodge decomposition. Denote by $H^*(X; R)$ the cohomology of X with coefficients in the ring R; we shall use $R = \mathbb{Z}$, \mathbb{Q} , \mathbb{R} or \mathbb{C} . The group $\operatorname{Aut}(X)$ acts on $H^*(X; \mathbb{C})$, preserving the image of $H^*(X; \mathbb{Z})$; $\operatorname{Aut}(X)^*$ will denote the image of $\operatorname{Aut}(X)$ in $\operatorname{GL}(H^2(X; \mathbb{C}))$. The Hodge decomposition

(2.1)
$$H^{k}(X; \mathbf{C}) = \bigoplus_{p+q=k} H^{p,q}(X; \mathbf{C})$$

is Aut(X)-invariant. On $H^{0,0}(X; \mathbb{C})$ and $H^{2,2}(X; \mathbb{C})$, Aut(X) acts trivially. Throughout the paper we denote by $[\alpha]$ the cohomology class of a closed differential form (or current) α .

The intersection form on $H^2(X; \mathbb{Z})$ will be denoted by $\langle \cdot | \cdot \rangle$; the self-intersection $\langle a | a \rangle$ of a class a will also be denoted by a^2 for simplicity. This intersection form is Aut(X)-invariant. By the Hodge index theorem, it is positive definite on the real part of $H^{2,0}(X; \mathbb{C}) \oplus H^{0,2}(X; \mathbb{C})$ and it is non-degenerate and of signature $(1, h^{1,1}(X) - 1)$ on $H^{1,1}(X; \mathbb{R})$. Thus, we get:

Lemma 2.1. The restriction of $Aut(X)^*$ to the subspace $H^{2,0}(X; \mathbb{C})$ (resp. $H^{0,2}(X; \mathbb{C})$) is contained in a compact subgroup of $GL(H^{2,0}(X; \mathbb{C}))$ (resp. $GL(H^{0,2}(X; \mathbb{C}))$).

The Néron-Severi group $NS(X; \mathbb{Z})$ is, by definition, the intersection of $H^{1,1}(X; \mathbb{R})$ with the image of $H^2(X; \mathbb{Z})$ in $H^2(X; \mathbb{R})$. The Lefschetz theorem on (1, 1)-classes identifies $NS(X; \mathbb{Z})$ with the subgroup of $H^{1,1}(X; \mathbb{R})$ given by Chern classes of line bundles on X. The Néron-Severi group is Aut(X)-invariant, as well as $NS(X; R) := NS(X; \mathbb{Z}) \otimes_{\mathbb{Z}} R$ for $R = \mathbb{Q}$, \mathbb{R} , or \mathbb{C} . The dimension of $NS(X; \mathbb{R})$ is the **Picard number** $\rho(X)$.

2.1.2. Norm of f^* . Let $|\cdot|$ be any norm on the vector space $H^*(X; \mathbb{C})$. If L is a linear transformation of $H^*(X; \mathbb{C})$ we denote by ||L|| the associated operator norm and if $W \subset H^*(X; \mathbb{C})$ is an L-invariant subspace of $H^*(X; \mathbb{C})$, we denote by $||L||_W$ the operator norm of $L|_W$.

If u is an element of $H^{1,0}(X; \mathbb{C})$, then $u \wedge \overline{u}$ is an element of $H^{1,1}(X; \mathbb{R})$ such that $|u|^2 \leq C |u \wedge \overline{u}|$ for some constant C that depends only on the choice of norm on the cohomology; in particular, the norm of f^* on $H^{1,0}(X; \mathbb{C})$ is controlled by the norm of f^* on $H^{1,1}(X; \mathbb{C})$. Using complex conjugation, the same results hold on $H^{0,1}(X; \mathbb{C})$; by Poincaré duality we also control $||f^*||_{H^{p,q}(X; \mathbb{C})}$ for p + q > 2. Together with Lemma 2.1, we obtain:

Lemma 2.2. Let X be a compact Kähler surface. There exists a constant $C_0 > 1$ such that

$$C_0^{-1} \| f^* \|_{H^*(X;\mathbf{C})} \le \| f^* \|_{H^{1,1}(X;\mathbf{R})} \le \| f^* \|_{H^*(X;\mathbf{C})}$$

for every automorphism $f \in Aut(X)$.

2.2. The Kähler, nef, and pseudo-effective cones. (See [12, 62] for details on the notions introduced in this section.) Let $\operatorname{Kah}(X) \subset H^{1,1}(X; \mathbf{R})$ be the Kähler cone, i.e. the cone of classes of Kähler forms. Its closure $\overline{\operatorname{Kah}}(X)$ is a salient, closed, convex cone, and

(2.2)
$$\operatorname{Kah}(X) \subset \overline{\operatorname{Kah}}(X) \subset \{ v \in H^{1,1}(X; \mathbf{R}) \; ; \; \langle v \, | \, v \rangle \ge 0 \}.$$

The intersection $NS(X; \mathbf{R}) \cap Kah(X)$ is the **ample cone** Amp(X), while $NS(X; \mathbf{R}) \cap \overline{Kah}(X)$ is the **nef cone** Nef(X). They are all invariant under the action of Aut(X) on $H^{1,1}(X; \mathbf{R})$. We shall also say that the elements of $\overline{Kah}(X)$ are nef classes, but the notation Nef(X) will be reserved for $NS(X; \mathbf{R}) \cap \overline{Kah}(X)$. The set of classes of closed positive currents is the **pseudo-effective cone** Psef(X). This cone is an Aut(X)-invariant, salient, closed, convex cone. It is dual to $\overline{Kah}(X)$ for the intersection form (see [12, Lem. 4.1]):

(2.3)
$$\overline{\operatorname{Kah}}(X) = \{ u \in H^{1,1}(X; R) \; ; \; \langle u \, | \, v \rangle \ge 0 \quad \forall v \in \operatorname{Psef}(X) \}$$

and vice-versa.

We fix once and for all a reference Kähler form κ_0 with $[\kappa_0]^2 = \int \kappa_0 \wedge \kappa_0 = 1$. Then we define the **mass** of a pseudo-effective class a by $\mathbf{M}(a) = \langle a | [\kappa_0] \rangle$, or equivalently the mass of a closed positive current T by $\mathbf{M}(T) = \int T \wedge \kappa_0$; we may also extend this definition to any class, pseudo-effective or not (but then $\mathbf{M}(a) = \langle a | [\kappa_0] \rangle$ may be negative). By the compactness of the set of closed positive currents of mass 1, there exists a constant C such that

(2.4)
$$\forall a \in \operatorname{Psef}(X), \ C^{-1} |a| \leq \mathbf{M}(a) \leq C |a|.$$

If v is an element of $\operatorname{Psef}(X)$ and $v^2 \ge 0$, the Hodge index theorem implies that $\langle u | v \rangle \ge 0$ for every class $u \in H^{1,1}(X; \mathbb{R})$ such that $u^2 \ge 0$ and $\langle u | [\kappa_0] \rangle \ge 0$ (see Equation (2.5)). So, in Equation (2.3), the most important constraints come from the classes $v \in \operatorname{Psef}(X)$ with $v^2 < 0$. If v is such a class, its Zariski decomposition expresses v as a sum v = p(v) + n(v) with the following properties (see [12]):

- (1) this decomposition is orthogonal: $\langle p(v) | n(v) \rangle = 0$;
- (2) p(v) is a nef class, i.e. $p(v) \in \overline{\operatorname{Kah}}(X)$;
- (3) n(v) is negative: it is a sum $n(v) = \sum_{i} a_i [D_i]$ with positive coefficients $a_i \in \mathbf{R}^*_+$ of classes of irreducible curves $D_i \subset X$ such that the Gram matrix $(\langle D_i | D_j \rangle)$ is negative definite.

Proposition 2.3. If a ray $\mathbf{R}_+ v$ of the cone $\operatorname{Psef}(X)$ is extremal, then either $v^2 \ge 0$ or $\mathbf{R}_+ v = \mathbf{R}_+[D]$ for some irreducible curve D such that $D^2 < 0$. The cone $\operatorname{Psef}(X)$ contains at most countably many extremal rays $\mathbf{R}_+ v$ with $v^2 < 0$.

Let u be an isotropic element of $\overline{\text{Kah}}(X)$. If \mathbf{R}_+u is not an extremal ray of Psef(X), then u is proportional to an integral class $u' \in NS(X; \mathbf{Z})$.

Proof. If $\mathbf{R}_+ v$ is extremal, the Zariski decomposition v = p(v) + n(v) involves only one term. If v = p(v) then $v^2 \ge 0$. Otherwise v = n(v) and by extremality n(v) = a[D] for some irreducible curve D with $D^2 < 0$. The countability assertion follows, because $NS(X; \mathbf{Z})$ is countable. For the last assertion, multiply u by $\langle u | [\kappa_0] \rangle^{-1}$ to assume $\langle u | [\kappa_0] \rangle = 1$ and write u as a convex combination $u = \int v \, d\alpha(v)$, where α is a probability measure on Psef(X) such that α -almost every v satisfies

$$-\langle v | [\kappa_0] \rangle = 1,$$

- $\mathbf{R}_+ v$ is extremal in $\operatorname{Psef}(X)$ and does not contain u

Since u is nef, $\langle u | v \rangle \ge 0$ for each v; and u being isotropic, we get $v \in u^{\perp} \setminus \mathbf{R}u$ for α -almost every v. By the Hodge index theorem, $v^2 < 0$ almost surely. Now, the first assertion of this proposition implies that $v \in \mathbf{R}_+[D_v]$ for some irreducible curve $D_v \subset X$ with negative selfintersection; there are only countably many classes of that type, thus α is purely atomic, and ubelongs to $\operatorname{Vect}([D_v]; \alpha(v) > 0)$, a subspace of $\operatorname{NS}(X; \mathbf{R})$ defined over \mathbf{Q} . On this subspace, q_X is semi-negative, and by the Hodge index theorem its kernel is $\mathbf{R}u$. Since $\operatorname{Vect}([D_v]; \alpha(v) > 0)$ and q_X are defined over \mathbf{Q} , we deduce that u is proportional to an integral class. \Box

2.3. Non-elementary subgroups of Aut(X).

2.3.1. Isometries of Minkowski spaces. Consider the Minkowski space \mathbb{R}^{m+1} , endowed with its quadratic form q of signature (1, m) defined by $q(x) = x_0^2 - \sum_{i=1}^m x_i^2$. The corresponding bilinear form will be denoted $\langle \cdot | \cdot \rangle$. For future reference, note the following reverse Schwarz inequality:

with equality if and only if x and x' are collinear. We say that a subspace $W \subset \mathbf{R}^{m+1}$ is of **Minkowski type** if the restriction $q_{|W}$ is non-degenerate and of signature $(1, \dim(W) - 1)$.

In this section, we review some well-known facts concerning isometries of $\mathbf{R}^{1,m} = (\mathbf{R}^{m+1}, q)$ (see e.g. [73, 58, 49] for details). We denote by $|\cdot|$ the Euclidean norm on \mathbf{R}^{m+1} , and by $\mathbb{P}: \mathbf{R}^{m+1} \setminus \{0\} \to \mathbb{P}(\mathbf{R}^{m+1})$ the projection on the projective space $\mathbb{P}(\mathbf{R}^{m+1}) = \mathbb{P}^m(\mathbf{R})$.

The hyperboloid $\{x ; q(x) = 1\}$ has two components, and we denote by $O_{1,m}^+(\mathbf{R})$ the subgroup of the orthogonal group $O_{1,m}(\mathbf{R})$ that preserves the component $\mathcal{Q} = \{q(x) = 1 ; x_0 > 0\}$. Endowed with the distance $d_{\mathbb{H}}(x, y) = \cosh^{-1}\langle x | y \rangle$, \mathcal{Q} is a model of the real hyperbolic space \mathbb{H}^m of dimension m. The boundary at infinity of \mathbb{H}^m will be identified with $\partial \mathbb{P}(\mathcal{Q}) \subset \mathbb{P}(\mathbf{R}^{m+1})$ and will be denoted by $\partial \mathbb{H}^m$. It is the set of isotropic lines of q.

Any isometry γ of \mathbb{H}^m is induced by an element of $O_{1,m}^+(\mathbf{R})$, and extends continuously to $\partial \mathbb{H}^m$: its action on $\partial \mathbb{H}^m$ is given by its linear projective action on $\mathbb{P}(\mathbf{R}^{m+1})$. Isometries are classified in three types, according to their fixed point set in $\mathbb{H}^m \cup \partial \mathbb{H}^m$:

- $-\gamma$ is **elliptic** if γ has a fixed point in \mathbb{H}^m ;
- $-\gamma$ is **parabolic** if γ has no fixed point in \mathbb{H}^m and a unique fixed point in $\partial \mathbb{H}^m$;

 $-\gamma$ is **loxodromic** if γ has no fixed point in \mathbb{H}^m and exactly two fixed points in $\partial \mathbb{H}^m$.

A subgroup Γ of $O_{1,m}^+(\mathbf{R})$ is **non-elementary** if it does not preserve any finite subset of $\mathbb{H}^m \cup \partial \mathbb{H}^m$. Equivalently Γ is non-elementary if and only if it contains two loxodromic elements with disjoint fixed point sets. The group $O_{1,m}^+(\mathbf{R})$ admits a **Cartan** or **KAK decomposition**. To state it, denote by $e_0 = (1, 0, \dots, 0)$ the first vector of the canonical basis of \mathbf{R}^{m+1} ; it is an element of \mathbb{H}^m , and its stabilizer in $O_{1,m}^+(\mathbf{R})$ is a maximal compact subgroup, isomorphic to $O_{m-1}(\mathbf{R})$.

Lemma 2.4 (See §I.5 of [49]). Every $\gamma \in O_{1,m}^+(\mathbf{R})$ can be written (non-uniquely) as $\gamma = k_1 a k_2$, where $k_i \in \text{Stab}(\mathbf{e}_0)$ and a is a matrix of the form

$$\begin{pmatrix} \cosh r & \sinh r & 0\\ \sinh r & \cosh r & 0\\ 0 & 0 & \mathrm{id}_{m-1} \end{pmatrix}$$

with $r = d_{\mathbb{H}}(e_0, \gamma e_0)$.

Corollary 2.5. If $\|\cdot\|$ denotes the operator norm associated to the euclidean norm in \mathbb{R}^{m+1} , then $\|\gamma\| = \|a\|$, where $\gamma = k_1 a k_2$ is any Cartan decomposition of γ . In particular $\|\gamma\| = \|\gamma^{-1}\|$ and

$$\|\gamma\| \asymp \cosh d_{\mathbb{H}}(e_0, \gamma(e_0)) \asymp |\gamma e_0|.$$

Furthermore for every $e \in \mathbb{H}^m$ *and any* $\gamma \in \mathsf{O}^+_{1,m}(\mathbf{R})$

$$\|\gamma\| \asymp \cosh d_{\mathbb{H}}(e, \gamma(e)),$$

where the implied constant depends only on the base point e.

2.3.2. *Irreducibility*. A non-elementary subgroup of $O_{1,m}^+(\mathbf{R})$ does not need to act irreducibly on \mathbf{R}^{m+1} . Proposition 2.8, below, clarifies the possible situations.

Lemma 2.6. Let Γ be a non-elementary subgroup of $O_{1,m}^+(\mathbf{R})$ (resp. γ be an element of $O_{1,m}^+(\mathbf{R})$). Let W be a subspace of $\mathbf{R}^{1,m}$.

- (1) If W is Γ -invariant, then either $(W, q|_W)$ is a Minkowski space and $\Gamma|_W$ is non-elementary, or $q|_W$ is negative definite and $\Gamma|_W$ is contained in a compact subgroup of GL(W).
- (2) If W is γ -invariant and contains a vector w with q(w) > 0, then $\gamma|_W$ has the same type (elliptic, parabolic, or loxodromic) as γ ; in particular, W contains the γ -invariant isotropic lines if γ is parabolic or loxodromic.

Proof. The restriction $q|_W$ is either a Minkowski form or is negative definite. Indeed, it cannot be positive definite, because W would then be a Γ -invariant line intersecting the hyperbolic space \mathbb{H}^m in a fixed point; and it cannot be degenerate, since otherwise its kernel would give a Γ -invariant point on $\partial \mathbb{H}^m$. If $q|_W$ is a Minkowski form and $\Gamma|_W$ is elementary, then Γ preserves a finite subset of $(\mathbb{H}^m \cup \partial \mathbb{H}^m) \cap W$ and Γ itself is elementary. This proves the first assertion. The proof of the second one is similar.

Let Γ be a non-elementary subgroup of $O_{1,m}^+(\mathbf{R})$. Let $Zar(\Gamma) \subset O_{1,m}(\mathbf{R})$ be the Zariski closure of Γ , and $G = Zar(\Gamma)^{irr}$ the **identity component** of $Zar(\Gamma)$, for the Zariski topology. Note that the Lie group $G(\mathbf{R})$ is not necessarily connected for the euclidean topology.

Lemma 2.7 (see [30], §4.1). The group $\Gamma \cap G(\mathbf{R})$ has finite index in Γ . If Γ_0 is a finite index subgroup of Γ , then $\mathsf{Zar}(\Gamma_0)^{\mathrm{irr}} = G$.

10

Proposition 2.8. Let $\Gamma \subset O_{1,m}^+(\mathbf{R})$ be non-elementary.

(1) The representation of $\Gamma \cap G(\mathbf{R})$ (resp. of $G(\mathbf{R})$) on $\mathbf{R}^{1,m}$ splits as a direct sum of irreducible representations, with exactly one irreducible factor of Minkowski type:

$$\mathbf{R}^{1,m} = V_+ \oplus V_0;$$

here V_+ is of Minkowski type, and V_0 is an orthogonal sum of irreducible representations $V_{0,j}$ on which the quadratic form q is negative definite.

- (2) The restriction $G|_{V_+}$ coincides with $SO(V_+; q|_{V_+})$.
- (3) The subspaces V_+ and V_0 are Γ -invariant, and the representation of Γ on V_+ is strongly irreducible.

Proof. A group Γ is non-elementary if and only if any of its finite index subgroups is nonelementary. So, we can apply Lemma 2.6 to $\Gamma \cap G(\mathbf{R})$: if $W \subset \mathbf{R}^{1,m}$ is a non-trivial ($\Gamma \cap G(\mathbf{R})$)-invariant subspace, $q|_W$ is non-degenerate. As a consequence, $\mathbf{R}^{1,m}$ is the direct sum $W \oplus W^{\perp}$, where W^{\perp} is the orthogonal complement of W with respect to q. This implies that the representation of $\Gamma \cap G(\mathbf{R})$ on $\mathbf{R}^{1,m}$ splits as a direct sum of irreducible representations, with exactly one irreducible factor of Minkowski type, as asserted in (1).

The group G preserves this decomposition, and by Proposition 1 of [6], the restriction $G|_{V_+}$ coincides with $SO(V_+; q|_{V_+})$; this group is isomorphic to the almost simple group $SO_{1,k}(\mathbf{R})$, with $1 + k = \dim(V_+)$. This proves the second assertion.

Since G is normalized by Γ , we see that for any $\gamma \in \Gamma$, γV^+ is a G-invariant subspace of the same dimension as V^+ and on which q is of Minkowski type. Hence V_+ , as well as its orthogonal complement V_0 are Γ -invariant. By Lemma 2.7, the action of Γ on V_+ is strongly irreducible; indeed, if a finite index subgroup Γ_0 in Γ preserves a non-trivial subspace of V_+ then, by Zariski density of $\Gamma_0 \cap G(\mathbf{R})$ in $G(\mathbf{R})$, this subspace must be V_+ itself. On V_0 , Γ permutes the irreducible factors $V_{0,j}$.

Now, set $V = \mathbf{R}^{1,m}$ and assume that there is a lattice $V_{\mathbf{Z}} \subset V$ such that

- (i) $V_{\mathbf{Z}}$ is Γ -invariant;
- (ii) the quadratic form q is an integral quadratic form on $V_{\mathbf{Z}}$.

In other words, there is a basis of V with respect to which q and the elements of Γ are given by matrices with integer coefficients. In particular, V has a natural Q-structure, with $V(\mathbf{Q}) = V_{\mathbf{Z}} \otimes_{\mathbf{Z}} \mathbf{Q}$. This situation naturally arises for the action of automorphisms of compact Kähler surfaces on NS(X; **R**). The next lemma will be useful in [25].

Lemma 2.9. If Γ contains a parabolic element, the decomposition $V_+ \oplus V_0$ is defined over \mathbf{Q} , $\Gamma|_{V_0}$ is a finite group, and G is the subgroup $\mathsf{SO}(V_+;q) \times \{\mathrm{id}_{V_0}\}$ of $\mathsf{O}(V;q)$.

Proof. If $\gamma \in \Gamma$ is parabolic, it fixes pointwise a unique isotropic line, therefore this line is defined over \mathbf{Q} . In addition it must be contained in V_+ because $(\gamma^n(u))_{n\geq 0}$ converges to the boundary point determined by this line for every $u \in \mathbb{H}^m$. So, V_+ contains at least one non-zero element of $V_{\mathbf{Z}}$. Since the action of Γ on V_+ is irreducible, the orbit of this vector generates V_+ and is contained in $V_{\mathbf{Z}}$, so V_+ is defined over \mathbf{Q} . Its orthogonal complement V_0 is also defined over \mathbf{Q} , because q itself is defined over \mathbf{Q} . As a consequence, $\Gamma|_{V_0}$ preserves the lattice $V_0 \cap V_{\mathbf{Z}}$ and the negative definite form $q|_{V_0}$; hence, it is finite. Thus $G|_{V_0}$ is trivial and the last assertion follows from the above mentioned equality $G|_{V_+} = \mathrm{SO}(V_+; q|_{V_+})$.

In [24], an example is given which shows that the existence of parabolic element is necessary for Lemma 2.9 to hold.

2.3.3. The hyperbolic space \mathbb{H}_X . Let X be a compact Kähler surface. By the Hodge index theorem, the intersection form on $H^{1,1}(X, \mathbb{R})$ has signature $(1, h^{1,1}(X) - 1)$. The hyperboloid $\{u \in H^{1,1}(X, \mathbb{R}), \langle u | u \rangle\} = 1$ has two connected components, one of which intersecting the Kähler cone. By definition, this component is the hyperbolic space \mathbb{H}_X ; it is a model of \mathbb{H}^m , for $m = h^{1,1}(X) - 1$. We denote by $d_{\mathbb{H}}$ the hyperbolic distance: as before, $\cosh(d_{\mathbb{H}}(u, v)) = \langle u | v \rangle$. From Lemma 2.2 and Corollary 2.5 we see that if $|\cdot|$ is any norm on $H^*(X, \mathbb{C})$, then $||f^*|| \approx ||(f^*)^{-1}|| \approx \langle [\kappa_0] | f^*[\kappa_0] \rangle$ (here κ_0 is the fixed Kähler form introduced in Section 2.2).

According to the classification of isometries of hyperbolic spaces, there are three types of automorphisms: **elliptic**, **parabolic** and **loxodromic**. An important fact for us is that the type of isometry is related to the dynamics on X; for instance, every parabolic automorphism preserves a genus 1 fibration, every loxodromic automorphism has positive topological entropy (see [21]). A subgroup Γ of Aut(X) is said to be **non-elementary** if its action on \mathbb{H}_X is non-elementary.

2.3.4. Automorphisms and Néron-Severi groups. Let X be a compact Kähler surface and Γ be a non-elementary subgroup of Aut(X). Let $\Gamma_{p,q}^*$ be the image of Γ in GL($H^{p,q}(X; \mathbb{C})$), and Γ^* be its image in GL($H^2(X; \mathbb{C})$). If we combine Proposition 2.8 together with Lemma 2.1 for $\Gamma_{1,1}^*$, we get an invariant decomposition

(2.6)
$$H^{1,1}(X;\mathbf{R}) = H^{1,1}(X;\mathbf{R})_+ \oplus H^{1,1}(X;\mathbf{R})_0.$$

Denote by $H^2(X; \mathbf{R})_0$ the direct sum of $H^{1,1}(X; \mathbf{R})_0$ and of the real part of $H^{2,0}(X; \mathbf{C}) \oplus H^{0,2}(X; \mathbf{C})$; then

(2.7)
$$H^{2}(X; \mathbf{R}) = H^{1,1}(X; \mathbf{R})_{+} \oplus H^{2}(X; \mathbf{R})_{0}$$

and $\Gamma^*|_{H^2(X;\mathbf{R})_0}$ is contained in a compact group (see Lemma 2.1). The Néron-Severi group is Γ -invariant, and since X is projective it contains a vector with positive self-intersection. Then Proposition 2.8 and Lemma 2.6 imply:

Proposition 2.10. Let X be a compact Kähler surface and Γ be a non-elementary subgroup of Aut(X). Then $H^{1,1}(X; \mathbf{R})_+ = NS(X; \mathbf{R})_+$ is a Minkowski space, and the action of Γ on this space is non-elementary and strongly irreducible.

Since non-elementary groups of isometries of \mathbb{H}^m occur only for $m \ge 2$, we get:

Corollary 2.11. Under the assumptions of Proposition 2.10, the Picard number $\rho(X)$ is greater than or equal to 3. If equality holds then $NS(X; \mathbf{R})_+ = NS(X; \mathbf{R})$ and the action of Γ on $NS(X; \mathbf{R})$ is strongly irreducible.

From now on we set:

(2.8)
$$\Pi_{\Gamma} := H^{1,1}(X; \mathbf{R})_{+} = \mathrm{NS}(X; \mathbf{R})_{+}.$$

This is a Minkowski space on which Γ acts strongly irreducibly; the intersection form is negative definite on the orthogonal complement $\Pi_{\Gamma}^{\perp} \subset H^{1,1}(X; \mathbf{R})$. By Proposition 2.8.(2), the group $G = \operatorname{Zar}(\Gamma)^{\operatorname{irr}}$ satisfies $G(\mathbf{R})|_{\Pi_{\Gamma}} = \operatorname{SO}(\Pi_{\Gamma})$. If Γ contains a parabolic element, then Π_{Γ} is rational with respect to the integral structures of NS $(X; \mathbf{Z})$ and $H^2(X; \mathbf{Z})$, and $G(\mathbf{R}) =$ $\operatorname{SO}(\Pi_{\Gamma}) \times \{\operatorname{id}_{\Pi_{\Gamma}^{\perp}}\}$ (see Lemma 2.9). 2.3.5. Invariant algebraic curves I. Assume that Γ is non-elementary and let $C \subset X$ be an irreducible algebraic curve with a finite Γ -orbit. Then the action of Γ on Vect_Z { $f^*[C]$; $f \in \Gamma$ } \subset NS(X; Z) factors through a finite group. From Propositions 2.8 and 2.10 we deduce that the intersection form is negative definite on Vect_Z($\Gamma^* \cdot [C]$), thus Vect_R($\Gamma^* \cdot [C]$) is one of the irreducible factors of NS(X, R)₀. This argument, together with Grauert's contraction theorem, leads to the following result (we refer to [21, 60] for a proof; the result holds more generally for subgroups containing a loxodromic element):

Lemma 2.12. Let X be a compact Kähler surface and Γ be a non-elementary group of automorphisms on X. Then, there are at most finitely many Γ -periodic irreducible curves. The intersection form is negative definite on the subspace of $NS(X; \mathbb{Z})$ generated by the classes of these curves. There is a compact complex analytic surface X_0 and a Γ -equivariant bimeromorphic morphism $X \to X_0$ that contracts these curves and is an isomorphism in their complement.

2.3.6. The limit set. Let $\Gamma \subset \operatorname{Aut}(X)$ be non-elementary. The limit set of Γ is the closed subset $\operatorname{Lim}(\Gamma) \subset \partial \mathbb{H}_X \subset \mathbb{P}(H^{1,1}(X; \mathbf{R}))$ defined by one of the following equivalent assertions:

- (a) $\operatorname{Lim}(\Gamma)$ is the smallest, non-empty, closed, and Γ -invariant subset of $\mathbb{P}(\overline{\mathbb{H}_X})$;
- (b) Lim(Γ) ⊂ ∂H_X is the closure of the set of fixed points of loxodromic elements of Γ in ∂H_X;
- (c) $\operatorname{Lim}(\Gamma)$ is the accumulation set of any Γ -orbit $\Gamma(\mathbb{P}(v)) \subset \mathbb{P}(H^{1,1}(X;\mathbf{R}))$, for any $v \notin \Pi_{\Gamma}^{\perp}$.

We refer to [58, 73] for a study of such limit sets. From the second characterization we get:

Lemma 2.13. The limit set $\operatorname{Lim}(\Gamma)$ of a non-elementary group is contained in $\mathbb{P}(\Pi_{\Gamma}) \cap \partial \mathbb{H}_X$.

From the third characterization, $\operatorname{Lim}(\Gamma)$ is contained in the closure of $\Gamma(\mathbb{P}([\kappa]))$ for every Kähler form κ on X. Since X must be projective, we can chose $[\kappa]$ in $\operatorname{NS}(X; \mathbb{Z})$. As a consequence, $\operatorname{Lim}(\Gamma)$ is contained in $\operatorname{Nef}(X)$:

Lemma 2.14. Let X be a compact Kähler surface. If Γ is a non-elementary subgroup of Aut(X) its limit set satisfies $Lim(\Gamma) \subset \mathbb{P}(Nef(X)) \subset \mathbb{P}(NS(X; \mathbf{R}))$.

2.4. **Parabolic automorphisms.** The facts collected here will be used in Section 10. Let f be a parabolic automorphism of a compact Kähler surface. Then f^* preserves a unique point on $\partial \mathbb{H}_X$, and f preserves a unique genus 1 fibration $\pi_f \colon X \to B$ onto some Riemann surface B; the fixed point of f^* on $\partial \mathbb{H}_X$ is given by the class [F] of any fiber of π_f (see [21]). The fibers of π_f are the elements of the linear system |F|, π_f is uniquely determined by [F], and if g is another automorphism of X that preserves a smooth fiber of π_f (resp. the point $\mathbb{P}[F] \in \mathbb{P}NS(X; \mathbb{R})$), then g preserves the fibration and is elliptic or parabolic.

Proposition 2.15. Let X be a compact Kähler surface and let f be a parabolic automorphism of X, preserving the genus 1 fibration $\tau: X \to B$. Consider the group $\operatorname{Aut}(X; \tau) := \{g \in \operatorname{Aut}(X) ; \exists g_B \in \operatorname{Aut}(B), \tau \circ g = g_B \circ \tau\}$, and assume that the image of the homomorphism $g \in \operatorname{Aut}(X; \tau) \to g_B \in \operatorname{Aut}(B)$ is infinite. Then, X is a torus \mathbb{C}^2/Λ .

This result directly follows from the proof of Proposition 3.6 in [29]. In particular the automorphism $f_B \in Aut(B)$ such that $\pi_f \circ f = f_B \circ \pi_f$ has finite order when X is a K3, an Enriques, or a rational surface.

SERGE CANTAT AND ROMAIN DUJARDIN

3. EXAMPLES AND CLASSIFICATION

Here, after a few remarks on the classification of surfaces admitting non-elementary groups of automorphisms, we describe the main properties of Wehler examples, sufficient to derive the Corollary of Section 1.3.2 from our main theorems. Further examples are described in [27].

3.1. Surfaces admitting non-elementary groups of automorphisms.

3.1.1. Minimal models. We refer to Theorem 10.1 of [21] for the following result:

Theorem 3.1. Let X be a compact Kähler surface with a loxodromic automorphism. Either X is a rational surface, and there is a birational morphism $\pi: X \to \mathbb{P}^2_{\mathbf{C}}$. Or the Kodaira dimension of X is equal to 0, and there is an Aut(X)-equivariant bimeromorphic morphism $\pi: X \to X_0$ such that X_0 is a compact torus, a K3 surface, or an Enriques surface. In particular, $h^{2,0}(X)$ equals 0 or 1.

Remark 3.2. If X is a torus or K3 surface, there is a holomorphic 2-form Ω_X on X that does not vanish and satisfies $\int_X \Omega_X \wedge \overline{\Omega_X} = 1$. It is unique up to multiplication by a complex number of modulus 1. A consequence of utmost importance to us is that the volume form $\Omega_X \wedge \overline{\Omega_X}$ is Aut(X)-invariant. If X is an Enriques surface, and $\tilde{X} \to X$ is its universal cover, then \tilde{X} is a K3 surface: the volume form $\Omega_{\tilde{X}} \wedge \overline{\Omega_{\tilde{X}}}$ determines an Aut(X)-invariant volume form on X. So, if X is not rational, the dynamics of Aut(X) is conservative: it preserves a **canonical volume form** determined by the complex structure of X. Furthermore, if $Y \subset X$ is a totally real surface invariant by some subgroup $\Gamma \subset Aut(X)$, the invariant volume form induces a Γ -invariant volume form on Y (see [26, Rem. 2.3] for details). This holds in particular for $Y = X(\mathbf{R})$ when X is defined over \mathbf{R} .

It follows from Theorem 3.1 that, in most cases, Aut(X) is countable (see [21, Rem. 3.3]):

Proposition 3.3. Let X be a compact Kähler surface. If Aut(X) contains a loxodromic element, then the kernel of the homomorphism $Aut(X) \rightarrow Aut(X)^* \subset GL(NS(X; \mathbb{Z}))$ is finite unless X is a torus. So, if Aut(X) is non-elementary, then Aut(X) is discrete or X is a torus.

3.1.2. Projectivity. The next theorem is established in [27] (see also [24]).

Theorem 3.4. Let X be a compact Kähler surface such that there exists a non-elementary subgroup $\Gamma \leq Aut(X)$. Then X is projective.

3.2. Wehler surfaces (see [32, 76, 79, 80]). Consider the variety $M = \mathbb{P}^1 \times \mathbb{P}^1 \times \mathbb{P}^1$ and let π_1, π_2 , and π_3 be the projections on the first, second, and third factor: $\pi_i(z_1, z_2, z_3) = z_i$. Denote by L_i the line bundle $\pi_i^*(\mathcal{O}(1))$ and set $L = L_1^2 \otimes L_2^2 \otimes L_3^2 = \pi_1^*(\mathcal{O}(2)) \otimes \pi_2^*(\mathcal{O}(2)) \otimes \pi_3^*(\mathcal{O}(2))$. Since $K_{\mathbb{P}^1} = \mathcal{O}(-2)$, this line bundle L is the dual of the canonical bundle K_M . Let X be an irreducible surface in the linear system $|L| \simeq \mathbb{P}(H^0(M, L))$; using affine coordinates (x_1, x_2, x_3) on $M = \mathbb{P}^1 \times \mathbb{P}^1 \times \mathbb{P}^1$, it is defined by a polynomial equation $P(x_1, x_2, x_3) = 0$ whose degree with respect to each variable is ≤ 2 (see [18, 71] for explicit examples). These surfaces will be referred to as Wehler surfaces or (2,2,2)-surfaces; modulo Aut(M), they form a family of dimension 17. Fix $k \in \{1, 2, 3\}$ and denote by i < j the other indices. If we project X to $\mathbb{P}^1 \times \mathbb{P}^1$ by $\pi_{ij} = (\pi_i, \pi_j)$, we get a 2 to 1 cover (but some fibers may be rational curves). If X is smooth, it is a K3 surface and any birational self-map of X is an automorphism; in particular, the involution σ_k that permutes the two points in each (general) fiber of π_{ij} is an involutive automorphism of X. **Proposition 3.5.** There is a countable union of proper Zariski closed subsets $(W_i)_{i\geq 0}$ in |L| such that

- (1) if X is an element of $|L|\setminus W_0$, then X is a smooth K3 surface and X does not contain any fiber of the projections π_{ij} ;
- (2) if X is an element of $|L| \setminus (\bigcup_i W_i)$, the restriction morphism $\operatorname{Pic}(M) \to \operatorname{Pic}(X)$ is surjective. In particular its Picard number is $\rho(X) = 3$.

See [32] for the proof of this proposition, as well as that of Lemma 3.6. From the second assertion, we deduce that for a very general X, $\operatorname{Pic}(X)$ is isomorphic to $\operatorname{Pic}(M)$: it is the free Abelian group of rank 3, generated by the classes $c_i := [(L_i)_{|X}]$. The elements of $|(L_i)_{|X}|$ are the curves of X given by the equations $z_i = \alpha$ for some $\alpha \in \mathbb{P}^1$. Their arithmetic genus is equal to 1: the projection $(\pi_i)_{|X}: X \to \mathbb{P}^1$ is a genus 1 fibration. The intersection form is given by $c_i^2 = 0$ and $\langle c_i | c_j \rangle = 2$ if $i \neq j$.

Lemma 3.6. Assume that X does not contain any fiber of the projection π_{ij} . Then, the involution σ_k^* preserves the subspace $\mathbf{Z}c_1 \oplus \mathbf{Z}c_2 \oplus \mathbf{Z}c_3$ of $\operatorname{NS}(X; \mathbf{Z})$ and $\sigma_k^*c_i = c_i$, $\sigma_k^*c_j = c_j$, $\sigma_k^*c_k = -c_k + 2c_i + 2c_j$. Equivalently, the action of σ_k^* on $\operatorname{Vect}_{\mathbf{R}}(c_1, c_2, c_3)$ preserves the classes c_i and c_j and acts as a reflection with respect to the hyperplane $\operatorname{Vect}(c_i, c_j) \subset \operatorname{NS}(X; \mathbf{R})$. In other words, setting $u_k = (c_1 + c_2 + c_3) - 2c_k$, $\sigma_k(v) = v + \frac{1}{2}\langle v | u_k \rangle u_k$ for all v in $\mathbf{Z}c_1 \oplus \mathbf{Z}c_2 \oplus \mathbf{Z}c_3$.

We can now combine this lemma with the previous proposition to obtain:

Proposition 3.7. If X is a very general Wehler surface then:

- (1) X is a smooth K3 surface with Picard number 3;
- (2) Aut(X) is equal to $\langle \sigma_1, \sigma_2, \sigma_3 \rangle$, it is a free product of three copies of $\mathbb{Z}/2\mathbb{Z}$, and Aut(X)* is a finite index subgroup in the group of integral isometries of NS(X; \mathbb{Z});
- (3) $\operatorname{Aut}(X)^*$ acts strongly irreducibly on $\operatorname{NS}(X; \mathbf{R})$;
- (4) $\operatorname{Aut}(X)$ does not preserve any algebraic curve $D \subset X$;
- (5) the limit set of $\operatorname{Aut}(X)^*$ is equal to $\partial \mathbb{H}_X$;
- (6) the compositions $\sigma_i \circ \sigma_j$ and $\sigma_i \circ \sigma_j \circ \sigma_k$ are respectively parabolic and loxodromic for every triple (i, j, k) with $\{i, j, k\} = \{1, 2, 3\}$.

Proof. The first three assertions follow from Proposition 3.5, [18, §1.5] and [32, Thm 3.6]. For the fourth one, note that any invariant curve D would yield a non-trivial fixed point [D] in $NS(X; \mathbb{Z})$, contradicting assertion (3). The fifth one follows from the second because the limit set of a lattice in $Isom(NS(X; \mathbb{R}))$ is always equal to $\partial \mathbb{H}_X$. The last assertion follows from explicit linear algebra calculations (see [18]).

4. GLOSSARY OF RANDOM DYNAMICS, I

We now initiate the random iteration by introducing a probability measure on Aut(X). In this section we introduce a first set of ideas from the theory of random dynamical systems.

4.1. Random holomorphic dynamical systems. Let X be a compact Kähler surface, such that Aut(X) is non-elementary. Note that Aut(X) is locally compact for the topology of uniform convergence –in many interesting cases it is actually discrete (see Proposition 3.3)– so it admits a natural Borel structure. We fix some Riemannian structure on X, for instance the one induced

by the Kähler form κ_0 . For $f \in Aut(X)$, we denote by $||f||_{C^1}$ the maximum of $||Df_x||$ where the norm of $Df_x : T_x M \to T_{f(x)} M$ is computed with respect to this Riemannian metric.

We consider a probability measure ν on Aut(X) satisfying the **moment condition** (or integrability condition)

(4.1)
$$\int \left(\log \|f\|_{C^1(X)} + \log \|f^{-1}\|_{C^1(X)} \right) \, d\nu(f) < +\infty.$$

The finiteness of the integral in (4.1) does not depend on our choice of Riemannian metric. When the support of ν is finite, the integrability (4.1), as well as stronger moment conditions which will appear later (see Conditions (5.23) and (5.24)), are obviously satisfied.

Lemma 4.1. The measure ν satisfies the moment condition (4.1) if and only if, for all $k \ge 1$, it satisfies the higher moment conditions

$$\int \left(\log \|f\|_{C^k(X)} + \log \|f^{-1}\|_{C^k(X)} \right) \, d\nu(f) < \infty.$$

This lemma follows from the Cauchy estimates. In particular, if ν satisfies (4.1), then it satisfies a similar moment condition for the C^2 norm, a property required to apply Pesin's theory.

Given ν , we shall consider independent, identically distributed sequences $(f_n)_{n\geq 0}$ of random automorphisms of X with distribution ν , and study the dynamics of random compositions of the form $f_{n-1} \circ \cdots \circ f_0$. The data (X, ν) will be referred to as a **random holomorphic dynamical system** on X. Many properties of (X, ν) depend on the properties of the subgroup

(4.2)
$$\Gamma = \Gamma_{\nu} := \langle \operatorname{Supp}(\nu) \rangle$$

generated by the support of ν in Aut(X). If Γ_{ν} is non-elementary, we say that (X, ν) is **non-elementary**.

4.2. Invariant and stationary measures. Let G be a topological group and ν be a probability measure on G. Consider a measurable action of G on some measurable space (M, \mathcal{A}) . Every $f \in G$ determines a push-forward operator $\mu \mapsto f_*\mu$, acting on positive (resp. probability) measures μ on (M, \mathcal{A}) . By definition, a probability measure μ on (M, \mathcal{A}) is ν -stationary if

(4.3)
$$\int f_* \mu \, d\nu(f) = \mu,$$

and it is ν -almost surely invariant if $f_*\mu = \mu$ for ν -almost every f. Let us stress that we only deal with probability measures in this definition. A stationary measure is **ergodic** if it is an extremal point of the closed convex set of stationary measures (see [8, §2.1.3]). If μ is almost surely invariant then it is stationary. If M is compact, the action $G \times M \to M$ is continuous, and \mathcal{A} is the Borel σ -algebra, the Kakutani fixed point theorem implies the existence of at least one stationary measure. On the other hand the existence of an invariant measure is a very restrictive property (see Sections 1.2 and 5.3). Following Furstenberg [51] we say that an action is **stiff** (or ν -stiff) if any ν -stationary measure is ν -almost surely invariant.

We shall consider several measurable actions of Aut(X): its tautological action on X, but also its action on the projectivized tangent bundle $\mathbb{P}(TX)$, on cohomology groups of X and their projectivizations, on spaces of currents, etc. In all cases, M will be a locally compact space and \mathcal{A} its Borel σ -algebra, which will be denoted by $\mathcal{B}(M)$. **Remark 4.2.** Since X is compact and the action $\operatorname{Aut}(X) \times X \to X$ is continuous, a probability measure μ on $(X, \mathcal{B}(X))$ is ν -almost surely invariant if and only if it is invariant under the action of the closure of Γ_{ν} in $\operatorname{Aut}(X)$; this follows from the dominated convergence theorem.

4.3. **Random compositions.** Set $\Omega = \operatorname{Aut}(X)^{\mathbb{N}}$, endowed with its product topology. The associated Borel σ -algebra coincides with the product σ -algebra and is generated by cylinders (see § 7.1). We endow Ω with the product measure $\nu^{\mathbb{N}}$. For $\omega \in \Omega$, we set $f_{\omega}^{0} = \operatorname{id}$ and for n > 0 we denote by f_{ω}^{n} the left composition of the *n* first terms of ω , that is

(4.4)
$$f_{\omega}^{n} = f_{n-1} \circ \cdots \circ f_{0}.$$

In particular $f_{\omega}^1 = f_0$. Let us record for future reference the following consequence of the Borel-Cantelli lemma. We denote by $\sigma \colon \Omega \to \Omega$ the unilateral shift.

Lemma 4.3. If (X, ν) is a random dynamical system satisfying the moment condition (4.1), then for $\nu^{\mathbf{N}}$ -almost every sequence $\omega = (f_n) \in \Omega$,

$$\frac{1}{n} \left(\log \|f_n\|_{C^1} + \log \|f_n^{-1}\|_{C^1} \right) \xrightarrow[n \to \infty]{} 0.$$

5. Furstenberg theory in $H^{1,1}(X; \mathbf{R})$

Consider a non-elementary random holomorphic dynamical system (X, ν) on a compact Kähler surface, satisfying the moment condition (4.1). In this section, we analyze the linear action of (X, ν) on $H^{1,1}(X, \mathbf{R})$ by using the theory of random products of matrices. The books by Bougerol and Lacroix [13] and by Benoist and Quint [8] are good references to this theory.

5.1. Moments and cohomology. Let M be a compact connected manifold of dimension m, endowed with some Riemannian metric g. If $f: M \to M$ is a smooth map, $||f||_{C^1}$ denotes the maximum norm of its tangent action, computed with respect to g (see Section 4.1). Thus, f is a Lipschitz map with $\operatorname{Lip}(f) = ||f||_{C^1}$ for the distance determined by g; in particular $||f||_{C^1} \ge 1$ whenever f is onto. Fix a norm $|\cdot|_{H^k}$ on each cohomology group $H^k(M; \mathbf{R})$, for $0 \le k \le m$.

Lemma 5.1. There is a constant C > 0, that depends only on M, g, and the norms $|\cdot|_{H^k}$, such that $|f^*[\alpha]|_{H^k} \leq C^k \operatorname{Lip}(f)^k |[\alpha]|_{H^k}$ for every class $[\alpha] \in H^k(M; \mathbf{R})$ and every map $f: M \to M$ of class C^1 . In other words, the operator norm $||f^*||_{H^k}$ is controlled by the Lipschitz constant:

$$||f^*||_{H^k} \leq C^k \operatorname{Lip}(f)^k \leq C^k ||f||_{C^1}^k.$$

Proof. Pick a basis of $H_k(M; \mathbf{R}) \simeq H^k(M; \mathbf{R})^*$ given by smoothly immersed, compact, kdimensional manifolds $\iota_i \colon N_i \to M$, and a basis of $H^k(M; \mathbf{R})$ given by smooth k-forms α_j . The integral $\int_{N_i} \iota_i^*(f^*\alpha_j)$ is bounded from above by $C^k \|f\|_{C^1}^k$ for some constant C, because

(5.1)
$$|(f^*\alpha_j)_x(v_1,\ldots,v_k)| = |\alpha_j(f_*v_1,\ldots,f_*v_k)| \leq c_j ||f||_{C^1}^k \prod_{\ell=1}^k |v_\ell|_g$$

for every $x \in M$ and every k-tuple of tangent vectors $v_{\ell} \in T_x M$; here, c_j is the supremum of the norm of the multilinear map $(\alpha_j)_x$ over $x \in M$.

If ν is a probability measure on Diff(M) satisfying the moment condition (4.1), then

(5.2)
$$\forall 1 \leq k \leq m, \quad \int_{\mathsf{Diff}(M)} \log\left(\|f^*\|_{H^k}\right) + \log\left(\|(f^{-1})^*\|_{H^k}\right) \, d\nu(f) < +\infty.$$

If we specialize this to automorphisms of compact Kähler surfaces we get

(5.3)
$$\int_{\operatorname{Aut}(X)} \log\left(\|f^*\|_{H^{1,1}}\right) + \log\left(\|(f^{-1})^*\|_{H^{1,1}}\right) \, d\nu(f) < +\infty$$

which is actually equivalent to (5.2) by Lemma 2.2. We saw in §2.3.3 that $||f^*||_{H^{1,1}} \simeq ||(f^{-1})^*||_{H^{1,1}}$, so this last condition is in turn equivalent to

(5.4)
$$\int_{\operatorname{Aut}(X)} \log\left(\|f^*\|_{H^{1,1}}\right) \, d\nu(f) < +\infty.$$

5.2. Cohomological Lyapunov exponent. As in § 2.1.2, we denote by $|\cdot|$ a norm on $H^{1,1}(X, \mathbb{R})$ and by $\|\cdot\|$ the associated operator norm. The linear action induced by the random dynamical system (X, ν) on $H^{1,1}(X, \mathbb{R})$ defines a random product of matrices. Since the moment condition (5.4) is satisfied, we can define the **upper Lyapunov exponent** $\lambda_{H^{1,1}}$ (or $\lambda_{H^{1,1}}(\nu)$) by

(5.5)
$$\lambda_{H^{1,1}} = \lim_{n \to +\infty} \frac{1}{n} \int \log(\|(f_{\omega}^n)^*\|) d\nu^{\mathbf{N}}(\omega)$$

(5.6)
$$= \lim_{n \to +\infty} \frac{1}{n} \log \left\| (f_{\omega}^n)^* \right\|$$

where the second equality holds almost surely, i.e. for $\nu^{\mathbf{N}}$ -almost every $\omega \in \Omega$. This convergence follows from Kingman's subadditive ergodic theorem, since $\|\cdot\|$ being an operator norm, $(\omega, n) \mapsto \log(\|(f_{\omega}^n)^*\|)$ defines a subadditive cocycle (see [8, Thm 4.28] or [13, Thm I.4.1]).

Proposition 5.2. Let (X, ν) be a non-elementary holomorphic dynamical system on a compact Kähler surface, satisfying the moment condition (4.1), or more generally (5.4). Then the cohomological Lyapunov exponent $\lambda_{H^{1,1}}$ is positive and the other Lyapunov exponents of the linear action on $H^{1,1}(X, \mathbf{R})$ are $-\lambda_{H^{1,1}}$, with multiplicity 1, and 0, with multiplicity $h^{1,1}(X) - 2$.

Proof. Consider the Γ_{ν} -invariant decomposition $\Pi_{\Gamma_{\nu}} \oplus \Pi_{\Gamma_{\nu}}^{\perp}$ given by Proposition 2.10 and Equation (2.8). Since the intersection form is negative definite on $\Pi_{\Gamma_{\nu}}^{\perp}$, the group $\Gamma_{\nu}^{*}|_{\Pi_{\Gamma_{\nu}}^{\perp}}$ is bounded and all Lyapunov exponents of $\Gamma_{\nu}^{*}|_{\Pi_{\Gamma_{\nu}}^{\perp}}$ vanish. The linear action of Γ_{ν} on $\Pi_{\Gamma_{\nu}}$ is strongly irreducible and non-elementary, hence not relatively compact. Therefore Furstenberg's theorem asserts that $\lambda_{H^{1,1}} > 0$ (see e.g. [13, Thm III.6.3] or [8, Cor 4.32]). The remaining properties of the Lyapunov spectrum on $\Pi_{\Gamma_{\nu}}$ follow from the KAK decomposition in $O_{1,m}^{+}(\mathbf{R})$, with $1 + m = \dim(\Pi_{\Gamma_{\nu}})$ (see Lemma 2.4).

Lemma 5.3. If $a \in H^{1,1}(X; \mathbf{R})$ satisfies $a^2 > 0$, for instance if a is a Kähler class, then

$$\lim_{n \to +\infty} \frac{1}{n} \log |(f_{\omega}^n)^* a| = \lambda_{H^{1,1}}$$

for $\nu^{\mathbf{N}}$ -almost every ω .

Proof. Corollary 2.5 implies that if $a \in \mathbb{H}_X$ then for every $f \in Aut(X)$, $|f^*a| \simeq ||f^*||$, where the implied constants depend only on a. Thus the result follows from Equation (5.6).

If the order of compositions is reversed (which is less natural from the point of view of iterated pull-backs), then Lemma 5.3 indeed holds for any a in $\Pi_{\Gamma_{\nu}}$ (see [13, Cor. III.3.4.i]):

Lemma 5.4. For any $a \in \Pi_{\Gamma_{\nu}} \setminus \{0\}$ and for $\nu^{\mathbf{N}}$ -almost every $\omega = (f_n)_{n \ge 0} \in \Omega$ we have $\lim_{n \to +\infty} \frac{1}{n} \log |f_n^* \cdots f_1^* a| = \lambda_{H^{1,1}}.$

5.3. The measure μ_{∂} . By Furstenberg's theory the linear projective action of the random dynamical system (X, ν) on $\mathbb{P}\Pi_{\Gamma_{\nu}} \subset \mathbb{P}H^{1,1}(X; \mathbf{R})$ admits a unique stationary measure $\mu_{\mathbb{P}\Pi_{\Gamma_{\nu}}}$; this measure does not charge any proper projective subspace of $\mathbb{P}\Pi_{\Gamma_{\nu}}$. Recall that the mass of a class *a* is defined by $\mathbf{M}(a) = \langle a | [\kappa_0] \rangle$ (see § 2.2).

Lemma 5.5. For $\nu^{\mathbf{N}}$ -almost every ω , there exists a unique nef class $e(\omega)$ such that $\mathbf{M}(e(\omega)) = 1$ and

(5.7)
$$\frac{1}{\mathbf{M}((f_{\omega}^{n})^{*}a)}(f_{\omega}^{n})^{*}a \xrightarrow[n \to \infty]{} e(\omega)$$

for any pseudo-effective class a with $a^2 > 0$ (in particular for any Kähler class). In addition, the class $e(\omega)$ is almost surely isotropic and $\mathbb{P}(e(\omega))$ is a point of the limit set $\operatorname{Lim}(\Gamma_{\nu}) \subset \partial \mathbb{H}_X$.

Before starting the proof, note that $\Gamma_{\nu}^*|_{\Pi_{\Gamma_{\nu}}}$ is proximal in the sense of [8, §4.1]; equivalently, $\Gamma_{\nu}^*|_{\Pi_{\Gamma_{\nu}}}$ is contracting in the sense of [13, Def III.1.3]. In other words, there are sequences of elements $g_n \in \Gamma_{\nu}$ such that $||g_n^*||^{-1}g_n^*|_{\Pi_{\Gamma_{\nu}}}$ converges to a matrix of rank 1: for instance one can take $g_n = f^n$, where $f \in \Gamma_{\nu}$ is any loxodromic automorphism.

Proof. For $f \in Aut(X)$, we use the notation \underline{f}^* for its action on $\mathbb{P}H^{1,1}(X; \mathbf{R})$. Since the action of Γ_{ν} on $\Pi_{\Gamma_{\nu}}$ is strongly irreducible and proximal, its projective action satisfies the following contraction property (see [13, Thm III.3.1]): there is a measurable map $\omega \in \Omega \mapsto \underline{e}(\omega) \in \mathbb{P}\Pi_{\Gamma_{\nu}}$ such that for almost every ω , any cluster value $L(\omega)$ of

(5.8)
$$\frac{1}{\|f_0^* \cdots f_n^*\|} f_0^* \cdots f_n^*$$

in $\operatorname{End}(\Pi_{\Gamma_{\nu}})$ is an endomorphism of rank 1 whose range is equal to $\mathbf{R}\underline{e}(\omega)$.

Let $e(\omega)$ be the unique vector of mass 1 in the line $\mathbf{R}\underline{e}(\omega)$. If $a \in \Pi_{\Gamma_{\nu}}$ satisfies $a^2 > 0$ and $\mathbf{M}(a) > 0$, then any cluster value of $\mathbf{M}((f_{\omega}^n)^*a)^{-1}(f_{\omega}^n)^*a$ must coincide with $e(\omega)$ because by Corollary 2.5 the mass $\mathbf{M}((f_{\omega}^n)^*a)$ is comparable to the norm $||f_0^* \cdots f_n^*||$. Thus, the convergence (5.7) is satisfied. Furthermore $e(\omega)$ is nef, because we can apply this convergence to a nef class a and $\operatorname{Aut}(X)$ preserves the nef cone. Also, $e(\omega)$ belongs to $\operatorname{Lim}(\Gamma_{\nu})$, hence it is isotropic. Now, let a and a' be two classes of \mathbb{H}_X with $a \in \Pi_{\Gamma_{\nu}}$. Since the hyperbolic distance between $(f_{\omega}^n)^*(a)$ and $(f_{\omega}^n)^*(a')$ remains constant and the convergence (5.7) holds for a, it also holds for a'. This concludes the proof, for every class with positive self-intersection is proportional to a unique class in \mathbb{H}_X .

Here is a summary of the properties of the stationary measure $\mu_{\mathbb{P}\Pi_{\Gamma_{\nu}}}$; from now on, we view it as a measure on $\mathbb{P}H^{1,1}(X;\mathbf{R})$ and rename it as μ_{∂} because it is supported on $\partial \mathbb{H}_X$.

Theorem 5.6. The probability measure defined on $\mathbb{P}H^{1,1}(X; \mathbf{R})$ by

(5.9)
$$\mu_{\partial} = \int \delta_{\mathbb{P}(e(\omega))} \, d\nu^{\mathbf{N}}(\omega)$$

is ν -stationary and ergodic. It is the unique stationary measure on $\mathbb{P}H^{1,1}(X; \mathbf{R})$ such that $\mu_{\partial}(\mathbb{P}(\Pi_{\Gamma_{\nu}}^{\perp})) = 0$. The measure μ_{∂} has no atoms and is supported on $\operatorname{Lim}(\Gamma_{\nu})$; in particular, if $\Lambda' \subset \operatorname{Lim}(\Gamma_{\nu})$ is such that $\mu_{\partial}(\Lambda') > 0$ then Λ' is uncountable.

The top Lyapunov exponent satisfies the so-called Furstenberg formula:

(5.10)
$$\lambda_{H^{1,1}} = \int \log\left(\frac{|f^*\tilde{u}|}{|\tilde{u}|}\right) d\nu(f) d\mu_{\partial}(u),$$

where $\tilde{u} \in H^{1,1}(X, \mathbf{R}) \setminus \{0\}$ denotes any lift of $u \in \text{Lim}(\Gamma_{\nu}) \subset \mathbb{P}H^{1,1}(X, \mathbf{R})$.

Proof. The ergodicity of $\mu_{\partial} = \mu_{\mathbb{P}\Pi_{\Gamma_{\nu}}}$ as well as its representation (5.9) follow from the properties of the action of Γ_{ν} on $\mathbb{P}(\Pi_{\Gamma})$ (see [13, Chap. III]). Also, we know that $\lambda_{H^{1,1}}$ is equal to the top Lyapunov exponent of the restriction of the action to $\mathbb{P}(\Pi_{\Gamma_{\nu}})$, so the formula (5.10) follows from the strongly irreducible case (see [13, Cor III.3.4]).

Let now μ be a stationary measure on $\mathbb{P}H^{1,1}(X; \mathbf{R})$ such that $\mu(\mathbb{P}\Pi_{\Gamma_{\nu}}^{\perp}) = 0$. A martingale convergence argument shows that $(\underline{f}_{\omega}^{n})^{*}\mu$ converges to some measure μ_{ω} for almost every ω (see [13, Lem. II.2.1]). Since Γ_{ν} preserves the decomposition $\Pi_{\Gamma_{\nu}} \oplus \Pi_{\Gamma_{\nu}}^{\perp}$ and $\|(f_{\omega}^{n})^{*}\|$ tends to infinity while $\|(f_{\omega}^{n})^{*}\|_{\Pi_{\Gamma_{\nu}}^{\perp}}\|$ stays uniformly bounded, we get that $(f_{\omega}^{n})^{*}u$ converges to $\mathbb{P}\Pi_{\Gamma_{\nu}}$ for μ -almost every u and $\nu^{\mathbf{N}}$ -almost every ω ; thus μ_{ω} is almost surely supported on $\mathbb{P}\Pi_{\Gamma_{\nu}}$. Since by stationarity $\mu = \int \mu_{\omega} d\nu^{\mathbf{N}}(\omega)$ we conclude that μ gives full mass to $\mathbb{P}(\Pi_{\Gamma_{\nu}})$, hence $\mu = \mu_{\partial}$. \Box **Remark 5.7.** If $\operatorname{Supp}(\nu)$ generates Γ_{ν} as a semi-group, then $\operatorname{Supp}(\mu_{\partial}) = \operatorname{Lim}(\Gamma_{\nu})$, otherwise the inclusion can be strict: take a Schottky group $\Gamma = \langle f, g \rangle \subset \operatorname{PSL}(2, \mathbf{R})$ and $\nu = (\delta_f + \delta_g)/2$. **Remark 5.8.** Since $\operatorname{Lim}(\Gamma_{\nu}) \subset \operatorname{Psef}(X)$, for every $u \in \operatorname{Lim}(\Gamma_{\nu})$ there exists a unique \tilde{u} such

Remark 5.8. Since $\operatorname{Lim}(\Gamma_{\nu}) \subset \operatorname{Pset}(X)$, for every $u \in \operatorname{Lim}(\Gamma_{\nu})$ there exists a unique \tilde{u} such that $\mathbb{P}\tilde{u} = u$ and $\langle \tilde{u} | [\kappa_0] \rangle = \mathbf{M}(\tilde{u}) = 1$. Then the following formula holds:

(5.11)
$$\lambda_{H^{1,1}} = \int \log \left(\mathbf{M}(f^*\tilde{u}) \right) \, d\nu(f) \, d\mu_{\partial}(u) = \int \log \left(\frac{\mathbf{M}(f^*\tilde{u})}{\mathbf{M}(\tilde{u})} \right) \, d\nu(f) \, d\mu_{\partial}(u) \, d\mu_{\partial}(u$$

Indeed set $r(w) = \mathbf{M}(w)/|w|$. On the limit set this function satisfies $1/C \leq r(\tilde{u}) \leq C$, where C is the positive constant from Equation (2.4). Then, for all $m \ge 1$, the stationarity of μ_{∂} implies

$$\int \log\left(\frac{r(f^*\tilde{u})}{r(\tilde{u})}\right) d\nu(f) d\mu_{\partial}(u) = \int \log\left(\frac{r(f_m^* \cdots f_0^*\tilde{u})}{r(f_{m-1}^* \cdots f_0^*\tilde{u})}\right) d\nu(f_m) \cdots d\nu(f_0) d\mu_{\partial}(u).$$

Summing from m = 0 to n - 1, telescoping the sum, and dividing by n gives

$$\int \log\left(\frac{r(f^*\tilde{u})}{r(\tilde{u})}\right) d\nu(f) d\mu_{\partial}(u) = \frac{1}{n} \int \log\left(\frac{r(f^*_{n-1}\cdots f^*_0\tilde{u})}{r(\tilde{u})}\right) d\nu(f_{n-1})\cdots d\nu(f_0) d\mu_{\partial}(u).$$

Finally since $1/C \leq r \leq C$, the right hand side tends to zero as $n \to \infty$. Hence the integral of $\log(r \circ f^*/r)$ vanishes, and (5.11) follows from Furstenberg's formula.

Proposition 5.9. The point $\mathbb{P}(e(\omega))$ is $\nu^{\mathbf{N}}$ -almost surely extremal in $\mathbb{P}(\overline{\operatorname{Kah}}(X))$ and in $\mathbb{P}(\operatorname{Psef}(X))$.

Proof. The class $e(\omega)$ almost surely belongs to $\overline{\operatorname{Kah}}(X)$ and to the isotropic cone. By the Hodge index theorem –more precisely, by the case of equality in the reverse Schwarz Inequality (2.5)– $e(\omega)$ cannot be a non-trivial convex combination of classes with non-negative intersection and mass 1; so $\mathbb{P}(e(\omega))$ is an extremal point of the convex set $\mathbb{P}(\overline{\operatorname{Kah}}(X)) \subset \mathbb{P}H^{1,1}(X; \mathbf{R})$.

From Proposition 2.3, there are at most countably many points $\mathbb{P}(u)$ in $\mathbb{P}(\overline{\operatorname{Kah}}(X))$ such that $u^2 = 0$ and $\mathbb{P}(u)$ is not extremal in $\mathbb{P}(\operatorname{Psef}(X))$. Therefore the second assertion follows from the fact that μ_{∂} is atomless.

5.4. Some estimates for random products of matrices. The aim of this section is to establish some technical facts which will play a crucial role in our study of the closed positive currents T^s_{ω} in Section 6. The key results are Theorem 5.10 and Lemma 5.12.

5.4.1. Sequences of good times. Let us describe a theorem of Gouëzel and Karlsson, specialized to our context. Fix a point e_0 in \mathbb{H}_X , for instance $e_0 = [\kappa_0]$ with κ_0 a fixed Kähler form, as in Section 2.2. Consider the two functions of $(n, \omega) \in \mathbf{N} \times \Omega$ defined by

(5.12)
$$T(n,\omega) = d_{\mathbb{H}}(e_0, (f_{\omega}^n)^* e_0), \quad N(n,\omega) = \log \|(f_{\omega}^n)^*\|.$$

They satisfy the subadditive cocycle property

(5.13)
$$a(n+m,\omega) \leq a(n,\omega) + a(m,\sigma^n(\omega))$$

where σ is the unilateral shift on Ω (see § 4.3). Let $a(n, \omega)$ be such a subadditive cocycle; if $a(1, \omega)$ is integrable the asymptotic average is defined to be the limit

(5.14)
$$A = \lim_{n \to +\infty} \frac{1}{n} \int a(n,\omega) \, d\nu^{\mathbf{N}}(\omega);$$

it exists in $[-\infty, +\infty)$, and we say it is finite if $A \neq -\infty$. From Theorem 5.6, Remark 5.8, and Corollary 2.5, the asymptotic average of the cocycles T and N are both equal to $\lambda_{H^{1,1}}$.

Following [53], we say that a subadditive cocycle $a(n, \omega)$ is **tight along the sequence of** positive integers (n_i) if there is a sequence of real numbers $(\delta_\ell) = (\delta_\ell(\omega))_{\ell \ge 0}$ such that

- (i) δ_{ℓ} converges to 0 as ℓ goes to $+\infty$;
- (ii) for every *i*, and for every $0 \le \ell \le n_i$, $|a(n_i, \omega) a(n_i \ell, \sigma^{\ell}(\omega)) A\ell| \le \ell \delta_{\ell}$;
- (iii) for every *i* and for every $0 \le \ell \le n_i$, $a(n_i, \omega) a(n_i \ell, \omega) \ge (A \delta_\ell)\ell$.

Theorem 5.10 (Gouëzel and Karlsson [53]). Let $a(n, \omega)$ be an ergodic subadditive cocycle, with a finite asymptotic average A. Then, for almost every ω , the cocycle is tight along a subsequence $(n_i(\omega))$ of positive upper density.

Recall that the (asymptotic) upper density of a subset S of N is the non-negative number defined by $\overline{\text{dens}}(S) = \limsup_{k \to +\infty} \left(\frac{1}{k} |S \cap [0, k-1]|\right)$. A sequence $(n_i)_{i \ge 0}$ is said to have positive upper density if the set of its values $S = \{n_i : i \ge 0\}$ satisfies $\overline{\text{dens}}(S) > 0$.

Proof. Let us explain how this result follows from [53]. First, fix a small positive real number $\rho > 0$, and apply Theorem 1.1 and Remark 1.2 of [53] to get a set Ω_{ρ} of measure $1 - \rho$ such that the first two properties (i) and (ii) are satisfied for every $\omega \in \Omega_{\rho}$ with respect to a sequence (δ_{ℓ}) that does not depend on ω , and for a sequence of times $(n_i(\omega))$ of upper density $\geq 1 - \rho$. To get (iii), we apply Lemma 2.3 of [53] to the sub-additive cocycle $a(n, \omega)$ (not to the cocycle $b(n, \omega) = a(n, \sigma^{-n}(\omega))$ as done in [53]). For every $\varepsilon > 0$, there is a subset $\Omega'_{\varepsilon} \subset \Omega$ and a sequence $(\delta'_{\ell})_{\ell \geq 0}$ such that

- (a) $\nu^{\mathbf{N}}(\Omega'_{\varepsilon}) > 1 \varepsilon$, and δ'_{ℓ} converges towards 0 as ℓ goes to $+\infty$;
- (b) for every $\omega \in \Omega'_{\varepsilon}$, there is a set of bad times $B(\omega) \subset \mathbf{N}$ such that for every $k \ge 0$ $|B(\omega) \cap [0, k-1]| \le \varepsilon k$, and for every $n \notin B(\omega)$ and every $0 \le \ell \le n$,

$$a(n,\omega) - a(n-\ell,\omega) \ge (A-\delta'_{\ell})\ell.$$

If ω belongs to $\Omega_{\rho} \cap \Omega'_{\varepsilon}$, the set of indices *i* for which $n_i(\omega) \notin B(\omega)$ is infinite. More precisely, the set $S(\omega) = \{n_j(\omega) ; n_j(\omega) \notin B(\omega)\}$ has asymptotic upper density $\ge 1 - \rho - \varepsilon$. Along

this subsequence, the three properties (i), (ii), and (iii) are satisfied. Since this holds for all $\omega \in \Omega'_{\varepsilon} \cap \Omega_{\rho}$ and the measure of this set is $\ge 1 - \rho - \varepsilon$, this holds for $\nu^{\mathbf{N}}$ -almost every ω . \Box

Corollary 5.11. For $\nu^{\mathbf{N}}$ -almost every $\omega \in \operatorname{Aut}(X)^{\mathbf{N}}$, there is an increasing sequence of integers $(n_i(\omega))$ going to $+\infty$ and a real number $A(\omega)$ such that

$$\sum_{j=0}^{n_{i}(\omega)} \frac{\|(f_{\omega}^{j})^{*}\|}{\|(f_{\omega}^{n_{i}(\omega)})^{*}\|} \leq A(\omega) \text{ and } \sum_{j=0}^{n_{i}(\omega)} \frac{\|(f_{\sigma^{j}(\omega)}^{n_{i}(\omega)-j})^{*}\|}{\|(f_{\omega}^{n_{i}(\omega)})^{*}\|} \leq A(\omega)$$

for all indices $i \ge 0$.

Proof. Apply Theorem 5.10 to the subadditive cocyle $N(n, \omega)$ and note that

(5.15)
$$\sum_{j=0}^{n_i(\omega)} \frac{\|(f_{\omega}^j)^*\|}{\|(f_{\omega}^{n_i(\omega)})^*\|} = \sum_{\ell=0}^{n_i(\omega)} \frac{\|(f_{\omega}^{n_i-\ell})^*\|}{\|(f_{\omega}^{n_i})^*\|} = \sum_{\ell=0}^{n_i(\omega)} \frac{e^{N(n_i-\ell,\omega)}}{e^{N(n_i,\omega)}} \leq \sum_{\ell=0}^{n_i(\omega)} e^{-\ell(\lambda_{H^{1,1}}-\delta_{\ell})}$$

which is bounded as $n_i(\omega) \to \infty$. The second estimate is similar.

5.4.2. A mass estimate for pull-backs. Assume that (X, ν) is non-elementary and satisfies the condition (4.1). Recall from Lemma 5.4 that $\mathbf{M}((f_{\omega}^{n})^{*}a)^{-1}(f_{\omega}^{n})^{*}a$ converges to the pseudo-effective class $e(\omega)$ for almost every ω and every Kähler class a. Thus, on a set of total $\nu^{\mathbf{N}}$ -measure, this convergence holds for all $\sigma^{k}(\omega)$, $k \ge 0$. Since $\mathbf{M}(e(\omega)) = 1$, we obtain

(5.16)
$$f_0^* e(\sigma \omega) = \mathbf{M}(f_0^* e(\sigma \omega)) e(\omega);$$

more generally, for every $k \ge 1$,

(5.17)
$$(f_{\omega}^k)^* e(\sigma^k \omega) = \mathbf{M}((f_{\omega}^k)^* e(\sigma^k \omega)) e(\omega).$$

Lemma 5.12. For $\nu^{\mathbf{N}}$ -almost every ω , we have $\frac{1}{n} \log \mathbf{M}((f_{\omega}^n)^* e(\sigma^n \omega)) \xrightarrow[n \to \infty]{} \lambda_{H^{1,1}}$.

This does *not* follow from Lemma 5.3 because $e(\sigma^n \omega)$ depends on *n*.

Proof. For almost every ω , for every $k \ge 1$, and for every Kähler class a, $e(\sigma^k \omega)$ is the limit of $\mathbf{M}(f_k^* \cdots f_{n-1}^* a)^{-1} f_k^* \cdots f_{n-1}^* a$ as n goes to $+\infty$. So

(5.18)
$$f_0^* \cdots f_{k-1}^* e(\sigma^k(\omega)) = \left(\lim_{n \to \infty} \frac{\mathbf{M}(f_0^* \cdots f_{n-1}^* a)}{\mathbf{M}(f_k^* \cdots f_{n-1}^* a)}\right) e(\omega) =: \zeta(k, \omega) e(\omega)$$

where $\zeta(k,\omega)$ is both equal to $\mathbf{M}((f_{\omega}^k)^* e(\sigma^k(\omega)))$ and to the limit

(5.19)
$$\zeta(k,\omega) = \lim_{n \to \infty} \frac{\mathbf{M}(f_0^* \cdots f_{n-1}^* a)}{\mathbf{M}(f_k^* \cdots f_{n-1}^* a)} = \lim_{n \to \infty} \frac{\mathbf{M}((f_\omega^n)^* a)}{\mathbf{M}((f_{\sigma^k(\omega)}^{n-k})^* a)}$$

We want to show that, $\nu^{\mathbf{N}}$ -almost surely, $(1/k) \log \zeta(k, \omega)$ converges to $\lambda_{H^{1,1}}$.

Before starting the proof, note that ζ is a multiplicative cocycle: $\zeta(k,\omega) = \prod_{\ell=1}^{k} \zeta(1,\sigma^{\ell}\omega)$; in particular, $\log \zeta(k,\omega)$ is equal to the Birkhoff sum $\sum_{\ell=1}^{k} \log \zeta(1,\sigma^{\ell}\omega)$. Since

(5.20)
$$C^{-1} \| (f_0^{-1})^* \|_{H^{1,1}} \leq \mathbf{M}(f_0^* e(\sigma(\omega))) \leq C \| f_0^* \|_{H^{1,1}},$$

our moment condition shows that $\log(\zeta(1,\omega))$ is integrable. So, by the ergodic theorem of Birkhoff, $\lim_k \frac{1}{k} \log \zeta(k,\omega)$ exists $\nu^{\mathbf{N}}$ -almost surely.

Pick a sequence (n_i) of good times for ω , as in Theorem 5.10. If we compute the limit in Equation (5.19) along the subsequence (n_i) we see that $\zeta(k,\omega) \ge C \exp((\lambda_{H^{1,1}} - \delta(k))k)$ for some constant C > 0, and some sequence $\delta(k)$ converging to 0 as k goes to $+\infty$. This gives

(5.21)
$$\limsup_{k \to +\infty} \frac{1}{k} \log \zeta(k, \omega) \ge \lambda_{H^{1,1}}.$$

Now, consider the linear cocycle $\Upsilon : \Omega \times H^{1,1}(X, \mathbf{R}) \to \Omega \times H^{1,1}(X, \mathbf{R})$ defined by

(5.22)
$$\Upsilon(\omega, u) = (\sigma(\omega), (f_{\omega}^{1})_{*}u)$$

and let $\mathbb{P}\Upsilon$ be the associated projective cocycle on $\Omega \times \mathbb{P}H^{1,1}(X, \mathbf{R})$. The Lyapunov exponents of Υ are $\pm \lambda_{H^{1,1}}$, each with multiplicity 1, and 0, with multiplicity $h^{1,1}(X) - 2$. Since $\mathbb{P}((f_{\omega}^{1})^*e(\sigma(\omega))) = \mathbb{P}(e(\omega))$, the measurable section $\{(\omega, \mathbb{P}(e(\omega))) : \omega \in \Omega\}$ is $\mathbb{P}\Upsilon$ -invariant. Therefore, by ergodicity of σ with respect to $\nu^{\mathbf{N}}$, $m = \int \delta_{\mathbb{P}(e(\omega))} d\nu^{\mathbf{N}}(\omega)$ defines an invariant and ergodic measure for $\mathbb{P}\Upsilon$. It follows from the invariance of the decomposition into characteristic subspaces in Oseledets' theorem that $e(\omega)$ is contained in a given characteristic subspace, we get (as in Remark 5.8) that

$$\lambda = \int \log \frac{\left| (f_{\omega}^{1})_{*} u \right|}{|u|} \, dm(\omega, u) = \int \log \frac{\mathbf{M}((f_{\omega}^{1})_{*}(e(\omega)))}{\mathbf{M}(e(\omega))} \, d\nu^{\mathbf{N}}(\omega) = \int \log \zeta(1, \omega)^{-1} \, d\nu^{\mathbf{N}}(\omega)$$

(see Ledrappier [64, §1.5]). Birkhoff's ergodic theorem implies that $\lim_k \frac{1}{k} \log \zeta(k, \omega) = -\lambda$, with $\lambda \in \{\pm \lambda_{H^{1,1}}, 0\}$, therefore the Inequality (5.21) concludes the proof.

5.4.3. *Exponential moments*. The result of this section will only be used in Theorem 6.16 so this paragraph may be skipped on a first reading. Consider the exponential moment condition

(5.23)
$$\exists \tau > 0, \ \int \left(\|f\|_{C^1} + \|f^{-1}\|_{C^1} \right)^{\tau} \ d\nu(f) < +\infty$$

As in Section 5.1, this upper bound implies the cohomological moment condition

(5.24)
$$\exists \tau > 0, \ \int \left(\|f^*\|_{H^{1,1}} + \|(f^{-1})^*\|_{H^{1,1}} \right)^{\tau} d\nu(f) < +\infty.$$

Proposition 5.13. Assume that ν satisfies the Condition (5.23). Let $D: \operatorname{Aut}(X) \to \mathbf{R}_+$ be a measurable function such that $\int D(f)^{\tau'} d\nu(f) < \infty$ for some $\tau' > 0$. Then, there is a measurable function $B: \Omega \to \mathbf{R}_+$ satisfying

$$\int \log^+(B(\omega)) \, d\nu^{\mathbf{N}}(\omega) < \infty$$

such that for $\nu^{\mathbf{N}}$ -almost every $\omega = (f_n)$ and every $n \ge 0$

$$\sum_{j=1}^{n-1} D(f_{j-1}) \frac{\|f_j^* \cdots f_{n-1}^*\|}{\|f_0^* \cdots f_{n-1}^*\|} \leq B(\omega), \text{ and } \sum_{j=1}^{n-1} D(f_j) \frac{\|f_0^* \cdots f_{j-1}^*\|}{\|f_0^* \cdots f_{n-1}^*\|} \leq B(\omega).$$

This is a refined version of Corollary 5.11.

Proof. We are grateful to Sébastien Gouëzel for explaining this argument to us. We temporarily use the notation $\mathbb{P}(\cdot)$ for probability with respect to ν^n or ν^N (instead of projectivisation).

First Estimate. We start with the first estimate: $\sum_{j=1}^{n-1} D(f_{j-1}) \frac{\|f_j^* \cdots f_{n-1}^*\|}{\|f_0^* \cdots f_{n-1}^*\|} \leq B(\omega).$

Step 1.– For every $0 < \varepsilon < \lambda_{H^{1,1}}$ there exists constants c, C > 0 such that

(5.25)
$$\mathbb{P}\left(\left|(f_{\omega}^{n})^{*}b\right| \leq e^{\varepsilon n}\right) \leq Ce^{-cn}$$

for every $b \in \Pi_{\Gamma}$ with |b| = 1. This large deviation result, which is uniform in *n* and *b*, follows from condition (5.24) (see for instance [13, §V.6], and [8, §12]).

Step 2.– Let us prove that

(5.26)
$$\mathbb{P}\left(\frac{\|f_{j}^{*}\cdots f_{n-1}^{*}\|}{\|f_{0}^{*}\cdots f_{n-1}^{*}\|} > e^{-\varepsilon j}\right) \leq Ce^{-cj}.$$

For this, fix f_j, \ldots, f_{n-1} . Then, there is a point $a \in \Pi_{\Gamma}$ with |a| = 1 such that $\left\| f_j^* \cdots f_{n-1}^* \right\| = \left\| f_j^* \cdots f_{n-1}^* \right\| < \left\| f_j^* \cdots f_{n-1}^* \right\| < \| f_j^* \cdots f_{n-1}^* \| e^{\varepsilon j}$, we infer that

(5.27)
$$|f_0^* \cdots f_{n-1}^*a| < ||f_j^* \cdots f_{n-1}^*||e^{\varepsilon j} = |f_j^* \cdots f_{n-1}^*a|e^{\varepsilon j}.$$

Thus, if we set

(5.28)
$$b = \frac{1}{\left|f_{j}^{*}\cdots f_{n-1}^{*}a\right|}f_{j}^{*}\cdots f_{n-1}^{*}a,$$

we obtain that $|f_0^* \cdots f_{j-1}^* b| < e^{\varepsilon j}$; this happens with (conditional) probability $\leq Ce^{-cj}$ (relative to ν^{*j}), for the uniform constants given in Step 1. Averaging over f_j, \ldots, f_{n-1} , we get the result.

Step 3.– The moment condition satisfied by D and Markov's inequality imply $\mathbb{P}(D > K) \leq C_1 K^{-\tau'}$ for some constant $C_1 > 0$. Fix $\varepsilon \in \mathbf{R}^*_+$ small with respect to $\lambda_{H^{1,1}}$ and τ' . Then, on a set $\Omega(\varepsilon, J)$ of measure

(5.29)
$$\nu^{\mathbf{N}}(\Omega(\varepsilon,J)) \ge 1 - C_2(e^{-(\varepsilon\tau'/2)J} + e^{-\varepsilon cJ}),$$

for some $C_2 = C_2(\varepsilon) > 0$, we have both $D(f_{j-1}) \leq e^{\varepsilon j/2}$ and $\frac{\|f_j^* \cdots f_{n-1}^*\|}{\|f_0^* \cdots f_{n-1}^*\|} \leq e^{-\varepsilon j}$ for all $j \geq J$. For $\omega = (f_n)$ in $\Omega(\varepsilon, J)$, we get

(5.30)
$$\sum_{j=1}^{n-1} D(f_{j-1}) \frac{\left\|f_{j}^{*} \cdots f_{n-1}^{*}\right\|}{\left\|f_{0}^{*} \cdots f_{n-1}^{*}\right\|} \leqslant \sum_{j=1}^{J} D(f_{j-1}) \frac{\left\|f_{j}^{*} \cdots f_{n-1}^{*}\right\|}{\left\|f_{0}^{*} \cdots f_{n-1}^{*}\right\|} + \sum_{j=J+1}^{n-1} e^{-\varepsilon j/2}$$
$$\leqslant \sum_{j=1}^{J} D(f_{j-1}) \left\|(f_{j-1}^{-1})^{*} \cdots (f_{0}^{-1})^{*}\right\| + C_{3}$$
$$= C_{3} + \sum_{j=0}^{J-1} \left\|f_{0}^{*}\right\| \cdots \left\|f_{j}^{*}\right\| D(f_{j}).$$

The moment condition (5.23) gives $\mathbb{P}(||f^*|| > K) \leq C_4 K^{-\tau}$ and as already noticed, we also have $\mathbb{P}(D(f) > K) \leq C_1 K^{-\tau'}$. So, with $\eta = \min(\tau, \tau')$, there is a set of probability at least $1 - C_5 J K^{-\eta}$ on which

(5.31)
$$\sum_{j=0}^{J-1} D(f_j) \|f_0^*\| \cdots \|f_j^*\| \leq C_6 J K^{J+2}.$$

Taking $K = J^{3/\eta}$, we have $JK^{-\eta} = J^{-2}$, and we obtain

(5.32)
$$\mathbb{P}\left(\sum_{j=0}^{J-1} D(f_j) \| f_0^* \| \cdots \| f_j^* \| > J^{1+3(J+2)/\eta}\right) \leq C_7 J^{-2}.$$

Also, note that $J^{1+(3J+6)/\eta} \leq \exp(CJ^{3/2})$.

By the Borel-Cantelli lemma, the sum in (5.30) is almost surely bounded by some constant $B(\omega)$ which satisfies $\mathbb{P}(\log B > J^{3/2}) \leq CJ^{-2}$; in particular $\mathbb{E}(\log^+ B) < \infty$.

Second Estimate.– To obtain the second estimate of Proposition 5.13, we apply the above proof to the reversed random dynamical system, induced by $\check{\nu} : f \mapsto \nu(f^{-1})$. Indeed, the core of the argument is the inequality (5.30) which is not sensitive to the order of compositions. \Box

6. LIMIT CURRENTS

In this section, we establish the counterpart of the convergence (5.7) at the level of closed positive currents on X. Throughout this section we fix a non-elementary random holomorphic dynamical system (X, ν) satisfying the moment condition (4.1), so that all results of §5 apply. We refer the reader to [38] and [56] (in particular Chapter 8) for basics on pluripotential theory.

6.1. Potentials and cohomology classes of positive closed currents. Let us fix once and for all a family of Kähler forms $(\kappa_i)_{1 \le i \le h^{1,1}(X)}$ such that $[\kappa_i]^2 = 1$ and the $[\kappa_i]$ form a basis of $H^{1,1}(X; \mathbf{R})$; in addition we require that the κ_i satisfy

(6.1)
$$\kappa_0 = \beta \sum_i \kappa_i$$

for some $\beta > 0$, where κ_0 is the Kähler form chosen in Section 2.2 (note that necessarily $\beta < 1$). We also fix a smooth volume form vol_X on X, normalized by $\int_X \operatorname{vol} = 1$. On tori, K3 and Enriques surfaces, we choose vol_X to be the canonical $\operatorname{Aut}(X)$ -invariant volume form (see Remark 3.2). It is convenient to assume in all cases that vol_X is also the volume form associated with the Kähler metric κ_0 (up to scaling).

Unless otherwise specified, the currents we shall consider will be of type (1, 1). The action of a current T on a test form φ will be denoted by $\langle T, \varphi \rangle$ or $\int T \wedge \varphi$. If T is closed, we denote its cohomology class by [T]; so, if φ is a closed form, $\langle T, \varphi \rangle = \langle [T] | [\varphi] \rangle$. By definition the **mass** of a current is the quantity $\mathbf{M}(T) = \int T \wedge \kappa_0$; so $\mathbf{M}(T) = \langle [T] | [\kappa_0] \rangle$ when T is closed.

6.1.1. Normalized potentials. If a is an element of $H^{1,1}(X; \mathbf{R})$, we denote by $(c_i(a))_{1 \leq i \leq h^{1,1}(X)}$ its coordinates in the basis $([\kappa_i])$, so that $a = \sum_i c_i(a)[\kappa_i]$. Then, we set

(6.2)
$$\Theta(a) = \sum_{i} c_i(a) \kappa_i.$$

Likewise, given a closed (1,1)-form α or a closed current of bidegree (1,1), we set $c_i(\alpha) = c_i([\alpha])$ and $\Theta(\alpha) = \Theta([\alpha])$; hence, $[\Theta(\alpha)] = [\alpha]$. It is worth keeping in mind that some coefficients $c_i(\alpha)$ can be negative and $\Theta(\alpha)$ need not be semi-positive, even if α is a Kähler form. If T is a closed positive current of bidegree (1,1) on X we define its **normalized potential** to be the unique function $u_T \in L^1(X)$ such that

(6.3)
$$T = \Theta(T) + dd^c(u_T) \text{ and } \int_X u_T \text{ vol} = 0$$

(see [56, §8.1]). The function u_T is locally given as the difference v - w of a psh potential v of T and a smooth potential w of $\Theta(T)$.

Lemma 6.1. There is a constant A > 0 such that the following properties are satisfied for every closed positive current T of mass 1

(1) $-A \leq c_i(T) \leq A$ for all $1 \leq i \leq h^{1,1}(X)$, and $-A\kappa_0 \leq \Theta(T) \leq A\kappa_0$. (2) the function u_T is $(A\kappa_0)$ -psh: $dd^c(u_T) + A\kappa_0$ is a positive current.

Proof. Since the coefficients $T \mapsto c_i(T)$ are continuous functions on the space of currents and closed positive currents of mass 1 form a compact set K, the functions $|c_i|$ are bounded by some uniform constant A' on K. Setting $A = A'\beta^{-1}$, with β as in Equation (6.1), we get $-A\kappa_0 \leq \Theta(T) \leq A\kappa_0$ for all $T \in K$. Then $dd^c u_T = T - \Theta(T) \geq -A\kappa_0$ and (2) follows. \Box

Corollary 6.2. The set of potentials $\{u_T \mid T \text{ is a closed positive current of mass 1 on } X\}$ is a compact subset of $L^1(X; vol)$.

Proof. Since this is a set of $(A\kappa_0)$ -psh functions which are normalized with respect to a smooth volume form, the result follows from Proposition 8.5 and Remark 8.6 in [56].

Remark 6.3. Another usual normalization is $\sup_{x \in X} u_T(x) = 0$; by compactness this only changes u_T by some uniformly bounded constant. Since many of our dynamical examples preserve a natural volume form it is more convenient for us to normalize as in (6.3).

6.1.2. The diameter of a pseudo-effective class. For a class $a \in Psef(X)$ we define

(6.4) $\operatorname{Cur}(a) = \{T ; T \text{ is a closed positive current with } [T] = a\},\$

This is a compact convex subset of the space of currents. If S and T are two elements of Cur(a), then $\Theta(S) = \Theta(T) = \Theta(a)$ and $T - S = dd^c(u_T - u_S)$. We set

(6.5)
$$\operatorname{dist}(S,T) = \int_X |u_S - u_T| \operatorname{vol} .$$

This is a distance that metrizes the weak topology on Cur(a): this follows for instance from the fact that by Corollary 6.2 (Cur(a), dist) is compact. By definition, the **diameter** of *a* is

If $a \in \operatorname{Psef}(X)$, then $\operatorname{Diam}(a)$ is a non-negative real number which is finite by Corollary 6.2. If $\operatorname{Cur}(a) = \emptyset$, we set $\operatorname{Diam}(a) = -\infty$. Note that Diam is homogeneous of degree 1: $\operatorname{Diam}(ta) = t \operatorname{Diam}(a)$ for every $a \in \operatorname{Psef}(X)$ and t > 0.

Lemma 6.4. On $\operatorname{Psef}(X)$, $a \mapsto \operatorname{Diam}(a)$ is upper semi-continuous, hence measurable.

Proof. Let (a_n) be a sequence of pseudo-effective classes converging to a. For every n we choose a pair of currents (S_n, T_n) in $\operatorname{Cur}(a_n)^2$ such that $\operatorname{dist}(S_n, T_n) \ge \operatorname{Diam}(a_n) - 1/n$. The masses of S_n and T_n are uniformly bounded because they depend only on a_n . By Corollary 6.2, we can extract a subsequence such that S_n and T_n converge towards closed positive currents $S, T \in \operatorname{Cur}(a)$, and u_{S_n} and u_{T_n} converge towards their respective potentials u_S and u_T in $L^1(X, \operatorname{vol})$. Then, $\operatorname{dist}(S, T) = \int_X |u_S - u_T| \operatorname{vol} = \lim_n \operatorname{dist}(S_n, T_n)$, which shows that $\operatorname{Diam}(a) \ge \limsup_n (\operatorname{Diam}(a_n))$.

6.2. Action of Aut(X).

6.2.1. A volume estimate. Let X be a compact, complex manifold, and let vol be a C^0 -volume form on X with vol(X) = 1. If f is an automorphism of X, let $Jac(f): X \to \mathbf{R}$ denote its Jacobian determinant with respect to the volume form vol: $f^*vol = Jac(f)vol$. The following lemma is a variation on well-known ideas in holomorphic dynamics (see for instance [55]).

Lemma 6.5. Let κ be a hermitian form on X. Let h be a κ -psh function on X such that $\int_X h \operatorname{vol} = 0$, and let f be an automorphism of X. Then,

$$\int_X |h \circ f| \operatorname{vol} \leqslant C \log(C \|\operatorname{Jac}(f^{-1})\|_{\infty})$$

for some positive constant C that depends on (X, κ) but neither on f nor on h.

Proof. We first observe that there is a constant c > 0 such that $vol\{|h| \ge t\} \le c \exp(-t/c)$; this follows from Lemma 8.10 and Theorem 8.11 in [56], together with Chebychev's inequality (see Remark 6.3 for the normalization). Then, we get

(6.7)
$$\int_{X} |h \circ f| \operatorname{vol} = \int_{0}^{\infty} \operatorname{vol}\{|h \circ f| \ge t\} dt$$
$$= \int_{0}^{\infty} \operatorname{vol}(f^{-1}\{|h| \ge t\}) dt$$
$$\leqslant \int_{0}^{s} \operatorname{vol}(X) dt + \|\operatorname{Jac}(f^{-1})\|_{\infty} \int_{s}^{\infty} c \exp(-t/c) dt$$
$$\leqslant s \operatorname{vol}(X) + \|\operatorname{Jac}(f^{-1})\|_{\infty} c^{2} \exp(-s/c)$$

where the inequality in the third line follows from the change of variable formula. Now, we minimize (6.8) by choosing $s = c \log(c \|\operatorname{Jac}(f^{-1})\|_{\infty}/\operatorname{vol}(X))$ and we infer that

(6.9)
$$\int_{X} |h \circ f| \operatorname{vol} \leq c \operatorname{vol}(X) \left(1 + \log \left(\frac{c \|\operatorname{Jac}(f^{-1})\|_{\infty}}{\operatorname{vol}(X)} \right) \right).$$

Since the total volume is invariant, $\|\operatorname{Jac}(f)\|_{\infty} \ge 1$, and the asserted estimate follows.

6.2.2. Equivariance. Let us come back to the study of (X, ν) . If f is an automorphism of X, then $f^* \operatorname{Cur}(a) = \operatorname{Cur}(f^*(a))$ for every class $a \in H^{1,1}(X, \mathbf{R})$. If $a \in \operatorname{Psef}(X)$ and $T \in \operatorname{Cur}(a)$, then $T = \Theta(a) + dd^c(u_T)$ and

(6.10)
$$f^*T = f^*\Theta(a) + dd^c(u_T \circ f) = \Theta(f^*a) + dd^c(u_{f^*\Theta(a)} + u_T \circ f).$$

This shows that the normalized potential of f^*T is given by

(6.11)
$$u_{f^*T} = u_{f^*\Theta(a)} + u_T \circ f + E(f,T)$$

where $E(f,T) \in \mathbf{R}$ is the constant for which the integral of u_{f^*T} vanishes; since $u_{f^*\Theta(a)}$ has mean 0, we get

(6.12)
$$E(f,T) = -\int_X \left(u_{f^*\Theta(a)} + u_T \circ f \right) \operatorname{vol} = -\int_X u_T \circ f \operatorname{vol}.$$

Remark 6.6. If vol is *f*-invariant, for instance if it is the canonical volume on a K3 or Enriques surface, then E(f,T) = 0, which simplifies a little bit the analysis of the potentials below.

Lemma 6.7. On the set of closed positive currents of mass 1, the function $(f,T) \mapsto E(f,T)$ satisfies

 $|E(f,T)| \leq C \log \left(C \| \operatorname{Jac}(f^{-1}) \|_{\infty} \right)$

where the implied positive constant C depends neither on f nor on T.

Proof. From Lemma 6.1, the potentials u_T are uniformly $(A\kappa_0)$ -psh, so the conclusion follows from Equation (6.12) and Lemma 6.5.

Lemma 6.8. There exists a constant C such that if a is any pseudo-effective class of mass 1, and f is any automorphism of X, then

$$\operatorname{Diam}(f^*a) \leq C \log \left(C \left\| \operatorname{Jac}(f^{-1}) \right\|_{\infty} \right)$$

Proof. Indeed, if S and T belong to Cur(a), by Equation (6.11) we have $u_{f^*T} - u_{f^*S} = (u_T - u_S) \circ f + E(f,T) - E(f,S)$, so

(6.13)
$$\operatorname{dist}(f^*T, f^*S) \leq \int |u_T \circ f| \operatorname{vol} + \int |u_S \circ f| \operatorname{vol} + |E(f, T)| + |E(f, S)|$$

and the result follows from Lemmas 6.5 and 6.7, since u_S and u_T are uniformly $(A\kappa_0)$ -psh. \Box

6.2.3. An estimate for canonical potentials.

Lemma 6.9. For any Kähler form κ on X there exists a positive constant $C(\kappa)$ such that for every $f \in Aut(X)$,

$$\|u_{f^{*}\kappa}\|_{C^{1}} \leq C(\kappa) \|f\|_{C^{1}}^{2}.$$

In addition $C(\kappa) \leq C' \|\kappa\|_{\infty}$, where $\|\kappa\|_{\infty}$ is the sup norm of the coefficients of κ in a system of coordinate charts, and C' depends only on X (and the choice of these coordinate charts).

Recall the choice of Kähler forms (κ_i) from § 6.1 and the definition of $\Theta(\cdot)$ from § 6.1.1.

Corollary 6.10. If $\kappa = \sum_i c_i \kappa_i$ in Lemma 6.9, then the constant $C(\kappa)$ satisfies $C(\kappa) \leq C'' \mathbf{M}(\kappa)$. Likewise, $\|u_{f^*\Theta(a)}\|_{C^1} \leq C''' \mathbf{M}(a) \|f\|_{C^1}^2$ for all $a \in \operatorname{Psef}(X)$.

Indeed
$$C(\kappa) \leq C' \|\kappa\|_{\infty} \leq C'' \sum_{i} |c_i|$$
 and $u_{f^*\Theta(a)} = \sum c_i(a) u_{f^*\kappa_i}$.

Proof of Lemma 6.9. By definition $f^*\kappa - \Theta(f^*\kappa) = dd^c(u_{f^*\kappa})$. The desired estimate will be obtained by constructing a solution ϕ to the equation

(6.14)
$$dd^c \phi = f^* \kappa - \Theta(f^* \kappa)$$

which satisfies $\|\phi\|_{C^1} \leq C \|f\|_{C^1}^2$. Then, since $u_{f*\kappa}$ and ϕ differ by a constant and $u_{f*\kappa}$ is known to vanish at some point, it follows that $u_{f*\kappa}$ satisfies the same estimate. To construct the potential ϕ , we follow the method of Dinh and Sibony [39, Prop. 2.1] which is itself based on [11] (we keep the notation from [39]). Let α be a closed (2, 2)-form on $X \times X$ which is cohomologous to the diagonal Δ . In [11], Bost, Gillet and Soulé construct an explicit (1, 1)form K on $X \times X$ such that $dd^c K = [\Delta] - \alpha$; they refer to it as the "Green current". It is C^{∞} outside the diagonal, and along Δ , it satisfies the estimates

(6.15)
$$K(x,y) = O\left(\frac{\log|x-y|}{|x-y|^2}\right) \text{ and } \nabla K(x,y) = O\left(\frac{\log|x-y|}{|x-y|^3}\right)$$

(here we mean that these estimates hold for the coefficients of K and ∇K in local coordinates). These estimates are easily deduced from the explicit expression of K as $\pi_*(\hat{\varphi}\eta - \beta)$ given in the proof of Proposition 2.1 of [39], where $\pi : \overline{X \times X} \to X \times X$ is the blow-up of the diagonal, η and β are smooth (1,1) forms on $\overline{X \times X}$ and $\hat{\varphi}$ is a function with logarithmic singularities along the proper transform of Δ in $X \times X$. It is shown in [39, Prop. 2.1] that a solution to Equation (6.14) is given by

(6.16)
$$\phi(x) = \int_{y \in X} K(x, y) \wedge (f^* \kappa(y) - \Theta(f^* \kappa)(y))$$

(in the notation of [39], $f^*\kappa$ and $\Theta(f^*\kappa)$ correspond to Ω^+ and Ω^- respectively). The coefficients of the smooth (1,1)-forms $f^*\kappa$ and $\Theta(f^*\kappa)$ have their uniform norms bounded by $C \|f\|_{C^1}^2$, where $C = C(\kappa) \leq C' \|\kappa\|_{\infty}$. The first estimate in (6.15) implies that the coefficients of K belong to L^p_{loc} for p < 2, so it follows from the Hölder inequality that $\|\phi\|_{C^0} \leq C'' \|\kappa\|_{\infty} \|f\|_{C^1}^2$ (for some constant C'' depending only on X). A similar estimate for $\nabla \phi$ is obtained from derivation under the integral sign and the fact that $\nabla K \in L^p_{\text{loc}}$ for p < 4/3. This concludes the proof.

6.3. Convergence and extremality.

Theorem 6.11. Let (X, ν) be a non-elementary random holomorphic dynamical system on a compact Kähler surface X, satisfying the moment condition (4.1). Then for μ_{∂} -almost every point $\underline{a} \in \text{Lim}(\Gamma)$, the following properties hold:

- (1) there is a unique nef and isotropic class $a \in H^{1,1}(X; \mathbf{R})$ of mass 1 with $\mathbb{P}(a) = \underline{a}$:
- (2) the convex set Cur(a) is a singleton $\{T_a\}$;
- (3) the class \underline{a} is an extremal point of $\mathbb{P}(\overline{\operatorname{Kah}}(X))$ and of $\mathbb{P}(\operatorname{Psef}(X))$;
- (4) the current T_a is extremal in the convex set of closed positive currents of mass 1.

With Lemma 5.5 and Equation (5.9), this theorem gives the first and second assertions of the following corollary; the third one follows from the first and the equivariance relation (5.16).

Corollary 6.12. The following properties are satisfied for $\nu^{\mathbf{N}}$ -almost every ω :

(1) there exists a unique closed positive current T^s_{ω} in the cohomology class $e(\omega)$; (2) for every Kähler form κ ,

$$\frac{1}{\mathbf{M}\left((f_{\omega}^{n})^{*}\kappa\right)}(f_{\omega}^{n})^{*}\kappa \xrightarrow[n \to \infty]{} T_{\omega}^{s}.$$

(3) the currents T^s_{ω} satisfy the equivariance property

$$(f_{\omega})^* T^s_{\sigma(\omega)} = \frac{\mathbf{M}((f_{\omega})^* T^s_{\sigma(\omega)})}{\mathbf{M}(T^s_{\omega})} T^s_{\omega} = \mathbf{M}((f_{\omega})^* T^s_{\sigma(\omega)}) T^s_{\omega}.$$

Proof of Theorem 6.11. The first and third properties were already established, respectively in Lemma 2.13 and 2.14 and Proposition 5.9. Property (4) follows from (2) and (3). It remains to prove (2). For this, we denote by \underline{f}^* the projective action of f^* on $\mathbb{P}H^{1,1}(X; \mathbf{R})$. For $\underline{a} \in \text{Lim}(\Gamma)$, let us set diam (\underline{a}) = Diam(a), where a is the unique pseudo-effective class of mass 1 such that $\mathbb{P}(a) = \underline{a}$; this defines a measurable function on $\text{Lim}(\Gamma)$, by Lemma 6.4. Our purpose is to show that diam (\underline{a}) = 0 for μ_{∂} -almost every \underline{a} . The stationarity of μ_{∂} reads

(6.17)
$$\int \operatorname{diam}\left(\underline{a}\right) \, d\mu_{\partial}\left(\underline{a}\right) = \iint \operatorname{diam}\left(\underline{f}^{*}\left(\underline{a}\right)\right) \, d\nu(f) d\mu_{\partial}\left(\underline{a}\right)$$

and iterating this relation gives

(6.18)
$$\int \operatorname{diam}\left(\underline{a}\right) \, d\mu_{\partial}\left(\underline{a}\right) = \int \operatorname{diam}\left(\underline{f}_{n}^{*}\cdots\underline{f}_{1}^{*}\left(\underline{a}\right)\right) \, d\nu(f_{1})\cdots d\nu(f_{n})d\mu_{\partial}\left(\underline{a}\right)$$

(notice the order of compositions chosen here). Since the diameter is upper-semicontinuous it is uniformly bounded on $\operatorname{Lim}(\Gamma)$. So, if we prove that

(6.19)
$$\lim_{n \to +\infty} \operatorname{diam}\left(\underline{f}_n^* \cdots \underline{f}_1^* \left(\underline{a}\right)\right) = 0$$

for $\nu^{\mathbf{N}}$ -almost every (f_n) and every \underline{a} , then we can apply the dominated convergence theorem to infer that diam $(\underline{a}) = 0 \ \mu_{\partial}$ -almost surely. To derive the convergence (6.19), note that

(6.20)
$$\operatorname{diam}\left(\underline{f}_{n}^{*}\cdots\underline{f}_{1}^{*}\left(\underline{a}\right)\right) = \frac{\operatorname{Diam}\left(f_{n}^{*}\cdots\overline{f}_{1}^{*}a\right)}{\mathbf{M}\left(f_{n}^{*}\cdots\overline{f}_{1}^{*}a\right)}$$

because Diam is homogeneous. Applying Lemma 6.8 and the multiplicativity of the Jacobian we get that

(6.21)
$$\operatorname{diam}\left(\underline{f}_{n}^{*}\cdots\underline{f}_{1}^{*}\left(\underline{a}\right)\right) \leqslant \frac{C\log\left(C\left\|\operatorname{Jac}(f_{1}\circ\cdots\circ f_{n})^{-1}\right\|_{\infty}\right)}{\mathbf{M}\left(f_{n}^{*}\cdots f_{1}^{*}a\right)} \leqslant C\frac{\sum_{i=0}^{n-1}\log\left\|f_{i}^{-1}\right\|_{C^{1}}}{\mathbf{M}\left(f_{n}^{*}\cdots f_{1}^{*}a\right)}$$

We conclude with two remarks. Firstly, the moment condition (4.1) implies that the sequence $\frac{1}{n}\sum_{i=0}^{n-1}\log \|f_i^{-1}\|_{C^1}$ is almost surely bounded. Secondly, Lemma 5.4 shows that $\mathbf{M}(f_n^*\cdots f_1^*a)$ goes exponentially fast to infinity for $\nu^{\mathbf{N}}$ -almost every $\omega = (f_n)$ (this is where the order of compositions matters). Thus diam $(\underline{f}_n^*\cdots \underline{f}_1^*(\underline{a})) \to 0$ almost surely, and we are done. \Box

Remark 6.13. The uniqueness of T_a in its cohomology class implies that T_a depends measurably on a. Indeed there is a set $E \subset \text{Lim}(\Gamma)$ of full measure along which the map $\underline{a} \mapsto T_a$ is continuous (recall that the space $\text{Cur}_1(X)$ of positive closed currents of mass 1 on X is a compact metrizable space). This implies that $\underline{a} \mapsto T_a$ is a measurable map from $\text{Lim}(\Gamma)$, endowed with the μ_{∂} -completion of the Borel σ -algebra, to $\text{Cur}_1(X)$, endowed with its Borel σ -algebra.

6.4. Continuous potentials. We now study the limit currents T^s_{ω} introduced in Corollary 6.12.

Theorem 6.14. Let (X, ν) be a non-elementary random holomorphic dynamical system on a compact Kähler surface X, satisfying the moment condition (4.1). Then for $\nu^{\mathbf{N}}$ -almost every ω the current T^s_{ω} has continuous potentials.

Lemma 6.15. Let κ be any Kähler form on X. For $\nu^{\mathbf{N}}$ -almost every ω , there exists an increasing sequence of integers $(n_i)_{i\geq 0} = (n_i(\omega))$ such that the potentials of the pull-back currents $\mathbf{M}((f_{\omega}^{n_i})^*\kappa)^{-1}u_{(f_{\omega}^{n_i})}^*\kappa$ are uniformly bounded.

If furthermore the exponential moment condition (5.23) holds, this assertion holds for all n (i.e. extracting a subsequence (n_i) is not necessary), and the function $\omega \mapsto \log^+ ||u_{T_{\omega}^s}||_{\infty}$ is $\nu^{\mathbf{N}}$ -integrable.

Proof of the Lemma. Recall the notation $\omega = (f_n)_{n \ge 0}$. First,

(6.22)
$$f_{n-1}^*\kappa = f_{n-1}^*\Theta(\kappa) + dd^c \left(u_\kappa \circ f_{n-1}\right)$$
$$= \Theta(f_{n-1}^*\kappa) + dd^c \left(u_{f_{n-1}^*\Theta(\kappa)} + u_\kappa \circ f_{n-1}\right)$$

(For the moment, we do not introduce the constants $E(f_n; \kappa)$ in the computation). We obtain

$$\begin{aligned} f_{n-2}^* f_{n-1}^* \kappa &= f_{n-2}^* \Theta(f_{n-1}^* \kappa) + dd^c \left(u_{f_{n-1}^* \Theta(\kappa)} \circ f_{n-2} + u_{\kappa} \circ (f_{n-1} \circ f_{n-2}) \right) \\ &= \Theta(f_{n-2}^* f_{n-1}^* \kappa) + dd^c \left(u_{f_{n-2}^* \Theta(f_{n-1}^* \kappa)} + u_{f_{n-1}^* \Theta(\kappa)} \circ f_{n-2} + u_{\kappa} \circ (f_{n-1} \circ f_{n-2}) \right). \end{aligned}$$

Setting $G_{j,k} = f_{k-1} \circ \cdots \circ f_j$, for $j \leq k-1$, (so in particular $G_{0,j} = f_{\omega}^j$ for all $j \geq 1$) and $G_{j,j} = \operatorname{id}_X$, we get

(6.23)
$$(f_{\omega}^n)^*\kappa = \Theta((f_{\omega}^n)^*\kappa) + dd^c \left(u_{\kappa} \circ f_{\omega}^n + \sum_{j=0}^{n-1} u_{f_j^*\Theta(G_{j+1,n}^*\kappa)} \circ G_{0,j} \right)$$

Let u_n denote the function in the parenthesis. We want to estimate the sup-norm $||u_n||_{\infty}$. Lemma 6.9 and Corollary 6.10 provide successively the following upper bounds

(6.24)
$$\left\| u_{f_{j}^{*}\Theta(G_{j+1,n}^{*}\kappa)} \right\|_{\infty} \leq C \|f_{j}\|_{C^{1}}^{2} \mathbf{M}(G_{j+1,n}^{*}\kappa) \leq C \mathbf{M}(\kappa) \|f_{j}\|_{C^{1}}^{2} \|G_{j+1,n}^{*}\|,$$

(6.25)
$$\left\|\frac{1}{\mathbf{M}((f_{\omega}^{n})^{*}\kappa)}u_{n}\right\|_{\infty} \leq \frac{\|u_{\kappa}\|_{\infty}}{\mathbf{M}((f_{\omega}^{n})^{*}\kappa)} + C\mathbf{M}(\kappa)\sum_{j=0}^{n-1}\|f_{j}\|_{C^{1}}^{2}\frac{\left\|G_{j+1,n}^{*}\right\|}{\mathbf{M}((f_{\omega}^{n})^{*}\kappa)}$$

To estimate this sum we apply Theorem 5.10 to the subadditive cocycle $N(n, \omega) = \log ||(f_{\omega}^n)^*||$, as we did for Corollary 5.11: there exists a sequence (δ_j) of positive numbers converging to 0, an increasing sequence $n_i = n_i(\omega)$ of integers, and a constant $C'(\omega)$ such that

(6.26)
$$\frac{\|G_{j+1,n_i}^*\|}{\mathbf{M}((f_{\omega}^{n_i})^*\kappa)} \approx \frac{\|f_{j+1}^* \cdots f_{n_i-1}^*\|}{\|f_0^* \cdots f_{n_i-1}^*\|} \leqslant C' \exp(-(\lambda_1 - \delta_j)j)$$

for all $i \ge 1$ and all $0 \le j \le n_i$. Fix any real number ε with $0 < \varepsilon < \lambda_1$. Then from Lemma 4.3, we know that, for almost every ω , there is a constant $C''(\omega)$ such that $||f_j||_{C^1}^2 \le C'' \exp(\varepsilon j)$. So from (6.25) we get

(6.27)
$$\left\|\frac{1}{\mathbf{M}((f_{\omega}^{n_i})^*\kappa)}u_{n_i}\right\|_{\infty} \leq \frac{\|u_{\kappa}\|_{\infty}}{\mathbf{M}((f_{\omega}^{n_i})^*\kappa)} + C'''(\omega)\mathbf{M}(\kappa)\sum_{j=0}^{n_i-1}\exp(-(\lambda_1 - \varepsilon - \delta(j))j)\right\|_{\infty}$$

This inequality shows that $\|\mathbf{M}((f_{\omega}^{n_i})^*\kappa)^{-1}u_{n_i}\|_{\infty}$ is uniformly bounded.

Now, note that $u_{(f_{\omega}^n)*\kappa} = u_n + E_n$ with $E_n = -\int u_n \text{vol.}$ Since $\|\mathbf{M}((f_{\omega}^{n_i})*\kappa)^{-1}u_{n_i}\|_{\infty}$ is uniformly bounded, so is $\mathbf{M}((f_{\omega}^{n_i})*\kappa)^{-1}E_{n_i}$, and the first assertion of the lemma is established.

Now if the exponential moment condition (5.23) holds, we follow the same argument and apply Proposition 5.13 – instead of Theorem 5.10 – to (6.25), with $D(f) = ||f||_{C^1}^2$.

Proof of Theorem 6.14. First, we prove that the normalized potential $u_{T_{\omega}^s}$ is bounded, for $\nu^{\mathbf{N}}$ almost every ω . To see this, recall that $\mathbf{M}((f_{\omega}^n)^*\kappa)^{-1}(f_{\omega}^n)^*\kappa$ converges to T_{ω}^s as $n \to \infty$. From Lemma 6.15, we know that the normalized potentials $\mathbf{M}((f_{\omega}^n)^*\kappa)^{-1}u_{(f_{\omega}^n)^*\kappa}$ of the currents $\mathbf{M}((f_{\omega}^n)^*\kappa)^{-1}(f_{\omega}^n)^*\kappa$ are uniformly bounded along some subsequence $n_i = n_i(\omega)$. These potentials are $A\kappa_0$ -psh functions on X so, by compactness, they converge to $u_{T_{\omega}^s}$ in $L^1(X; \text{vol})$. Thus, $u_{T_{\omega}^s}$ is essentially bounded. We conclude that $u_{T_{\omega}^s}$ is bounded because quasi-plurisubharmonic functions are upper semi-continuous and have a value (in $\mathbf{R} \cup \{-\infty\}$) at every point. Now, we show that $u_{T^s_{\omega}}$ is continuous. Here, the argument is similar to the one used to prove Theorem 6.11. If T is a positive closed current with bounded potential on X, we define

(6.28)
$$\operatorname{Jump}(T) = \max_{x \in X} \left(\limsup_{y \to x} u_T(y) - \liminf_{y \to x} u_T(y) \right)$$

Then $0 \leq \text{Jump}(T) \leq 2 ||u_T||_{\infty}$, and Jump(T) = 0 if and only if u_T is continuous. In addition $\text{Jump}(f^*T) = \text{Jump}(T)$ for every $f \in \text{Aut}(X)$ because $f^*T = \Theta(f^*a) + dd^c(u_{f^*\Theta(a)} + u_T \circ f)$ and $u_{f^*\Theta([T])}$ is continuous (see Equation (6.10)). From the equivariance relation

(6.29)
$$T_{\omega}^{s} = \frac{1}{\mathbf{M}\left(\left(f_{\omega}^{n}\right)^{*} T_{\sigma^{n}\omega}^{s}\right)} T_{\sigma^{n}\omega}^{s},$$

which follows from the third assertion of Corollary 6.12, we get

(6.30)
$$\operatorname{Jump}\left(T_{\omega}^{s}\right) = \frac{1}{\mathbf{M}\left(\left(f_{\omega}^{n}\right)^{*}T_{\sigma^{n}\omega}^{s}\right)}\operatorname{Jump}\left(T_{\sigma^{n}\omega}^{s}\right).$$

Remark 6.13 says that $\omega \mapsto T_{\omega}^{s}$ is measurable; hence, $\omega \mapsto u_{T_{\omega}^{s}}$ is measurable. If C is large enough, the first step of the proof gives a subset $\Omega_{C} \subset \Omega$ such that $\nu(\Omega_{C}) > 0$ and $||u_{T_{\omega}^{s}}||_{\infty} \leq C$ for all $\omega \in \Omega_{C}$. By ergodicity of the shift, $\sigma^{n}\omega \in \Omega_{C}$ for almost every ω and infinitely many n; for such an n, $||u_{T_{\sigma^{n}\omega}^{s}}||_{\infty} \leq C$ and $\operatorname{Jump}(T_{\sigma^{n}\omega}^{s}) \leq 2C$. By Lemma 5.12, $\mathbf{M}\left((f_{\omega}^{n})^{*}T_{\sigma^{n}\omega}^{s}\right)$ goes to infinity almost surely. So, $\operatorname{Jump}(T_{\omega}^{s}) = 0$, and the proof is complete.

Theorem 6.16. Let (X, ν) be a non-elementary random holomorphic dynamical system on a compact Kähler surface X, satisfying the exponential moment condition (5.23). Then there exists $\theta > 0$ such that for $\nu^{\mathbf{N}}$ -almost every ω the potential $u_{T_{\omega}^{s}}$ is Hölder continuous of exponent θ .

Proof. The initial computations are similar (but not identical) to those used to reach Lemma 6.15. Keeping the notation $G_{j,n} = f_{n-1} \circ \cdots \circ f_j$, a descending induction starting from

(6.31)
$$f_{n-1}^*T_{\sigma^n\omega}^s = \Theta(f_{n-1}^*T_{\sigma^n\omega}^s) + dd^c \left(u_{f_{n-1}^*\Theta(T_{\sigma\omega}^s)} + u_{T_{\sigma^n\omega}^s} \circ f_{n-1}\right)$$

yields

$$(6.32) \quad (f^n_{\omega})^* T^s_{\sigma^n \omega} = \Theta\left((f^n_{\omega})^* T^s_{\sigma^n \omega}\right) + dd^c \left(\sum_{j=0}^{n-1} u_{f^*_j \Theta(G^*_{j+1,n} T^s_{\sigma^n \omega})} \circ f^j_{\omega} + u_{T^s_{\sigma^n \omega}} \circ f^n_{\omega}\right).$$

Thus, there is a constant of normalization $E = E(\omega; n)$ such that

(6.33)
$$u_{T^s_{\omega}} = \frac{1}{\mathbf{M}((f^n_{\omega})^*(T^s_{\sigma^n\omega}))} \left(\sum_{j=0}^{n-1} u_{f^*_j \Theta(G^*_{j+1,n}T^s_{\sigma^n\omega})} \circ f^j_{\omega} + u_{T^s_{\sigma^n\omega}} \circ f^n_{\omega} \right) + E.$$

Note that the additional term E does not affect the modulus of continuity of $u_{T^s_{\omega}}$. Since $\operatorname{Lip}(f_j) \leq ||f_j||_{C^1}$ for all j, Lemma 6.9 and Corollary 6.10 imply $\operatorname{Lip}(u_{f^*_j\Theta(a)}) \leq C||f_j||_{C^1}^2 \mathbf{M}(a)$ for every class $a \in \operatorname{Psef}(X)$; hence

(6.34)
$$\operatorname{Lip}\left(u_{f_{j}^{*}\Theta(G_{j+1,n}^{*}T_{\sigma^{n}\omega}^{s})}\right) \leq C \|f_{j}\|_{C^{1}}^{2} \mathbf{M}(G_{j+1,n}^{*}T_{\sigma^{n}\omega}^{s}) \leq C \|f_{j}\|_{C^{1}}^{2} \|G_{j+1,n}^{*}\|$$

(6.35)
$$\leq C \|f_{j}\|_{C^{1}}^{2} \prod_{\ell=j+1}^{n-1} \|f_{\ell}^{*}\|_{H^{1,1}} \leq C \prod_{\ell=j}^{n-1} \|f_{\ell}\|_{C^{1}}^{2}.$$

Finally, since $1 \leq \text{Lip}(f_j)$ for every $0 \leq j \leq n-1$, we obtain

(6.36)
$$\operatorname{Lip}\left(u_{f_{j}^{*}\Theta(G_{j+1,n}^{*}T_{\sigma^{n}\omega}^{s})} \circ f_{\omega}^{j}\right) \leq \operatorname{Lip}\left(u_{f_{j}^{*}\Theta(G_{j+1,n}^{*}T_{\sigma^{n}\omega}^{s})}\right) \prod_{\ell=0}^{j-1} \operatorname{Lip}(f_{\ell}) \leq C \prod_{\ell=0}^{n-1} \|f_{\ell}\|_{C^{1}}^{2}.$$

Denoting the modulus of continuity by $modc(u, r) = \sup_{d(x,x') \leq r} |u(x) - u(x')|$, we infer from Equation (6.33) that

(6.37)
$$\operatorname{modc}(u_{T_{\omega}^{s}}, r) \leq \frac{1}{\mathbf{M}\left((f_{\omega}^{n})^{*}(T_{\sigma^{n}\omega}^{s})\right)} \left(Cn \prod_{\ell=0}^{n-1} \|f_{\ell}\|_{C^{1}}^{2} \cdot r + \|u_{T_{\sigma^{n}\omega}^{s}}\|_{\infty}\right).$$

To ease notation set $\lambda = \lambda_{H^{1,1}}$. Fix a small $\varepsilon > 0$. By Lemma 5.12, for almost every ω there exists $C = C_{\varepsilon}(\omega)$ such that $\mathbf{M}((f_{\omega}^{n})^{*}(T_{\sigma^{n}\omega}^{s}))^{-1} \leq Ce^{-n(\lambda-\varepsilon)}$ for every n. Fix M larger than but close to $\exp(\mathbb{E}(\log \|f\|_{C^{1}}))$. Applied to the $\nu^{\mathbf{N}}$ -integrable function $\omega = (f_{n}) \mapsto \log \|f_{0}\|_{C^{1}}$, the Birkhoff ergodic theorem gives

(6.38)
$$\prod_{\ell=0}^{n-1} \|f_{\ell}\|_{C^{1}}^{2} \leq CM^{n} \text{ as well as } n \prod_{\ell=0}^{n-1} \|f_{\ell}\|_{C^{1}}^{2} \leq CM^{n}$$

for some $C = C_M(\omega)$ (increase M to deduce the second inequality from the first). Thus,

(6.39)
$$\operatorname{modc}(u_{T_{\omega}^{s}}, r) \leq C_{1} e^{-n(\lambda-\varepsilon)} \left(M^{n} r + \left\| u_{T_{\sigma^{n}\omega}^{s}} \right\|_{\infty} \right)$$

for some $C_1 > 0$. By the last assertion of Lemma 6.15, $\omega \mapsto \log^+ \|u_{T^s_\omega}\|_{\infty}$ is integrable, so for almost every ω there exists $C_2 = C_{\varepsilon}(\omega)$ such that $\|u_{T^s_{\sigma^n\omega}}\|_{\infty} \leq C_2 e^{\varepsilon n}$ holds for all n, and we infer that

(6.40)
$$\operatorname{modc}(u_{T_{\omega}^{s}}, r) \leq C_{3}e^{-n(\lambda-\varepsilon)}(M^{n}r + e^{\varepsilon n}) = C_{3}e^{-n(\lambda-2\varepsilon)}\left((Me^{-\varepsilon})^{n}r + 1\right).$$

Choosing n so that $r \simeq (Me^{-\varepsilon})^{-n}$ we get $\operatorname{modc}(u_{T^s_{\omega}}, r) \leq C_4 r^{\theta}$ with $\theta = \frac{\lambda - 2\varepsilon}{\log M + \varepsilon}$ and the proof of the theorem is complete.

7. GLOSSARY OF RANDOM DYNAMICS, II

In this section we consider a random holomorphic dynamical system (X, ν) on a compact Kähler surface, satisfying the moment condition (4.1). We collect a number of facts from the ergodic theory of random dynamical systems, including the associated skew products, fibered entropy and Lyapunov exponents of stationary measures, stable and unstable manifolds, and various measurable partitions. Here the group Γ_{ν} may a priori be elementary; also, the compactness assumption on X can be dropped in most of these results if (4.1) is strengthened to a C^2 -moment condition. Since some subsequent arguments rely on the work [16] of Brown and Rodriguez-Hertz, we have tried to make notation consistent with that paper as much as possible.

7.1. Skew products and stationary measures associated to (X, ν) . Define:

- $\Omega = \operatorname{Aut}(X)^{\mathbb{N}}$, whose elements are denoted by $\omega = (f_n)_{n \ge 0}$. On Ω , the one-sided shift is denoted by $\sigma \colon \Omega \to \Omega$.
- $-\Sigma = \operatorname{Aut}(X)^{\mathbb{Z}}$, whose elements are denoted by $\xi = (f_n)_{n \in \mathbb{Z}}$. On Σ , the two-sided shift is denoted by $\vartheta \colon \Sigma \to \Sigma$.

 $-\mathcal{X} = \Sigma \times X$ and $\mathcal{X}_+ = \Omega \times X$, whose elements are denoted by $x = (\xi, x)$ and $x = (\omega, x)$ respectively. The natural projections are denoted by $\pi_{\Sigma} : \mathcal{X} \to \Sigma$ (resp. $\pi_{\Omega} : \mathcal{X}_+ \to \Omega$) and $\pi_X : \mathcal{X} \to X$ (resp. $\pi_X : \mathcal{X}_+ \to X$, using the same notation).

Recall that the product σ -algebra on Ω (resp. Σ) is generated by **cylinders** (¹), and that it coincides with the Borel σ -algebra $\mathcal{B}(\Omega)$ (resp. $\mathcal{B}(\Sigma)$) (see [9, Lem. 6.4.2]).

7.1.1. Skew products. For $\omega \in \Omega$ and $n \ge 1$, f_{ω}^n is the left composition $f_{\omega}^n = f_{n-1} \circ \cdots \circ f_0$; in particular, $f_{\omega}^1 = f_0$ (see § 4.3). For n = 0, we set $f_{\omega}^0 = \text{id}$. This is consistent with the notation used in the previous sections. The same notation f_{ξ}^n is used for $\xi \in \Sigma$ and $n \ge 0$. When n < 0, we set $f_{\xi}^n = (f_n)^{-1} \circ \cdots \circ (f_{-1})^{-1}$. With this definition the cocycle formula $f_{\xi}^{n+m} = f_{\vartheta^m \xi}^n \circ f_{\xi}^m$ holds for all $(m, n) \in \mathbb{Z}^2$ and $\xi \in \Sigma$. By definition, the skew products induced by the random dynamical system (X, ν) are the transformations $F_+: \mathcal{X}_+ \to \mathcal{X}_+$ and $F: \mathcal{X} \to \mathcal{X}$ defined by

(7.1)
$$F_+: (\omega, x) \longmapsto (\sigma \omega, f^1_{\omega}(x))$$
 and $F: (\xi, x) \longmapsto (\vartheta \xi, f^1_{\xi}(x))$.

If $\varpi : \mathcal{X} \to \mathcal{X}_+$ denotes the natural projection, then $\varpi \circ F = F_+ \circ \varpi$. Note that F is invertible, with $F^{-1}(x) = (\vartheta^{-1}\xi, f_{\vartheta^{-1}\xi}^{-1}(x))$, but F_+ is not; indeed (\mathcal{X}, F) is the natural extension of (\mathcal{X}_+, F_+) .

Lemma 7.1. The measure μ on X is stationary if and only if the product measure

$$m_+ := \nu^{\mathbf{N}} \times \mu$$

on \mathcal{X}_+ is invariant under F_+ .

A stationary measure is said to be **ergodic** if it is an extremal point in the convex set of stationary measures; hence, μ is ergodic if and only if m_+ is F_+ -ergodic. Actually μ is ergodic if and only if every ν -almost surely invariant measurable subset $A \subset X$ (that is a measurable subset such that $\mu(A\Delta f^{-1}(A)) = 0$ for ν -almost every f) has measure $\mu(A) = 0$ or 1. This statement is part of the so-called **random ergodic theorem** (see [8, Propositions 1.8 and 1.9]).

Proposition 7.2. There exists a unique *F*-invariant probability measure *m* on \mathcal{X} projecting on m_+ under the natural projection $\mathcal{X} \to \mathcal{X}_+$. Moreover,

(1) the measure m is equal to the weak- \star limit

$$m = \lim_{n \to \infty} (F^n)_* \left(\nu^{\mathbf{Z}} \times \mu \right).$$

- (2) the projections $(\pi_{\Sigma})_*m$ and $(\pi_X)_*m$ are respectively equal to $\nu^{\mathbf{Z}}$ and μ ;
- (3) the equality $m = \nu^{\mathbf{Z}} \times \mu$ holds if and only if μ is f-invariant for ν -almost every f;
- (4) (\mathcal{X}, F, m) is ergodic if and only if $(\mathcal{X}_+, F_+, m_+)$ is.

The existence and uniqueness of m, as well as the characterization of its ergodicity, follow from the fact that (\mathcal{X}, F) is the natural extension of (\mathcal{X}_+, F_+) (see [61, §1.2] for a detailed explanation). See [69, §I.1] for the proof of Assertions (1), (2) and (3).

¹Cylinders are products $C = \prod C_j$ of Borel sets, all of which are equal to Aut(X) except finitely many of them. For simplicity, we denote a cylinder by $C = \prod_{j=0}^{N} C_j$ if $C_k = Aut(X)$ for |k| > N.

7.1.2. *Past, future, and partitions.* Let \mathcal{F} denote the σ -algebra on \mathcal{X} obtained by taking the *m*-completion of $\mathcal{B}(\Sigma) \otimes \mathcal{B}(X)$. It will often be important to detect objects depending only on the "future" or on the "past". To formalize this, we define two σ -algebras on Σ :

- $\hat{\mathcal{F}}^+$ is the $\nu^{\mathbf{Z}}$ -completion of the σ -algebra generated by the cylinders $C = \prod_{j=0}^N C_j$.
- $-\hat{\mathcal{F}}^-$ is the $\nu^{\mathbf{Z}}$ -completion of the σ -algebra generated by the cylinders $C = \prod_{j=-N}^{-1} C_j$.

To formulate it differently, we define **local stable and unstable sets** for the shift ϑ :

(7.2)
$$\Sigma_{\text{loc}}^{s}(\xi) = \{\eta \in \Sigma; \forall i \ge 0, \eta_i = \xi_i\} \text{ and } \Sigma_{\text{loc}}^{u}(\xi) = \{\eta \in \Sigma; \forall i < 0, \eta_i = \xi_i\}.$$

Then a subset of Σ is $\hat{\mathcal{F}}^+$ -measurable (resp. $\hat{\mathcal{F}}^-$ measurable) if, up to a set of zero $\nu^{\mathbb{Z}}$ -measure, it is Borel and saturated by local stable sets $\Sigma_{\text{loc}}^s(\xi)$ (resp. unstable sets $\Sigma_{\text{loc}}^u(\xi)$). The σ -algebra \mathcal{F}^+ on \mathcal{X} will be the *m*-completion of $\hat{\mathcal{F}}^+ \otimes \mathcal{B}(X)$. An \mathcal{F}^+ -measurable object should be understood as "depending only on the future", thus it makes sense on \mathcal{X} and on \mathcal{X}_+ . Actually \mathcal{F}^+ coincides with the completion of the pull-back of $\mathcal{B}(\mathcal{X}_+)$ under $\varpi : \mathcal{X} \to \mathcal{X}_+$. The σ -algebra \mathcal{F}^- of "objects depending only on the past" is defined analogously. Consider the partition into the subsets $\mathcal{F}^-(x) := \Sigma_{\text{loc}}^u(\xi) \times \{x\}$ (each of them can be naturally identified to Ω). Then, modulo *m*-negligible sets, the elements of \mathcal{F}^- are saturated by this partition.

For $\xi \in \Sigma$ we set $X_{\xi} = \{\xi\} \times X = \pi_{\Sigma}^{-1}(\xi)$, which can be naturally identified with X via π_X . The disintegration of the probability measure m with respect to the partition into fibers of π_{Σ} gives rise to a family of conditional probabilities m_{ξ} such that $m = \int m_{\xi} d\nu^{\mathbf{Z}}(\xi)$, because $(\pi_{\Sigma})_* m = \nu^{\mathbf{Z}}$.

Lemma 7.3. The conditional measure m_{ξ} on X_{ξ} satisfies $\nu^{\mathbf{Z}}$ -almost surely

$$m_{\xi} = \lim_{n \to +\infty} (f_{-1} \circ \cdots \circ f_{-n})_{*} \mu = \lim_{n \to +\infty} (f_{\vartheta^{-n}\xi}^{n})_{*} \mu.$$

In particular, the family of measures $\xi \mapsto m_{\xi}$ is \mathcal{F}^- -measurable.

Indeed, the convergence is a consequence of the martingale convergence theorem (see [8, §2.5] for details) and the second assertion easily follows.

Since $\xi \mapsto m_{\xi}$ is \mathcal{F}^- -measurable, the conditional measures of m on the atoms $\mathcal{F}^-(x) = \Sigma_{\text{loc}}^u(\xi) \times \{x\}$ of the partition generating \mathcal{F}^- are induced by the lifts of the conditionals of $\nu^{\mathbf{Z}}$ on the $\Sigma_{\text{loc}}^u(\xi)$, via the natural projection $\pi_{\Sigma} : \mathcal{X} \to \Sigma$. In addition we can simultaneously identify $\Sigma_{\text{loc}}^u(\xi)$ to Ω and $\nu^{\mathbf{Z}}(\cdot | \Sigma_{\text{loc}}^u)$ to $\nu^{\mathbf{N}}$. In this way we get

(7.3)
$$m(\cdot \mid \mathcal{F}^{-}(x)) = \nu^{\mathbf{Z}}(\cdot \mid \Sigma_{\text{loc}}^{u}(\xi)) \times \delta_{x} \simeq \nu^{\mathbf{N}}$$

for *m*-almost every $x = (\xi, x) \in \mathcal{X}$. This corresponds to Equation (9) in [16]. By [16, Prop. 4.6], this implies that $\mathcal{F}^+ \cap \mathcal{F}^-$ is equivalent, modulo *m*-negligible sets, to $\{\emptyset, \Sigma\} \otimes \mathcal{B}(X)$.

7.2. Lyapunov exponents. Let μ be a stationary measure for (X, ν) ; assume that μ (or equivalently m or m_+) is ergodic. The upper and lower Lyapunov exponents $\lambda^+ \ge \lambda^-$ are respectively defined by the almost sure limits

(7.4)
$$\lambda^{+} = \lim_{n \to \infty} \frac{1}{n} \log \|D_{x} f_{\omega}^{n}\| \text{ and } \lambda^{-} = \lim_{n \to \infty} \frac{1}{n} \log \|(D_{x} f_{\omega}^{n})^{-1}\|^{-1};$$

the existence of these limits is guaranteed by Kingman's subadditive ergodic theorem, thanks to the moment condition (4.1), and the convergence also holds on average. Let us now apply the Oseledets theorem successively to the tangent cocycle defined by the fiber dynamics $(\mathcal{X}_+, F_+, m_+)$, and then to the cocycle associated to (\mathcal{X}, F, m) . 7.2.1. The non-invertible setting. Define the tangent bundles $T\mathcal{X}_+ := \Omega \times TX$ and $T\mathcal{X} := \Sigma \times TX$, and denote by DF and DF_+ the natural tangent maps, that is $D_{(\xi,x)}F : \{\xi\} \times T_xX \rightarrow \{\vartheta\xi\} \times T_{f_{\mathcal{E}}(x)}X$ is induced by $D_x f_{\mathcal{E}}^1$:

(7.5)
$$D_{(\xi,x)}F(v) = D_x f_{\xi}^1(v) \quad (\forall v \in T_x X_{\xi} = T_x X)$$

For the non-invertible dynamics on \mathcal{X}_+ , the Oseledets theorem gives: for m_+ -almost every (ω, x) , there exists a non-trivial complex subspace $V^-(\omega, x)$ of $\{\omega\} \times T_x X$ such that

(7.6)
$$\forall v \in V^{-}(\omega, x) \setminus \{0\}, \quad \lim_{n \to +\infty} \frac{1}{n} \log \|D_x f_{\omega}^n(v)\| = \lambda^{-}$$

(7.7)
$$\forall v \notin V^{-}(\omega, x), \quad \lim_{n \to +\infty} \frac{1}{n} \log \|D_x f_{\omega}^n(v)\| = \lambda^+.$$

The field of subspaces V^- is measurable and almost surely invariant. Two cases can occur: either $\lambda^- < \lambda^+$ and $V^-(\omega, x)$ is almost surely a complex line, or $\lambda^- = \lambda^+$ and $V^-(\omega, x) = \{\omega\} \times T_x X$.

7.2.2. The invertible setting. For the dynamical system $F: \mathcal{X} \to \mathcal{X}$, the statement is:

- if
$$\lambda^- = \lambda^+$$
 then for *m*-almost every $x = (\xi, x)$, for every non-zero $v \in T_x X_{\xi} \simeq T_x X$,

(7.8)
$$\lim_{n \to \pm \infty} \frac{1}{n} \log \left\| D_x f_{\xi}^n(v) \right\| = \lambda^-;$$

- if $\lambda^- < \lambda^+$ then for *m*-almost every *x* there exists a decomposition $T_x X_{\xi} = E^-(\xi, x) \oplus E^+(\xi, x)$ such that for $\star \in \{-, +\}$ and every $v \in E^{\star}(\xi, x) \setminus \{0\}$,

(7.9)
$$\lim_{n \to \pm \infty} \frac{1}{n} \log \left\| D_x f_{\xi}^n(v) \right\| = \lambda^{\star}.$$

Furthermore the line fields E^{\pm} are measurable and invariant, and $\log |\angle (E^-, E^+)|$ is integrable (here, the "angle" $\angle (E^-(x), E^+(x))$ is the distance between the two lines $E^-(x)$ and $E^+(x)$ in $\mathbb{P}(T_x \mathcal{X})$).

7.2.3. *Invariant volume forms*. When there is an invariant volume form, it is well-known that the sum of Lyapunov exponents vanishes. So from Remark 3.2 we obtain:

Corollary 7.4. Assume that X is an Abelian, or K3, or Enriques surface. Let ν be a probability measure on Aut(X) satisfying the integrability condition (4.1), and μ be an ergodic ν -stationary measure. Then $\lambda^- + \lambda^+ = 0$.

7.2.4. Hyperbolicity. It can happen that λ^- and λ^+ have the same sign. If λ^- and λ^+ are both negative, the conditional measures m_{ξ} are atomic: this can be shown by adapting a classical Pesin-theoretic argument (see e.g. [59, Cor. S.5.2]) to the fibered dynamics of F on \mathcal{X} (see [63, Prop. 2] for a direct proof and an example where the m_{ξ} have several atoms). Such random dynamical systems are called **proximal**. For instance, generic random products of automorphisms of $\mathbb{P}^2(\mathbf{C})$, that is of matrices in PGL(3, \mathbf{C}), are proximal; in such examples the stationary measure is not invariant. Other examples are given by contracting iterated function systems.

When λ^+ and λ^- are both non-negative, we have the so-called **invariance principle**:

Theorem 7.5. Let (X, ν) be a random holomorphic dynamical system satisfying the integrability condition (4.1), and let μ be an ergodic stationary measure. If $\lambda^+(\mu) \ge \lambda^-(\mu) \ge 0$ then μ is almost surely invariant.
This result was proven by Crauel, building on ideas of Ledrappier (see Theorem 5.1, Corollary 5.3 and Remark 5.6 in [37], and also Avila-Viana [1, Thm B]).

Remark 7.6. If λ^- and λ^+ are both positive then μ is atomic. Indeed, since μ is almost surely invariant we get $m = \nu^{\mathbf{Z}} \times \mu$. Reversing time, the Lyapunov exponents of m become negative, so as explained above the measures m_{ξ} are atomic. By invariance $m_{\xi} = \mu$, so μ is atomic too.

By definition, μ is **hyperbolic** if $\lambda^- < 0 < \lambda^+$. In this case we rather use the conventional superscripts s/u instead of -/+ for stable and unstable objects. We also have $E^s = V^s$ in this case (and more generally when $\lambda^- < \lambda^+$); so, it follows that the complex line field E^s on $T\mathcal{X}$ is \mathcal{F}^+ -measurable. Conversely the unstable line field E^u is \mathcal{F}^- -measurable.

7.3. Intermezzo: local complex geometry. Recall that X is endowed with a Riemannian structure, hence a distance, induced by the Kähler metric κ_0 . For $x \in X$, we denote by euc_x the translation-invariant Hermitian metric on $T_x X$ (which is considered here as a manifold in its own right) associated to the Riemannian structure induced by $(\kappa_0)_x$. Given any orthonormal basis (e_1, e_2) of $T_x X$ for this metric, we obtain a linear isometric isomorphism from $T_x X$ to \mathbb{C}^2 , endowed respectively with euc_x and the standard euclidean metric; we shall implicitly use such identifications in what follows.

We denote by $\mathbb{D}(z; r)$ the disk of radius r around z in C, and set $\mathbb{D}(r) = \mathbb{D}(0; r)$.

7.3.1. Hausdorff and C^1 -convergence. Let $U \subset \mathbb{C}$ be a domain. If $\gamma: U \to X$ is a holomorphic curve, we can lift it canonically to a curve $\gamma^{(1)}: U \to TX$ by setting $\gamma^{(1)}(z) = (\gamma(z), \gamma'(z)) \in T_{\gamma(z)}X$, where $\gamma'(z)$ denotes the velocity of γ at z. The Kähler form κ_0 induces a Riemannian metric and therefore a distance dist $_{TX}$ on TX. We say that two parametrized curves γ_1 and γ_2 are δ -close in the C^1 -topology if dist $_{TX}(\gamma_1^{(1)}(z), \gamma_2^{(1)}(z)) \leq \delta$ uniformly on U. This implies that $\gamma_1(U)$ and $\gamma_2(U)$ are δ -close in the Hausdorff sense, but the converse does not hold (take $U = \mathbb{D}(1), \gamma_1(z) = (z, 0), \text{ and } \gamma_2(z) = (z^k, \varepsilon z^\ell)$ with k and ℓ large while ε is small).

7.3.2. *Good charts.* Let R_0 be the injectivity radius of κ_0 . We fix once and for all a family of maps $\Phi_x : U_x \subset T_x X \to X$ satisfying the following properties (for some uniform $C_0 > 0$):

- (i) U_x is an open neighborhood of 0 in $T_x X$ and Φ_x is a holomorphic diffeomorphism from U_x to an open subset V_x of X contained in the ball of radius R_0 around x;
- (ii) $\Phi_x(0) = x$ and $(D \Phi_x)_0 = id_{T_x X}$;
- (iii) on U_x , the Riemannian metrics euc_x and $\Phi_x^* \kappa_0$ satisfy $C_0^{-1} \leq \operatorname{euc}_x / \Phi_x^* \kappa_0 \leq C_0$;
- (iv) the family of maps Φ_x depends continuously on x.

With $r_0 \leq R_0/(\sqrt{2}C_0)$, we can add:

(v) for every orthonormal basis (e_1, e_2) of $T_x X$, the bidisk $\mathbb{D}(r_0)e_1 + \mathbb{D}(r_0)e_2$ is contained in U_x ; in particular, the ball of radius r_0 centered at the origin for euc_x is contained in U_x .

To make assertion (iv) more precise, fix a continuous family of orthonormal basis $(e_1(x), e_2(x))$ on some open set V of X: Assertion (iv) means that, if we compose Φ_x with the linear isomorphism $(z_1, z_2) \in \mathbb{C}^2 \mapsto z_1 e_1(x) + z_2 e_2(x) \in T_x X$ we obtain a continuous family of maps. If needed, we can also add the following property (see [54, pp. 107-109]):

(iii') euc_x osculates $\Phi_x^* \kappa_0$ up to order 2 at x.

The maps Φ_x have to be thought of as "holomorphic exponential maps"; they are used in the next paragraph to get a definite notion of local orthogonal projection in X.

7.3.3. Families of disks. A holomorphic disk $\Delta \subset X$ containing x is said to be a disk of size (at least) r at x (resp. of size exactly r at x), for some $r < r_0$, if there is an orthonormal basis (e_1, e_2) of $T_x X$ such that $\Phi_x^{-1}(\Delta)$ contains (resp. is) the graph $\{ze_1 + \varphi(z)e_2 ; z \in \mathbb{D}(r)\}$ for some holomorphic map $\varphi : \mathbb{D}(r) \to \mathbb{D}(r)$. By the Koebe distortion theorem its geometric characteristics around x at scale r are then comparable to that of a flat disk. An alternative definition for the concept of disks of size $\ge r$ could be that Δ contains the image of an injective holomorphic map $\gamma : \mathbb{D}(r) \to X$ such that $\gamma(\partial \mathbb{D}(r)) \subset X \setminus B_X(x; r)$ and $\|\gamma'\| \le D$, for some fixed constant D. Then, if Δ contains a disk of size r for one of these definitions, it contains a disk of size $\varepsilon_0 r$ for the other one, for some uniform $\varepsilon_0 > 0$; in particular, there is a constant Cdepending only on (X, κ_0) such that a disk of size r at x contains an embedded submanifold of $B_X(x; Cr)$.

Let (x_n) be a sequence converging to x in X, and let r be smaller than the radius r_0 introduced in Assertion (v), § 7.3.2. Let Δ_n be a family of disks of size at least r at x_n and Δ be a disk of size at least r at x. We say that Δ_n converges towards Δ as a sequence of disks of size r, if there is an orthonormal basis (e_1, e_2) of $T_x X$ for euc_x such that

- (i) $\Phi_x^{-1}(\Delta)$ contains the graph $\{ze_1 + \varphi(z)e_2; z \in \mathbb{D}(r)\}$ for some holomorphic function $\varphi \colon \mathbb{D}(r) \to \mathbb{D}(r);$
- (ii) for every s < r, if *n* is large enough, the disk $\Phi_x^{-1}(\Delta_n)$ contains the graph $\{ze_1 + \varphi_n(z)e_2; z \in \mathbb{D}(s)\}$ of a holomorphic function $\varphi_n \colon \mathbb{D}(s) \to \mathbb{D}(r);$
- (iii) for every $\varepsilon > 0$, we have $|\varphi(z) \varphi_n(z)| < \varepsilon$ on $\mathbb{D}(s)$ if n is large enough.

By the Cauchy estimates, the convergence then holds in the C^1 -topology (see § 7.3.1). It follows from the usual compactness criteria for holomorphic functions that the space of disks of size ron X is compact (for the topology induced by the Hausdorff topology in X). Likewise, if a sequence of disks of size r converges in the Hausdorff sense, then it also converges in the C^1 sense, at least as disks of size s < r, because two holomorphic functions φ and ψ from $\mathbb{D}(r)$ to $\mathbb{D}(r)$ whose graphs are ε -close are also $\varepsilon(r-s)^{-1}$ -close in the C^1 -topology.

It may also be the case that the Δ_n are contained in different fibers X_{ξ_n} of \mathcal{X} . By definition, we say that the sequence Δ_n converges to $\Delta \subset X_{\xi}$ if ξ_n converges to ξ and the projections of Δ_n converge to Δ in X.

7.3.4. Entire curves. An entire curve in X is a holomorphic map $\psi : \mathbb{C} \to X$. It is immersed if its velocity ψ' does not vanish. Our main examples of immersed curves will, in fact, be injective and immersed entire curves. If ψ_1 and ψ_2 are two immersed entire curves with the same image, there exists a holomorphic diffeomorphism of \mathbb{C} , i.e. a non-constant affine map $A: z \mapsto az + b$, such that $\psi_2 = \psi_1 \circ A$. If ψ is an immersed entire curve and $|\psi'| \ge \eta$ on $\mathbb{D}(z_0, s)$, its image contains a disk of size Cs at $\psi(z_0)$, for some C > 0 that depends only on η and κ_0 .

7.4. Stable and unstable manifolds. By Lemma 4.1, Condition (4.1) implies similar moment conditions for higher derivatives, so Pesin's theory applies. The following proposition summarizes the main properties of Pesin local stable and unstable manifolds. Recall that a function h is ε -slowly varying, relatively to some dynamical system g, if $e^{-\varepsilon} \leq h(g(x))/h(x) \leq e^{\varepsilon}$ for every x. We view the stable manifold of $x = (\xi, x)$ as contained in X_{ξ} ; it can also be viewed as a subset of X: whether we consider one or the other point of view should be clear from the context. If $x = (\xi, x)$ and $y = (\xi, y)$ are points of the same fiber X_{ξ} , we denote by $dist_X(x, y)$ the Riemannian distance between x and y computed in X.

Proposition 7.7. Let (X, ν) be a random holomorphic dynamical system, and μ be an ergodic and hyperbolic stationary measure. Then, for every $\delta > 0$, there exists measurable positive δ -slowly varying functions r and C on \mathcal{X} (depending on δ) and, for m-almost every $x = (\xi, x) \in \mathcal{X}$, local stable and unstable manifolds $W^s_{r(x)}(x)$ and $W^u_{r(x)}(x)$ in X_{ξ} such that m-almost surely:

- (1) $W_{r(x)}^{s}(x)$ and $W_{r(x)}^{u}(x)$ are holomorphic disks of size at least 2r(x) at x respectively tangent to $E^{s}(x)$ and $E^{u}(x)$;
- (2) for every $y \in W^s_{r(x)}(x)$ and every $n \ge 0$,

$$\operatorname{dist}_X(F^n(x), F^n(y)) \leq C(x) \exp((\lambda^- + \delta)n);$$

likewise for every $y \in W^u_{r(x)}(x)$ *and every* $n \ge 0$

dist_X(
$$F^{-n}(x), F^{-n}(y)$$
) $\leq C(x) \exp(-(\lambda^{+} - \delta)n);$
(3) $F(W^{s}_{r(x)}(x)) \subset W^{s}_{r(F(x))}(F(x))$ and $F^{-1}(W^{u}_{r(F(x))}(F(x))) \subset W^{u}_{r(x)}(x).$

By Lusin's theorem, for every $\varepsilon > 0$ we can select a compact subset $\mathcal{R}_{\varepsilon} \subset \mathcal{X}$ with $m(\mathcal{R}_{\varepsilon}) > 1 - \varepsilon$, on which r(x) and C(x) can be replaced by uniform constants (respectively denoted by r and C) and the following additional property holds:

(4) on $\mathcal{R}_{\varepsilon}$ the local stable and unstable manifolds $W_r^{s/u}(x)$ vary continuously for the C^1 -topology (in the sense of § 7.3.1 and 7.3.3).

The subsets $\mathcal{R}_{\varepsilon}$ are usually called **Pesin sets**, or regular sets. We also denote the local stable or unstable manifolds by $W_{\text{loc}}^{s/u}(x)$, or by $W_r^{s/u}(x)$ when x is in a Pesin set on which $r(\cdot) \ge r$. On several occasions we will have to deal with measurability issues for $W_{\text{loc}}^{s/u}(x)$ as a function of x: this will be done by exhausting \mathcal{X} by Pesin sets and using their continuity on $\mathcal{R}_{\varepsilon}$.

The global stable and unstable manifolds of x are respectively defined by the following increasing unions:

(7.10)
$$W^{s}(x) = \bigcup_{n \ge 0} F^{-n} \left(W^{s}_{r(x)}(F^{n}(x)) \right) \text{ and } W^{u}(x) = \bigcup_{n \ge 0} F^{n} \left(W^{u}_{r(x)}(F^{-n}(x)) \right).$$

In particular, they are injectively immersed holomorphic curves in X_{ξ} . Pesin theory shows that:

(7.11)
$$W^{s}(x) = \left\{ (\xi, y) \in X_{\xi} ; \limsup_{n \to \infty} \frac{1}{n} \log \operatorname{dist}_{X}(F^{n}(\xi, y), F^{n}(\xi, x)) < 0 \right\}$$

(7.12)
$$W^{u}(x) = \left\{ (\xi, y) \in X_{\xi} ; \limsup_{n \to -\infty} \frac{1}{|n|} \log \operatorname{dist}_{X}(F^{n}(\xi, y), F^{n}(\xi, x)) < 0 \right\}.$$

Proposition 7.8. Under the assumptions of Proposition 7.7, $W^{s}(x)$ and $W^{u}(x)$ are biholomorphic to **C** for m-almost every x.

More precisely, $W^s(x)$ is parametrized by an injectively immersed entire curve $\psi_x^s : \mathbf{C} \to X$ such that $\psi_x^s(0) = x$ and this parametrization is unique, up to an homothety $z \mapsto az$ of \mathbf{C} . Likewise, $W^u(x)$ is parametrized by such an entire curve ψ_x^u .

Proof. By (7.10) and Proposition 7.7.(3), $W^s(x)$ is an increasing union of disks and is therefore a Riemann surface homeomorphic to \mathbb{R}^2 ; so, it is biholomorphic to \mathbb{C} or \mathbb{D} . Let $A \subset \mathcal{X}$ be a set of positive measure on which $r \ge r_0$ and $C \le C_0$. By Proposition 7.7.(2), there exists $n_0 \in \mathbb{N}$ and $m_0 > 0$ such that if $n \ge n_0$ and if x and $F^n(x)$ belong to A, then $W_r^s(F^n(\xi, x)) \setminus (F^n W_r^s(\xi, x))$ is an annulus of modulus $\ge m_0$. Now for m-almost every $x \in \mathcal{X}$ there is an infinite sequence (k_j) such that $F^{k_j}(x) \in A$ and $k_{j+1} - k_j > n_0$. For such an x, $W^s(x) \setminus W_r^s(x)$ contains an infinite nested sequence of annuli of modulus at least m_0 , namely the $F^{-k_{j+1}}(W_r^s(F^{k_{j+1}}(x)) \setminus F^{k_{j+1}-k_j}(W_r^s(F^{k_j}(x)))$. Thus, $W^s(x)$ is biholomorphic to \mathbb{C} . \Box

If we are only interested in stable manifolds, there is a simplified version of Proposition 7.7 which takes place on X:

Proposition 7.9. Let (X, ν) be a random holomorphic dynamical system and μ an ergodic stationary measure, whose Lyapunov exponents satisfy $\lambda^- < 0 \leq \lambda^+$. Then for m_+ -almost every (ω, x) the stable set

$$W^{s}(\omega, x) = \left\{ y \in X ; \lim_{n \to \infty} \sup_{n \to \infty} \frac{1}{n} \log \operatorname{dist}_{X}(f_{\omega}^{n}(y), f_{\omega}^{n}(x)) < 0 \right\}$$

is an injectively immersed entire curve in X.

Indeed, stable manifolds can be obtained from a purely "one-sided" construction, that is, by considering only positive iterates (see [69, Chap. III]). This also shows that local stable manifolds in \mathcal{X} are \mathcal{F}^+ -measurable, and may be viewed as living in \mathcal{X}_+ .

7.5. Fibered entropy. Here we recall the definition of the metric fibered entropy of a stationary measure μ (see [61, §2.1] or [69, Chap. 0 and I] for more details). If η is a finite measurable partition of X, its entropy relative to μ is $H_{\mu}(\eta) = -\sum_{C \in \eta} \mu(C) \log \mu(C)$. Then, we set

(7.13)
$$h_{\mu}(X,\nu;\eta) = \lim_{n \to \infty} \frac{1}{n} \int H_{\mu} \left(\bigvee_{k=0}^{n-1} \left(f_{\xi}^{k} \right)^{-1}(\eta) \right) d\nu^{\mathbf{N}}(\xi),$$

(7.14) $h_{\mu}(X,\nu) = \sup \{h_{\mu}(X,\nu;\eta); \eta \text{ a finite measurable partition of } X\}.$

Actually $h_{\mu}(X, \nu; \eta)$ can be interpreted as a conditional (or fibered) entropy for the skewproducts F_+ on \mathcal{X}_+ and F on \mathcal{X} . Indeed, the so-called Abramov-Rokhlin formula holds [10]:

(7.15)
$$h_{\mu}(X,\nu) = h_{\nu^{\mathbf{N}}\times\mu}(F_{+}|\eta_{\Omega}) = h_{m_{+}}(F_{+}) - h_{\nu^{\mathbf{N}}}(\sigma)$$

(7.16)
$$= h_m(F|\eta_{\Sigma}) = h_m(F) - h_{\nu}\mathbf{z}(\vartheta),$$

where η_{Ω} (resp. η_{Σ}) denotes the partition into fibers of the first projection $\pi_{\Omega} \colon \mathcal{X}_+ \to \Omega$ (resp. $\pi_{\Sigma} \colon \mathcal{X} \to \Sigma$) and in the second and fourth equalities we assume $h_{\nu^{N}}(\sigma) = h_{\nu^{Z}}(\vartheta) < \infty$. The next result is the fibered version of the **Margulis-Ruelle inequality**.

Proposition 7.10. Let (X, ν) be a random holomorphic dynamical system satisfying the moment condition (4.1) and μ be an ergodic stationary measure. If $h_{\mu}(X, \nu) > 0$ then μ is hyperbolic and $\min(\lambda^+, -\lambda^-) \ge \frac{1}{2}h_{\mu}(X, \nu)$.

Proof. See [2] or [69, Chap. II] for the inequality $\lambda^+ \ge \frac{1}{2}h_{\mu}(X,\nu)$. For $-\lambda^- \ge \frac{1}{2}h_{\mu}(X,\nu)$, we use the fact that $h_m(F|\eta_{\Sigma}) = h_m(F^{-1}|\eta_{\Sigma})$ (see e.g. [69, I.4.2]) and apply the Margulis-Ruelle inequality to F^{-1} . Beware that there is a slightly delicate point here: (F^{-1}, m) is not associated to a random dynamical system in our sense; fortunately, the statement of the Margulis-Ruelle inequality in [2] (see also [69, Appendix A]) covers this situation.

7.6. Unstable conditionals and entropy. Assume μ is ergodic and hyperbolic. By definition, an unstable Pesin partition η^u on \mathcal{X} is a measurable partition of $(\mathcal{X}, \mathcal{F}, \mu)$ with the following properties:

- η is increasing: $F^{-1}\eta^u$ refines η^u ;
- for *m*-almost every x, $\eta^u(x)$ is an open subset of $W^u(x)$ and

(7.17)
$$\bigcup_{n \ge 0} F^n \left(\eta^u(F^{-n}(x)) \right) = W^u(x);$$

 $-\eta^u$ is a generator, i.e. $\bigvee_{n=0}^{\infty} F^{-n}(\eta^u)$ coincides *m*-almost surely with the partition into points.

Here, as usual, $\eta^u(x)$ denotes the atom of η^u containing x, and $F^{-1}\eta^u$ is the partition defined by $(F^{-1}\eta^u)(x) = F^{-1}(\eta^u(F(x)))$. The definition of a **stable Pesin partition** η^s is similar. A neat proof of the existence of such a partition is given by Ledrappier and Strelcyn in [66], which easily adapts to the random setting (see [69, §IV.2], and [24]).

Lemma 7.11. There exists a stable (resp. unstable) Pesin partition whose atoms are \mathcal{F}^+ -measurable (resp. \mathcal{F}^- -measurable), that is, saturated by local stable (resp. unstable) sets $\Sigma_{\text{loc}}^s \times \{x\}$ (resp. $\Sigma_{\text{loc}}^u \times \{x\}$).

The existence of unstable partitions enables us to give a meaning to the **unstable conditionals** of m. Indeed, first observe that if η^u and ζ^u are two unstable Pesin partitions, then m-almost surely $m(\cdot|\eta^u)$ and $m(\cdot|\zeta^u)$ coincide up to a multiplicative factor on $\eta^u(x) \cap \zeta^u(x)$. Furthermore, there exists a sequence of unstable partitions η^u_n such that for almost every x, if K is a compact subset of $W^u(x)$ for the intrinsic topology (i.e. the topology induced by the biholomorphism $W^u(x) \simeq \mathbf{C}$) then $K \subset \eta^u_n(x)$ for sufficiently large n: indeed by (7.17), the sequence of partitions $F^n \eta^u$ does the job. Hence almost surely the conditional measure of m on $W^u(x)$ is well-defined up to scale; we define m^u_x by normalizing so that $m^u_x(\eta^u(x)) = 1$.

The next proposition is known as the (relative) **Rokhlin entropy formula**, stated here in our specific context.

Proposition 7.12. Let (X, ν) be a random holomorphic dynamical system satisfying the moment condition (4.1), and μ be an ergodic and hyperbolic stationary measure. Let η^u be an unstable Pesin partition. Then

$$h_{\mu}(X,\nu) = H_m(F^{-1}\eta^u | \eta^u) := \int \log J_{\eta^u}(x) dm(x),$$

where $J_{\eta^u}(x)$ is the "Jacobian" of F relative to η^u , that is

$$J_{\eta^{u}}(x) = m \left(F^{-1} \left(\eta^{u}(F(x)) \right) | \eta^{u}(x) \right)^{-1}.$$

Sketch of proof. The argument is based on the following sequence of equalities, in which η_{Σ} is the partition into fibers of π_{Σ} , as before:

(7.18)

$$h_{\mu}(X,\nu) = h_{m}(F|\eta_{\Sigma}) = h_{m}(F^{-1}|\eta_{\Sigma})$$

$$= h_{m}(F^{-1}|\eta^{u} \vee \eta_{\Sigma})$$

$$:= H_{m}(\eta^{u}|F\eta^{u} \vee \eta_{\Sigma}) = H_{m}(\eta^{u}|F\eta^{u}) = H_{m}(F^{-1}\eta^{u}|\eta^{u})$$

The equalities in the first and last line follow from general properties of conditional entropy (see [69, Chap. 0], the conditional entropy is denoted by $h_m^{\eta\Sigma}$ there). The Equality (7.18) is

non-trivial. If η^u were of the form $\bigvee_{n=0}^{+\infty} \eta$, where η is a 2-sided generator with finite entropy, it would follow from the general theory. For a Pesin unstable partition the result is established for diffeomorphisms in [67, Cor 5.3] and for random dynamics in [69, Cor. VI.7.1].

Remark 7.13. A similar formula holds in the stable direction:

$$h_{\mu}(X,\nu) = \int \log J_{\eta^{s}}(x) dm(x)$$
 where $J_{\eta^{s}}(x) = m \left(F \left(\eta^{s}(F^{-1}(x)) \right) | \eta^{s}(x) \right)^{-1}$.

The proof is identical to that of Proposition 7.12, applied to F^{-1} , with the same caveat as in Proposition 7.10: (F^{-1}, m) is not associated to a random dynamical system in our sense. The non-trivial point is to check that Equality (7.18) holds. Fortunately, the main purpose of [3] is to explain how to adapt [69, Chap. VI], hence Equality (7.18), to a more general notion of "random dynamical system" which covers the case of (F^{-1}, m) (see the last lines of [3, §5] for a short discussion of the Rokhlin formula).

The following consequence of the Rokhlin formula will play an important role in Section 9.

Corollary 7.14. Under the assumptions of the previous proposition, the following assertions are equivalent:

- (a) $h_{\mu}(X,\nu) = 0;$
- (b) $m(\cdot|\eta^u(x)) = \delta_x$ for m-almost every x;
- (c) $m(\cdot|\eta^u(x))$ is atomic for m-almost every x.

The same result holds for the stable Pesin partition η^s .

Proof. In view of the definition of J_{η^u} , the entropy vanishes if and only if for *m*-almost every x, $m(\cdot|\eta^u(x))$ is carried by a single atom of the finer partition $F^{-1}\eta^u$. Now since $H_m(F^{-1}\eta^u|\eta^u) = \frac{1}{n}H_m(F^{-n}\eta^u|\eta^u)$, the same is true for $F^{-n}\eta^u$, and finally since $(F^{-n}\eta^u)$ is generating, we conclude that (a) \Leftrightarrow (b). That (c) implies (a) follows from the same ideas but it is slightly more delicate, see [78, §2.1-2.2] for a clear exposition in the case of the iteration a single diffeomorphism, which readily adapts to our setting.

The result for the stable partition η^s follows by changing F to F^{-1} (see Remark 7.13).

8. STABLE MANIFOLDS AND LIMIT CURRENTS

Let (X, ν) be a non-elementary random holomorphic dynamical system on a Kähler surface. Assume that μ is an ergodic stationary measure admitting exactly one negative Lyapunov exponent, as in Proposition 7.9. Our purpose in this section is to relate the stable manifolds $W^s(\omega, x)$ to the stable currents T^s_{ω} constructed in §6. According to Proposition 7.9, the stable manifolds are parametrized by injective entire curves; the link between these curves and the stable currents will be given by the well-known Ahlfors-Nevanlinna construction of positive closed currents associated to entire curves.

8.1. Ahlfors-Nevanlinna currents. We denote by $\{V\}$ the integration current on a (possibly non-closed, or singular) curve V. Let $\phi : \mathbb{C} \to X$ be an entire curve. By definition, if α is a test 2-form, $\langle \phi_* \{\mathbb{D}(0,t)\}, \alpha \rangle = \int_{\mathbb{D}(0,t)} \phi^* \alpha$, which accounts for possible multiplicities coming from the lack of injectivity of ϕ ; $\{\mathbb{D}(0,t)\} = \{\phi(\mathbb{D}(0,t))\}$ when ϕ is injective. Set

(8.1)
$$A(R) = \int_{\mathbb{D}(0,R)} \phi^* \kappa_0 \text{ and } T(R) = \int_0^R A(t) \frac{dt}{t}$$

for R > 0. When ϕ is an immersion, A(R) is the area of $\phi(\mathbb{D}(0, R))$; in all cases, A(R) is the mass of $\phi_*\{(\mathbb{D}(0, R))\}$.

Proposition 8.1 (see Brunella [17, §1]). If $\phi : \mathbb{C} \to X$ is a non-constant entire curve, there exist sequences of radii (R_n) increasing to infinity such that the sequence of currents

$$N(R_n) = \frac{1}{T(R_n)} \int_0^{R_n} \phi_* \{ \mathbb{D}(0, t) \} \frac{dt}{t}$$

converges to a closed positive current T. If furthermore $\phi(\mathbf{C})$ is Zariski dense, the class $[T] \in H^{1,1}(X, \mathbf{R})$ is nef. In particular $\langle [T] | [T] \rangle \ge 0$ and $\langle [T] | [C] \rangle \ge 0$ for every algebraic curve $C \subset X$.

Such limit currents T will be referred to as Ahlfors-Nevanlinna currents associated to the entire curve $\phi \colon \mathbf{C} \to X$. If $\phi(\mathbf{C})$ is not Zariski dense then the closure $\overline{\phi(\mathbf{C})}$ (for the euclidean topology) is a (possibly singular) curve of genus 0 or 1; if ϕ is injective, then $\overline{\phi(\mathbf{C})}$ is rational.

8.2. Equidistribution of stable manifolds. If μ is hyperbolic, or more generally if it admits exactly one negative Lyapunov exponent, then, for m_+ -almost every $x = (\omega, x) \in \mathcal{X}_+$, the stable manifold $W^s(x)$, which is viewed here as a subset of X as in Proposition 7.9, is parametrized by an injectively immersed entire curve. Then we can relate the Ahlfors-Nevanlinna currents to the limit currents T^s_{ω} ; here are the three main results that will be proved in this section.

Theorem 8.2. Let (X, ν) be a non-elementary random holomorphic dynamical system on a compact Kähler surface, satisfying (4.1). Let μ be an ergodic stationary measure such that $\lambda^{-}(\mu) < 0 \leq \lambda^{+}(\mu)$. Then exactly one of the following alternative holds.

- (a) For m_+ -almost every x, the stable manifold $W^s(x)$ is not Zariski dense. Then μ is supported on a Γ_{ν} -invariant curve $Y \subset X$ and for m_+ -almost every x, $W^s(x) \subset Y$. In addition every component of Y is a rational curve, and the intersection form is negative definite on the subspace of $H^{1,1}(X; \mathbf{R})$ generated by the classes of components of Y.
- (b) For m_+ -almost every x the stable manifold $W^s(x)$ is Zariski dense and the only normalized Ahlfors-Nevanlinna current associated to $W^s(x)$ is T^s_{ω} .

Corollary 8.3. Under the assumptions of Theorem 8.2, if in addition μ is hyperbolic and nonatomic, then the Alternative (b) is equivalent to

(b') μ is not supported on a Γ_{ν} -invariant curve.

The next corollary follows from Theorem 6.16 and an ergodicity argument (see [24]).

Corollary 8.4. Under the assumptions of Theorem 8.2, assume furthermore that ν satisfies the exponential moment condition (5.23). Then in Alternative (b) there exists $\theta > 0$ such that for m_+ -almost every $x \in \mathcal{X}_+$ the Hausdorff dimension of $\overline{W^s(x)}$ equals $2 + \theta$.

8.3. Proof of Theorem 8.2 and its corollary. We work under the assumptions of Theorem 8.2.

Lemma 8.5. If there exists a proper Zariski closed subset of X with positive μ -measure, then:

- either μ is the uniform counting measure on a finite orbit of Γ_{ν} ;
- or μ has no atom and it is supported on a Γ_{ν} -invariant algebraic curve, which is the Γ_{ν} -orbit of an irreducible algebraic curve.

Proof. Consider the real number $\delta_{\max}^0(\mu) = \max_{x \in X} \mu(\{x\})$. If $\delta_{\max}^0(\mu) > 0$, there is a nonempty finite set $F \subset X$ for which $\mu(\{x\}) = \delta_{\max}^0(\mu)$. By stationarity, F is Γ_{ν} -invariant, and by ergodicity μ is the uniform measure on F. Now, assume that μ has no atom. Let $\delta_{\max}^1(\mu)$ be the maximum of $\mu(D)$ among all irreducible curves $D \subset X$. If $\mu(Z) > 0$ for some proper Zariski closed subset $Z \subset X$, then $\delta_{\max}^1(\mu) > 0$. Since two distinct irreducible curves intersect in at most finitely many points and μ has no atom, there are only finitely many irreducible curves Esuch that $\mu(E) = \delta_{\max}^1(\mu)$. To conclude, we argue as in the zero dimensional case.

If $V \subset X$ is a smooth curve, possibly with boundary, if T is a closed positive (1, 1)-current on X with a continuous normalized potential u_T (as in § 6.1.1), then, by definition of $\Theta(T)$ (see (6.2)),

(8.2)
$$\langle T \wedge \{V\}, \varphi \rangle = \int_{V} \varphi \Theta(T) + \int_{V} \varphi \, dd^{c}(u_{T}|_{V}),$$

for every test function φ . Here is the key relation between stable manifolds and limit currents:

Lemma 8.6. For m_+ -almost every $x = (\omega, x)$, if Δ is a disk contained in $W^s(x)$, then $T^s_{\omega} \land {\Delta} = 0$.

Proof. Without loss of generality we assume that the boundary of the disk Δ in $W^s(x) \simeq \mathbf{C}$ is smooth. We consider points $x = (\omega, x) \in \mathcal{X}_+$ which are generic in the following sense: they are regular from the point of view of Pesin's theory, and T^s_{ω} satisfies the conclusions of §6. By Pesin's theory, for every $\varepsilon > 0$, there is a set $A_{\varepsilon} \subset \mathbf{N}$ of density larger than $1 - \varepsilon$, such that for n in A_{ε} , the local stable manifold $W^s_r(F^n_+(x))$ is a disk of size $r = r(\varepsilon)$ at $f^n_{\omega}(x)$ and $f^n_{\omega}(\Delta)$ is a disk contained in an exponentially small neighborhood of $f^n_{\omega}(x)$. We have

(8.3)
$$\mathbf{M}(T^{s}_{\sigma^{n}\omega} \wedge \{f^{n}_{\omega}(\Delta)\}) = \int_{W^{s}_{r}(F^{n}_{+}(\varkappa))} \mathbf{1}_{f^{n}_{\omega}(\Delta)} \Theta(T^{s}_{\sigma^{n}\omega}) + \int_{W^{s}_{r}(F^{n}_{+}(\varkappa))} \mathbf{1}_{f^{n}_{\omega}(\Delta)} dd^{c} u_{T^{s}_{\sigma^{n}\omega}}.$$

Since $\mathbf{M}(T^s_{\sigma^n\omega}) = 1$, Lemma 6.1 shows that $\Theta(T^s_{\sigma^n\omega})$ is bounded by $A\kappa_0$; so the first integral on the right hand side of (8.3) is bounded by a constant times the area of $f^n_{\omega}(\Delta)$, which is exponentially small. By ergodicity, there exists $A'_{\varepsilon} \subset A_{\varepsilon}$ of density at least $1 - 2\varepsilon$ such that if $n \in A'_{\varepsilon}$, $\|u_{T^s_{\sigma^n\omega}}\|_{\infty}$ is bounded by some contant $D_{\varepsilon} > 0$. For such an n, let χ be a test function in $W^s_r(F^n_+(x))$ such that $\chi = 1$ in $W^s_{r/2}(F^n_+(x))$, and vanishing near $\partial W^s_r(F^n_+(x))$. Note that since $W^s_r(F^n_+(x))$ is of size r, the C^2 -norm of χ depends only on r. We write

(8.4)
$$\int_{W_r^s(F_+^n(x))} \mathbf{1}_{f_\omega^n(\Delta)} dd^c u_{T_{\sigma^n\omega}^s} \leqslant \int_{W_r^s(F_+^n(x))} \chi dd^c u_{T_{\sigma^n\omega}^s}$$
$$= \int_{W_r^s(F_+^n(x))} u_{T_{\sigma^n\omega}^s} dd^c \chi$$
$$\leqslant C(r) \|\chi\|_{C^2} \|u_{T_{\sigma^n\omega}^s}\|_{\infty}$$

where C(r) bounds the area of $W_r^s(F_+^n(x))$; this last term is uniformly bounded because $n \in A'_{\varepsilon}$. Thus we conclude that $\mathbf{M}(T^s_{\sigma^n\omega} \wedge \{f^n_{\omega}(\Delta)\})$ is bounded along such a subsequence.

On the other hand, the relation $(f^n_{\omega})^*T^s_{\sigma^n\omega} = \mathbf{M}((f^n_{\omega})^*T^s_{\sigma^n\omega})T^s_{\omega}$ gives

(8.5)
$$T^{s}_{\sigma^{n}(\omega)} \wedge \{f^{n}_{\omega}(\Delta)\} = \mathbf{M}\left((f^{n}_{\omega})^{*}T^{s}_{\sigma^{n}(\omega)}\right)(f^{n}_{\omega})_{*}(T^{s}_{\omega} \wedge \{\Delta\}).$$

The mass $\mathbf{M}((f_{\omega}^n)_*(T_{\omega}^s \wedge \{\Delta\}))$ is constant, equal to the mass of the measure $T_{\omega}^s \wedge \{\Delta\}$; so

(8.6)
$$\mathbf{M}\left(T^{s}_{\sigma^{n}(\omega)} \wedge \{f^{n}_{\omega}(\Delta)\}\right) = \mathbf{M}((f^{n}_{\omega})^{*}T^{s}_{\sigma^{n}(\omega)})\mathbf{M}(T^{s}_{\omega} \wedge \{\Delta\}).$$

By Lemma 5.12, $\mathbf{M}((f_{\omega}^n)^*T_{\sigma^n(\omega)}^s)$ goes exponentially fast to infinity. Since the left hand side is bounded, this shows that $\mathbf{M}(T_{\omega}^s \wedge \{\Delta\}) = 0$, as desired.

With Lemma 2.12, the following statement takes care of the first alternative in Theorem 8.2.

Lemma 8.7. If there is a Borel subset $A \subset \mathcal{X}_+$ of positive measure such that for every $x \in A$, the stable manifold $W^s(x)$ is contained in an algebraic curve, then μ is supported on a Γ_{ν} -invariant algebraic curve. In addition, for m_+ -almost every x, $\overline{W^s(x)}$ is an irreducible rational curve of negative self-intersection.

Proof. For $x \in A$, let D(x) be the Zariski closure of $W^s(x)$. Discarding a set of measure zero if needed, $W^s(x)$ is biholomorphic to \mathbb{C} so D(x) is a (possibly singular) irreducible rational curve, and $D(x)\setminus W^s(x)$ is reduced to a point. By Lemma 8.6, $T^s_{\omega} \wedge \{\Delta\} = 0$ for every disk $\Delta \subset W^s(x)$. Since T^s_{ω} has continuous potentials, $T^s_{\omega} \wedge \{D(x)\}$ gives no mass to points (see e.g. [28, Lem. 10.13] for the singular case). Therefore $T^s_{\omega} \wedge \{D(x)\}$ carries no mass on $D(x) = W^s(x) \cup (D(x)\setminus W^s(x))$, hence $T^s_{\omega} \wedge \{D(x)\} = 0$, and taking cohomology classes we infer that $\langle e(\omega) | [D(x)] \rangle = 0$. Then, by the Hodge index theorem, either $[D(x)]^2 < 0$ or [D(x)] is proportional to $e(\omega)$, however this latter case would contradict the fact that $e(\omega)$ is $\nu^{\mathbf{N}}$ -almost surely irrational (see Theorem 5.6; one could also use that $\operatorname{Cur}(e(\omega))$ is reduced to T^s_{ω}). Thus, $[D(x)]^2 < 0$, as asserted.

An irreducible curve with negative self-intersection is uniquely determined by its cohomology class; since $NS(X; \mathbb{Z})$ is countable, there are only countably many irreducible curves $(D_k)_{k \in \mathbb{N}}$ with negative self intersection. Since $W_{loc}^s(x) \subset D_k$ if and only if $D(x) = D_k$, and since local stable manifolds vary continuously on the Pesin regular set $\mathcal{R}_{\varepsilon}$ for every $\varepsilon > 0$, we infer that $\{x \in A ; D(x) = D_k\}$ is measurable for every k. Hence there exists an index k such that $m_+(\{x \in A ; [D(x)] = [D_k]\}) > 0$. Since x belongs to $W_{loc}^s(x)$, Fubini's theorem implies that $\mu(D_k) > 0$, and Lemma 8.5 shows that μ is supported on the Γ_{ν} -orbit of D_k .

Finally, this argument shows that the property $W^s_{loc}(x) \subset \bigcup_{k \in \mathbb{N}} D_k$, or equivalently that $W^s_{loc}(x)$ is contained in a rational curve of negative self intersection, is invariant and measurable, so by ergodicity of m_+ it is of full measure. The proof is complete.

We are now ready to conclude the proof of Theorem 8.2. Let A be the set of Pesin regular points such that $W^s(x)$ is contained in an algebraic curve. From the proof of Lemma 8.7, xbelongs to A if and only if $W_{loc}^s(x)$ is contained in one of the countably many irreducible curves $D_k \subset X$ of negative self-intersection. This condition determines a countable union of closed subsets in the Pesin sets $\mathcal{R}_{\varepsilon}$, hence A is Borel measurable. By Lemma 8.7, if A has positive m_+ -measure then Alternative (a) holds. So, if (a) is not satisfied, $W^s(x)$ is almost surely Zariski dense. Pick such a generic x, which further satisfies the conclusion of Lemma 8.6, and let N be an Ahlfors-Nevanlinna current associated to $W^s(x)$. By Proposition 8.1, [N] is a nef class so $[N]^2 \ge 0$. Thus, if we are able to show that $\langle [N] | [T_{\omega}^s] \rangle = 0$, we deduce from the Hodge index theorem and $\mathbf{M}(N) = 1$ that $[N] = [T_{\omega}^s] = e(\omega)$, hence $N = T_{\omega}^s$ by Theorem 6.11. So, it only remains to prove that $\langle [N] | [T_{\omega}^s] \rangle = 0$, or equivalently

$$(8.7) N \wedge T^s_{\omega} = 0.$$

This is intuitively clear because N is an Ahlfors-Nevanlinna current associated to the entire curve $W^s(x)$ and $T^s_{\omega} \wedge \{\Delta\} = 0$ for every bounded disk $\Delta \subset W^s(x)$. However, there is a technical difficulty to derive (8.7) from $T^s_{\omega} \wedge \{\Delta\} = 0$, even if $W^s(x)$ is an increasing union of such disks Δ .

To deal with this situation we use the geometric intersection theory of laminar currents (see [5, 41]). Unfortunately these papers only deal with currents of the form $\lim_{n} \frac{1}{A(R_n)}\phi(\mathbb{D}(0, R_n))$, instead of the Ahlfors-Nevanlinna currents introduced in Section 8.1, which were designed to get the nef property (Proposition 8.1). Let us explain how the formalism of [5, 41] can be adapted to the Ahlfors-Nevanlinna currents of Proposition 8.1. Following [44] we say that T is an **Ahlfors current** if there exists a sequence (Δ_n) of *unions* of smoothly bounded holomorphic disks such that length $(\partial \Delta_n) = o(\mathbf{M}(\Delta_n))$ and T is the limit as $n \to \infty$ of the sequence of normalized integration currents $\frac{1}{\mathbf{M}(\Delta_n)} \{\Delta_n\}$; here, $\operatorname{length}(\partial \Delta_n)$ is by definition the sum of the lengths of the boundaries of the disks constituting Δ_n , computed with respect to the Riemannian metric induced by κ_0 . We say furthermore that T is an **injective Ahlfors current** if the disks constituting Δ_n are disjoint or intersect along subsets with relative non-empty interior. By discretizing the integral defining the currents $N(R_n)$ in Proposition (8.1) we see that any Ahlfors-Nevanlinna current is an Ahlfors current.

Strongly approximable laminar currents are a class of positive currents introduced in [41] which are well suited for geometric intersection theory. In a nutshell, a current T is a strongly approximable laminar current if for every r > 0, there exists a uniformly laminar current T_r (non closed in general) made of disks of size r, and such that $\mathbf{M}(T - T_r) = O(r^2)$. Since these notions have been studied in a number of papers, we refer to [5, 41, 21] for definitions, the basic properties of these currents, and technical details. This presentation in terms of disks of size r is from [42, §4]. The next lemma is a mild generalization of the methods of [5, §7], [18, §4.3] and [41, §4]. For completeness we provide the details in Appendix A.

Lemma 8.8. Any injective Ahlfors current T on a projective surface X is a strongly approximable laminar current: if $T = \lim_{n \to \infty} \frac{1}{\mathbf{M}(\Delta_n)} \{\Delta_n\}$ as above, there exists a family of uniformly laminar currents T_r increasing to T whose constitutive disks are C^1 limits of pieces of the Δ_n , and such that if S is any closed positive current with continuous potential on X, $S \wedge T_r$ increases to $S \wedge T$ as r decreases to 0.

We can now conclude the proof of Theorem 8.2. Since by Theorem 3.4, X is projective, we can apply the previous lemma to any Ahlfors-Nevanlinna current N associated to $W^s(x)$. In this way we get a family of currents N_r such that $N_r \wedge T^s_{\omega}$ increases to $N \wedge T^s_{\omega}$ as r decreases to 0. On the other hand, by Lemma 8.6, the intersection of T^s_{ω} with every disk contained in $W^s(x)$ vanishes, so again using the fact that T^s_{ω} has a continuous potential, we infer that if Δ is any disk subordinate to N_r , $T^s_{\omega} \wedge \{\Delta\} = 0$. Hence $N_r \wedge T^s_{\omega} = 0$ for every r > 0, and finally $N \wedge T^s_{\omega} = 0$, as desired.

Proof of Corollary 8.3. Since (b') and (a) are contradictory, (b') implies (b). Conversely assume that μ is hyperbolic, non atomic and supported on a Γ_{ν} -invariant curve C. Since μ has no atom, it gives full mass to the regular set of C, hence $\Sigma \times T(\text{Reg}(C))$ defines a DF-invariant bundle, and by the Oseledets theorem the ergodic random dynamical system (C, ν, μ) must either have a positive or a negative Lyapunov exponent. If this exponent were positive then μ would be atomic, as observed in Section 7.2.4. Hence, the Lyapunov exponent tangent to C is negative and $W^s(x)$ is contained in C for m_+ -almost every x. So (b) implies (b').

9. NO INVARIANT LINE FIELDS

As above, let (X, ν) be a random holomorphic dynamical system satisfying the moment condition (4.1), and μ be an ergodic hyperbolic stationary measure. From §7.2 and §7.4, the local stable manifolds and stable Oseledets directions are \mathcal{F}^+ -measurable; so, $E^s(\xi, x)$ is naturally identified to $E^s(\omega, x)$ under the projection $(\xi, x) \in \mathcal{X} \mapsto (\omega, x) \in \mathcal{X}_+$, and the same property holds for stable manifolds. Then, m_+ -almost every $x \in \mathcal{X}_+$ has a Pesin stable manifold $W^s(x)$ (resp. direction $E^s(x)$). Let $V(x) = V(\omega, x)$ be such a measurable family of objects (stable manifolds, or stable directions, etc); we say that V(x) is **non-random** if for μ -almost every x, $V(\omega, x)$ does not depend on ω , that is, there exists V(x) such that $V(\omega, x) = V(x)$ for $\nu^{\mathbf{N}}$ almost every ω . If V is not non-random, we say that V **depends non-trivially on the itinerary**. Since stable directions depend only on the future, the random versus non-random dichotomy can be analyzed in \mathcal{X}_+ or in \mathcal{X} . Our purpose in this section is to establish the following result.

Theorem 9.1. Let (X, ν) be a non-elementary random holomorphic dynamical system on a compact Kähler surface, satisfying the Condition (4.1). Let μ be an ergodic and hyperbolic stationary measure, not supported on a Γ_{ν} -invariant curve. Then the following alternative holds:

- (a) either the Oseledets stable directions depend non-trivially on the itinerary;
- (b) or μ is Γ_{ν} -invariant and $h_{\mu}(X, \nu) = 0$.

In fact, the stiffness theorems of §10 imply that μ is often also invariant in case (a).

Remark 9.2. It turns out that unless μ is atomic, (a) and (b) are mutually exclusive. Indeed the main argument of [16] (²) implies that if the Oseledets stable directions depend non-trivially on the itinerary and μ is not atomic then its fiber entropy is positive (see also [16, Rmk 12.3]). This implies that for (X, ν, μ) as in Theorem 9.1, *if* μ *is not* Γ_{ν} -*invariant, then its fiber entropy is positive*.

9.1. Intersection multiplicities. If V_1 and V_2 are germs of curves at $0 \in \mathbb{C}^2$, with an isolated intersection at 0, the intersection multiplicity $\operatorname{inter}_0(V_1, V_2)$ is, by definition, the number of intersection points of V_1 and $V_2 + u$ in N for small generic $u \in \mathbb{C}^2$, where N is a neighborhood of 0 such that $V_1 \cap V_2 \cap N = \{0\}$ (see [34, §12]). It is a positive integer, and $\operatorname{inter}_0(V_1, V_2) = 1$ if and only if V_1 and V_2 are transverse at 0. We extend this definition by setting $\operatorname{inter}_0(V_1, V_2) = 0$ if V_1 or V_2 does not contain 0 and $\operatorname{inter}_0(V_1, V_2) = \infty$ if 0 is not an isolated point of $V_1 \cap V_2$, that is locally V_1 and V_2 share an irreducible component. The intersection multiplicity extends to analytic cycles (that is, formal integer combinations of analytic curves).

Lemma 9.3. The multiplicity of intersection $inter_0(\cdot, \cdot)$ is upper semi-continuous for the Hausdorff topology on analytic cycles.

In our situation we will only apply this result to holomorphic disks with multiplicity 1, in which case the topology is just the usual local Hausdorff topology.

Proof. Assume $\operatorname{inter}_0(V_1, V_2) = k$ and $V_{1,n} \to V_1$ (resp. $V_{2,n} \to V_2$) as cycles; we have to show that $\limsup \operatorname{inter}_0(V_{1,n}, V_{2,n}) \leq k$. If $k = \infty$ there is nothing to prove. Otherwise, $\{0\}$ is isolated in $V_1 \cap V_2$, so we can fix a neighborhood U of 0 such that $V_1 \cap V_2 \cap U = \{0\}$; then, the result follows from [34, Prop 2 p.141] (stability of proper intersections).

²This actually requires checking that the whole proof of [16] can be reproduced in our complex setting: we will come back to this issue in a forthcoming paper. Since this remark is not used in this paper, we take the liberty to anticipate on that research.

9.2. Generic intersection multiplicity of stable manifolds. Recall from §7.4 that for *m*-almost every $x = (\xi, x) \in \mathcal{X}$ there exists a local stable manifold $W^s_{r(x)}(x) \subset X_{\xi} \simeq X$, depending measurably on x; we might simply denote it by $W^s_{loc}(x)$.

Let us cover a subset of full measure in \mathcal{X} by Pesin subsets $\mathcal{R}_{\varepsilon_n}$. Take a point $x \in X$, and consider the set of points $((\xi, x), (\zeta, x)) \in \mathcal{R}_{\varepsilon_n} \times \mathcal{R}_{\varepsilon_m}$, for some fixed pair of indices (n, m); Lemma 9.3 shows that the intersection multiplicity inter_x $(W^s_{loc}(\xi, x), W^s_{loc}(\zeta, x))$ is an upper semi-continuous function of $((\xi, x), (\zeta, x))$ on that compact set. Thus, the intersection multiplicity inter_x $(W^s_{loc}(\xi, x), W^s_{loc}(\zeta, x))$ is a measurable function of (ξ, ζ) . Recall that

- the σ -algebra \mathcal{F}^- on \mathcal{X} is generated, modulo *m*-negligible sets, by the partition into subsets of the form $\Sigma_{\text{loc}}^u(\xi) \times \{x\}$ (see § 7.1, Equation (7.2));
- $\xi \mapsto m_{\xi}$ is \mathcal{F}^- -measurable, i.e $m_{\xi} = m_{\zeta}$ almost surely when $\zeta \in \Sigma^u_{\text{loc}}(\xi)$;
- the conditional measures of m with respect to this partition satisfy (see Equation (7.3))

(9.1)
$$m(\cdot \mid \mathcal{F}^{-}(x)) = \nu^{\mathbf{Z}}(\cdot \mid \Sigma_{\text{loc}}^{u}(\xi)) \times \delta_{x}.$$

The next lemma can be seen as a complex analytic version of [16, Lemma 9.9].

Lemma 9.4. Let $k \ge 1$ be an integer. Exactly one of the following assertions holds:

(a) for m-almost every $x = (\xi, x)$ and for $m(\cdot | \mathcal{F}^{-}(\xi, x))$ -almost every η

inter_x $(W_{loc}^s(\xi, x), W_{loc}^s(\eta, x)) \ge k + 1;$

(b) for m-almost every $x = (\xi, x)$ and for $m(\cdot | \mathcal{F}^{-}(\xi, x))$ -almost every η

inter_x $(W_{loc}^s(\xi, x), W_{loc}^s(\eta, x)) \leq k.$

Proof. The relation defined on \mathcal{X} by $(\xi, x) \simeq_k (\eta, y)$ if x = y and $W^s_{\text{loc}}(\xi, x)$ and $W^s_{\text{loc}}(\eta, y)$ have order of contact at least k + 1 at x is an equivalence relation which defines a partition \mathcal{Q}_k of \mathcal{X} . We shall see below that \mathcal{Q}_k is a measurable partition. Since $F: \mathcal{X} \to \mathcal{X}$ acts by diffeomorphisms on the fibers X of \mathcal{X} , we get that $F(\mathcal{Q}_k(x)) = \mathcal{Q}_k(F(x))$ for almost every $x \in \mathcal{X}$. Then, the proof of [16, Lemma 9.9] applies verbatim to show that if

(9.2)
$$m(\{x ; m(\mathcal{Q}_k(x)|\mathcal{F}^-(x)) > 0\}) > 0,$$

then

(9.3)
$$m(\{x \; ; \; m(\mathcal{Q}_k(x)|\mathcal{F}^-(x)) = 1\}) = 1.$$

This is exactly the desired statement. (This assertion says more than the mere ergodicity of m, which only implies that $m(\{x; m(\mathcal{Q}_k(x)|\mathcal{F}^-(x)) > 0\}) = 1.)$

It remains to explain why Q_k is a measurable partition. For this, we have to express the atoms of Q_k as the fibers of a measurable map to a Lebesgue space. As for the measurability of the intersection multiplicity, we consider an exhaustion of \mathcal{X} by countably many Pesin sets; then, it is sufficient to work in restriction to some compact set $\mathcal{K} \subset \mathcal{X}$ on which local stable manifolds have uniform size and vary continuously. Taking a finite cover of X by good charts (see § 7.3.2), and restricting \mathcal{K} again to keep only those local stable manifolds which are graphs over some fixed direction, we can also assume that $\pi_X(\mathcal{K})$ is contained in the image of a chart $\Phi_{x_0}: U_{x_0} \to$ $V_{x_0} \subset X$ and there is an orthonormal basis (e_1, e_2) such that for every $y \in \mathcal{K}$ the local stable manifold $\pi_X(W^s_{\text{loc}}(y))$ is a graph $\{ze_1 + \psi^s_y(z)e_2\}$ in this chart, for some holomorphic function ψ^s_y on $\mathbb{D}(r)$. Now the map from \mathcal{K} to $\mathbf{C}^2 \times \mathbf{C}^k$ defined by

is continuous. Since the fibers of this map are precisely the (intersection with \mathcal{K} of the) atoms of \mathcal{Q}_k , we are done.

The previous lemma is stated on \mathcal{X} because its proof relies on the ergodic properties of F. However, since stable manifolds depend only on the future, it admits the following more elementary formulation on X:

Corollary 9.5. Let $k \ge 1$ be an integer. Exactly one of the following assertions holds:

(a) for μ -almost every $x \in X$ and $(\nu^{\mathbf{N}})^2$ -almost every (ω, ω') ,

inter_x $(W_{loc}^s(\omega, x), W_{loc}^s(\omega', x)) \ge k + 1;$

(b) or for μ -almost every $x \in X$ and $(\nu^{\mathbf{N}})^2$ -almost every (ω, ω') ,

 $\operatorname{inter}_{x}\left(W^{s}_{\operatorname{loc}}(\omega, x), W^{s}_{\operatorname{loc}}(\omega', x)\right) \leq k.$

Combined with results from the previous sections, this alternative leads to the existence of a finite order of contact k_0 between generic stable manifolds $W_{\text{loc}}^s(\omega, x)$ and $W_{\text{loc}}^s(\omega', x)$. Note that the assumption that μ is not supported on an invariant curve is used exactly here.

Lemma 9.6. There exists a unique finite integer $1 \le k_0 < +\infty$ such that for μ -almost every $x \in X$ and $(\nu^N)^2$ -almost every pair (ω, ω') , inter_x $(W^s(\omega, x), W^s(\omega', x)) = k_0$.

Proof. Fix a small $\varepsilon > 0$ and consider a compact set $\mathcal{R}_{\varepsilon} \subset \mathcal{X}_{+}$ with $m_{+}(\mathcal{R}_{\varepsilon}) \ge 1 - \varepsilon$, along which local stable manifolds have size at least $r(\varepsilon)$ and vary continuously. Since by Theorem 8.2 for m_+ -a.e. x, the only Nevanlinna current associated to $W^s(x)$ is T^s_{ω} , we can further assume that this property holds for every $x \in \mathcal{R}_{\varepsilon}$. Let $A \subset X$ be a subset of full μ -measure on which the alternative of Corollary 9.5 holds for every $k \ge 1$. In \mathcal{X}_+ , consider the measurable partition into fibers of the form $\Omega \times \{x\}$; it corresponds to the partition \mathcal{F}^- in Lemma 9.4. Then, the associated conditional measures $m_+(\cdot | \Omega \times \{x\})$ are naturally identified with $\nu^{\mathbf{N}}$. Fix $x \in A$ such that $m_+(\mathcal{R}_{\varepsilon}|\Omega \times \{x\}) > 0$. Since (X, ν) is non-elementary, Theorems 5.6 and 6.11 provide pairs (ω_1, ω_2) in $(\pi_{\Omega}(\mathcal{R}_{\varepsilon}))^2$ for which the currents $T_{\omega_1}^s$ and $T_{\omega_2}^s$ are not cohomologous. By Theorem 8.2 these currents describe respectively the asymptotic distribution of $W^s(\omega_1, x)$ and $W^{s}(\omega_{2}, x)$ so we infer that $W^{s}(\omega_{1}, x) \neq W^{s}(\omega_{2}, x)$ and by the analytic continuation principle it follows that $W^s_{\text{loc}}(\omega_1, x) \neq W^s_{\text{loc}}(\omega_2, x)$. Let $k_1 < \infty$ be the intersection multiplicity of these manifolds at x. Since the intersection multiplicity is upper semi-continuous, we infer that for $\omega'_j \in \mathcal{R}_{\varepsilon}$ close to $\omega_j, j = 1, 2$, $\operatorname{inter}_x(W^s_{\operatorname{loc}}(\omega'_1, x), W^s_{\operatorname{loc}}(\omega'_2, x)) \leq k_1$. Thus for $k = k_1$ we are in case (b) of the alternative of Corollary 9.5. Applying then Corollary 9.5 successively for $k = 1, \ldots, k_1$, there is a first integer k_0 for which case (b) holds, and since (a) holds for $k_0 - 1$, we conclude that generically inter_x $(W_{loc}^s(\omega, x), W_{loc}^s(\omega', x)) = k_0.$

9.3. **Transversal perturbations.** The key ingredient in the proof of Theorem 9.1 is the following basic geometric lemma, which is a quantitative refinement of [5, Lemma 6.4].

Lemma 9.7. Let k be a positive integer. If r and c are positive real numbers, then there are two positive real numbers $\delta = \delta(k, r, c)$ and $\alpha = \alpha(k, r, c)$ with the following property. Let M_1 and M_2 be two complex analytic curves in $\mathbb{D}(r) \times \mathbb{D}(r) \subset \mathbb{C}^2$ such that

- (i) M_1 and M_2 are graphs $\{(z, f_j(z)) ; w \in \mathbb{D}_r\}$ of holomorphic functions $f_j : \mathbb{D}(r) \to \mathbb{D}(r)$;
- (ii) $M_1 \cap M_2 = \{(0,0)\}, and \operatorname{inter}_{(0,0)}(M_1, M_2) = k;$
- (iii) the k-th derivative satisfies $|(f_1 f_2)^{(k)}(0)| \ge c$.

If $M_3 \subset \mathbb{D}(r) \times \mathbb{D}(r)$ is a complex curve that does not intersect M_1 but is δ -close to M_1 in the C^1 -topology, then M_2 and M_3 have exactly k transverse intersection points in $\mathbb{D}(\alpha r) \times \mathbb{D}(\alpha r)$ (i.e. with multiplicity 1).

Proof. Without loss of generality we may assume that $\delta < 1$.

Step 1.– We claim that there exists $\alpha_1 = \alpha_1(k, r, c)$ such that for every $\alpha \leq \alpha_1$ and every $z \in \mathbb{D}(\alpha r)$ the following estimates hold:

(9.5)
$$\frac{1}{2} \frac{\left| (f_1 - f_2)^{(k)}(0) \right|}{k!} |z|^k \leq |f_1(z) - f_2(z)| \leq \frac{3}{2} \frac{\left| (f_1 - f_2)^{(k)}(0) \right|}{k!} |z|^k$$

(9.6)
$$\frac{1}{2} \frac{\left| (f_1 - f_2)^{(k)}(0) \right|}{(k-1)!} \left| z \right|^{k-1} \le \left| f_1'(z) - f_2'(z) \right| \le \frac{3}{2} \frac{\left| (f_1 - f_2)^{(k)}(0) \right|}{(k-1)!} \left| z \right|^{k-1}$$

Indeed put $g = f_1 - f_2 = \sum_{m \ge k} g_m z^m$. Assumptions (i) and (iii) give $|g(z)| \le 2r$ on $\mathbb{D}(r)$, and $g^{(k)}(0) \ne 0$. By the Cauchy estimates, $|g_n| \le 2r^{1-n}$ for all $n \ge 0$. Then on $\mathbb{D}(\alpha r)$ we get

$$\left| g(z) - \frac{g^{(k)}(0)}{k!} z^k \right| \le 2r \left(\frac{|z|}{r} \right)^{k+1} \left(1 - \frac{|z|}{r} \right)^{-1} \le 2r^{1-k} \frac{\alpha}{1-\alpha} |z|^k$$

There exists $\alpha_1(k, r, c)$ such that as soon as $\alpha \leq \alpha_1$, the right hand side of this inequality is smaller than $c |z|^k / (2k!)$; hence Estimate (9.5) follows. The same argument applies for (9.6) because

$$\left|g'(z) - \frac{g^{(k)}(0)}{(k-1)!} z^{k-1}\right| \leq 4(k+1) \left(\frac{|z|}{r}\right)^k \left(1 - \frac{|z|}{r}\right)^{-2} \leq 4(k+1)r^{1-k} \frac{\alpha}{(1-\alpha)^2} |z|^{k-1}.$$

Step 2.– For every $\alpha \leq \alpha_1$, if $\delta < c(\alpha r)^k/2k!$, M_2 and M_3 have exactly k intersection points, counted with multiplicities, in $\mathbb{D}(\alpha r) \times \mathbb{D}(\alpha r)$.

Indeed, the intersection points of M_3 and M_2 correspond to the solutions of the equation $f_3 = f_2$. To locate its roots, note that on the circle $\partial \mathbb{D}(\alpha r)$, the Inequality (9.5) implies

(9.7)
$$|f_1 - f_2| \ge \frac{1}{2} \frac{c}{k!} (\alpha r)^k.$$

Since $|f_1 - f_3| < \delta$, the choice $\delta < c(\alpha r)^k/2k!$ is tailored to assure that the hypothesis of the Rouché theorem is satisfied in $\mathbb{D}(\alpha r)$; so, counted with multiplicities, there are k solutions to the equation $f_3 = f_2$ on that disk. Furthermore by the Schwarz lemma $|f_2| < \alpha r$ on $\mathbb{D}(\alpha r)$ so the corresponding intersection points between M_2 and M_3 are contained in $\mathbb{D}(\alpha r) \times \mathbb{D}(\alpha r)$.

If k = 1 the proof is already complete at this stage, so from now on we assume $k \ge 2$.

Step 3.– Set $\delta_0 = |f_3(0)|$, and note that $\delta_0 \leq \delta$. Then for every $\alpha \leq 1/2$, in $\mathbb{D}(\alpha r)$ we have

(9.8)
$$\delta_0^{\frac{1+\alpha}{1-\alpha}} \le |f_1(z) - f_3(z)| \le \delta_0^{\frac{1-\alpha}{1+\alpha}}$$

(9.9)
$$\left|f_1'(z) - f_3'(z)\right| \leq \frac{1}{\alpha r} \delta_0^{\frac{1+2\alpha}{1+2\alpha}}$$

For this, recall the Harnack inequality: for any negative harmonic function in $\mathbb D$

(9.10)
$$\frac{1-|\zeta|}{1+|\zeta|} \le \frac{u(\zeta)}{u(0)} \le \frac{1+|\zeta|}{1-|\zeta|}.$$

Since $f_1 - f_3$ does not vanish and $|f_1 - f_3| \le \delta < 1$ in $\mathbb{D}(r)$, the function $\log |f_1 - f_3|$ is harmonic and negative there. Thus for $\alpha \le 1/2$, the Harnack inequality can be applied to $\zeta \mapsto (f_1 - f_3)(r\zeta)$ in \mathbb{D} : this gives (9.8). Likewise, we infer that

(9.11)
$$\delta_0^{\frac{1+2\alpha}{1-2\alpha}} \le |f_1(z) - f_3(z)| \le \delta_0^{\frac{1-2\alpha}{1+2\alpha}}$$

in $\mathbb{D}(2\alpha r)$, and (9.9) follows from the Cauchy estimate $\|g'\|_{\mathbb{D}(\alpha r)} \leq (\alpha r)^{-1} \|g\|_{\mathbb{D}(2\alpha r)}$.

Step 4. We now conclude the proof. Fix $\alpha = \alpha(k, r, c)$ such that $\alpha \leq \alpha_1$ and

(9.12)
$$\beta(\alpha) := \frac{1-2\alpha}{1+2\alpha} - \frac{k-1}{k} \times \frac{1+\alpha}{1-\alpha} > 0$$

(This will be our final choice for α .) Fix $\delta < c(\alpha r)^k/2k!$ and consider a solution z_0 of the equation $f_2(z) = f_3(z)$ in $\mathbb{D}(\alpha r)$ provided by Step 2. The transversality of M_2 and M_3 at $(z_0, f_2(z_0))$ is equivalent to $f'_3(z_0) \neq f'_2(z_0)$, so we only need

(9.13)
$$|(f_3 - f_1)'(z_0)| < |(f_2 - f_1)'(z_0)|.$$

Since $(f_1 - f_3)(z_0) = (f_1 - f_2)(z_0)$, combining the right hand side of Inequality (9.5) and the left hand side of Inequality 9.8, we get that

(9.14)
$$\frac{3}{2} \frac{|(f_1 - f_2)^{(k)}(0)|}{k!} |z_0|^k \ge \delta_0^{\frac{1+\alpha}{1-\alpha}}.$$

thus

(9.15)
$$|z_0| \ge \delta_0^{\frac{1}{k}\frac{1+\alpha}{1-\alpha}} \left(\frac{2k!}{3}\right)^{\frac{1}{k}} \left| (f_1 - f_2)^{(k)}(0) \right|^{-\frac{1}{k}}$$

Hence by (9.6) we get that

$$(9.16) \qquad \left| (f_2 - f_1)'(z_0) \right| \ge \frac{1}{2(k-1)!} \left(\frac{2k!}{3} \right)^{\frac{k-1}{k}} \delta_0^{\frac{k-1}{k} \frac{1+\alpha}{1-\alpha}} \left| (f_1 - f_2)^{(k)}(0) \right|^{\frac{1}{k}} \\ \ge \frac{1}{2(k-1)!} \left(\frac{2k!}{3} \right)^{\frac{k-1}{k}} \delta_0^{\frac{k-1}{k} \frac{1+\alpha}{1-\alpha}} c^{\frac{1}{k}}.$$

On the other hand by Estimate (9.9)

(9.17)
$$|(f_3 - f_1)'(z_0)| \leq \frac{1}{\alpha r} \delta_0^{\frac{1-2\alpha}{1+2\alpha}}$$

Since $\delta_0 \leq \delta$, we only need to impose one more constraint on δ (together with $\delta < c(\alpha r)^k/2k!$), namely

(9.18)
$$\delta^{\beta(\alpha)} < \frac{1}{2(k-1)!} \left(\frac{2k!}{3}\right)^{\frac{k-1}{k}} c^{\frac{1}{k}} r\alpha,$$

to get the desired inequality $|(f_3 - f_1)'(z_0)| < |(f_2 - f_1)'(z_0)|$.

Remark 9.8. Lemma 9.7 does not hold in the real analytic setting. Indeed, take an integer $n \equiv 1 \mod [4]$ and consider the *n*-th Chebychev polynomial T_n , defined by $T_n(\cos \theta) = \cos(n\theta)$; it satisfies $|T_n| \leq 1$ on [-1,1], $|T'_n| \leq 2n$ on [-1/2,1/2], and $T'_n(0) = n$. Then, set $P_n(x) = \frac{10}{n^2}T(x-\frac{5}{n}) + \frac{25}{n^2}$. This function satisfies $P'_n(5/n) = 10/n$, $P_n(5/n) = (5/n)^2$, and $15/n^2 \leq P_n \leq 35/n^2$ on [-1,1]. Now, if *n* is large, $M_1 = \{y = 0\}$, $M_2 = \{y = x^2\}$ and $M_3 = \{y = P_n(x)\}$ are three smooth algebraic curves in $(-1,1)^2 \subset \mathbb{R}^2$ such that M_3 is disjoint from M_1 but close to it in the C^1 topology, and M_3 is tangent to M_2 at $(5/n, 25/n^2)$. Similar arguments can be used to show that the semi-continuity of Lemma 9.3 fails for real analytic curves (though Corollary 9.5 may still be valid for real analytic random dynamical systems).

Let Δ_1 and Δ_2 be two disks of size r at $x \in X$, which are tangent at x; let $e_1 \in T_x X$ be a unit vector in $T_x \Delta_1 = T_x \Delta_2$ and e_2 a unit vector orthogonal to e_1 for κ_0 . Then, in the chart Φ_x , Δ_1 and Δ_2 are graphs $\{ze_1 + \psi_i(z)e_2\}$ of holomorphic functions $\psi_i \colon \mathbb{D}(r) \to \mathbb{D}(r), i = 1, 2$, such that $\psi_i(0) = 0$ and $\psi'_i(0) = 0$. If $\operatorname{inter}_x(\Delta_1, \Delta_2) = k$, then for $j = 1, \ldots, k - 1$ one has $\psi_1^{(j)}(0) = \psi_2^{(j)}(0)$ and $\psi_1^{(k)}(0) \neq \psi_2^{(k)}(0)$. We define the k-osculation of Δ_1 and Δ_2 at x to be

(9.19)
$$\operatorname{osc}_{k,x,r}(\Delta_1, \Delta_2) = \left| \psi_1^{(k)}(0) - \psi_2^{(k)}(0) \right|.$$

If $s \leq r$ and we consider Δ_1 and Δ_2 as disks of size s, then $\operatorname{osc}_{k,x,s}(\Delta_1, \Delta_2) = \operatorname{osc}_{k,x,r}(\Delta_1, \Delta_2)$. Thus, $\operatorname{osc}_{k,x,r}(\Delta_1, \Delta_2)$ does not depend on r, so we may denote this osculation number by $\operatorname{osc}_{k,x}(\Delta_1, \Delta_2)$. With this terminology, Lemma 9.7 directly implies the following corollary.

Corollary 9.9. Let k be a positive integer, and r and c be positive real numbers. Then, there are two positive real numbers δ and α , depending on (k, r, c), satisfying the following property. Let Δ_1 and Δ_2 be two holomorphic disks of size r through x, such that $\operatorname{inter}_x(\Delta_1, \Delta_2) = k$ and $\operatorname{osc}_{k,x}(\Delta_1, \Delta_2)) \ge c$. Let Δ_3 be a holomorphic disk of size r such that Δ_3 is δ -close to Δ_1 in the C^1 -topology but $\Delta_3 \cap \Delta_1 = \emptyset$. Then Δ_3 intersects Δ_2 transversely in exactly k points in $B_X(x, \alpha r)$.

The following lemma follows directly from the first step of the proof of Lemma 9.7.

Lemma 9.10. Let k be a positive integer, and r and c be positive real numbers. Then there exists a constant β depending only on (r, k, c) such that if Δ_1 and Δ_2 are two holomorphic disks of size r through x, such that $k = \operatorname{inter}_x(\Delta_1, \Delta_2)$ and $\operatorname{osc}_{k,x}(\Delta_1, \Delta_2)) \ge c$, then x is the only point of intersection between Δ_1 and Δ_2 in the ball $B_X(x, \beta r)$.

9.4. **Proof of Theorem 9.1.** Before starting the proof, we record the following two facts from elementary measure theory:

Lemma 9.11. Let $(\Omega, \mathcal{F}, \mathbb{P})$ be a probability space, and $\delta \in (0, 1)$.

(1) If φ is a measurable function with values in [0,1] and such that $\int \varphi d\mathbb{P} \ge 1 - \delta$, then

$$\mathbb{P}\left(\left\{x \; ; \; \varphi(x) \ge 1 - \sqrt{\delta}\right\}\right) \ge 1 - \sqrt{\delta}.$$

(2) If A_j is a sequence of measurable subsets such that $\mathbb{P}(A_j) \ge 1 - \delta$ for every j, then $\mathbb{P}(\limsup A_j) \ge 1 - \delta$.

Let us now prove Theorem 9.1. If the integer k_0 of Lemma 9.6 is equal to 1, then the Pesin stable manifolds corresponding to different itineraries at a μ -generic point $x \in X$ are generically transverse and we are in case (a) of the theorem. So, we now assume $k_0 > 1$ and we prove that μ is almost surely invariant (hence Γ_{ν} -invariant by Remark 4.2) and that its fiber entropy vanishes.

Step 1.– First, we construct a subset $\mathcal{G}_{\varepsilon}$ of "good points" in \mathcal{X} .

As described in Section 7.1.2, the atoms of \mathcal{F}^- are the sets $\mathcal{F}^-(x) = \Sigma^u_{\text{loc}}(\xi) \times \{x\}$ and the measures $m(\cdot | \mathcal{F}^-(x))$ can be naturally identified to $\nu^{\mathbb{N}}$ under the natural projections $\mathcal{F}^-(x) \xrightarrow{\sim} \Sigma^u_{\text{loc}}(\xi) \xrightarrow{\sim} \Omega$. For notational simplicity we denote these measures by $m_x^{\mathcal{F}^-}$.

For a small $\varepsilon > 0$, let $\mathcal{R}_{\varepsilon} \subset \mathcal{X}$ be a compact subset with $m(\mathcal{R}_{\varepsilon}) > 1 - \varepsilon$, along which local stable manifolds have size at least $2r(\varepsilon)$ and vary continuously. Since $\int m_{x}^{\mathcal{F}^{-}}(\mathcal{R}_{\varepsilon}) dm(x) \ge 1 - \varepsilon$, by Lemma 9.11 (1) we can select a compact subset $\mathcal{R}'_{\varepsilon} \subset \mathcal{R}_{\varepsilon}$ with $m(\mathcal{R}'_{\varepsilon}) \ge 1 - \sqrt{\varepsilon}$ such that for every $x \in \mathcal{R}'_{\varepsilon}$ one has $m_{x}^{\mathcal{F}^{-}}(\mathcal{R}_{\varepsilon}) \ge 1 - \sqrt{\varepsilon}$.

By assumption, $\operatorname{inter}_x(W^s_{\operatorname{loc}}(y_1), W^s_{\operatorname{loc}}(y_2)) = k_0$ for *m*-almost every $x = (\xi, x) \in \mathcal{R}'_{\varepsilon}$ and for $(m_x^{\mathcal{F}^-} \otimes m_x^{\mathcal{F}^-})$ -almost every pair of points $(y_1, y_2) \in (\mathcal{F}^-(x) \cap \mathcal{R}_{\varepsilon})^2$. Then there exists $\mathcal{R}''_{\varepsilon} \subset \mathcal{R}'_{\varepsilon}$ of measure at least $1 - 2\sqrt{\varepsilon}$ and a constant $c(\varepsilon) > 0$ such that

(9.20)
$$\operatorname{osc}_{k_0,x,r(\varepsilon)}(W^s_{\operatorname{loc}}(y_1),W^s_{\operatorname{loc}}(y_2)) \ge c(\varepsilon)$$

for every $x = (\xi, x) \in \mathcal{R}'_{\varepsilon}$ and all pairs (y_1, y_2) in a subset $A_{\varepsilon, x} \subset (\mathcal{F}^-(x) \cap \mathcal{R}_{\varepsilon})^2$ depending measurably on x and of measure

(9.21)
$$(m_{\chi}^{\mathcal{F}^{-}} \otimes m_{\chi}^{\mathcal{F}^{-}})(A_{\varepsilon,\chi}) \ge 1 - 4\sqrt{\varepsilon}$$

(we just used $(m_x^{\mathcal{F}^-} \otimes m_x^{\mathcal{F}^-})((\mathcal{F}^-(x) \cap \mathcal{R}_{\varepsilon})^2) \ge (1 - \sqrt{\varepsilon})^2 > 1 - 2\sqrt{\varepsilon})$. Finally, Fubini's theorem and Lemma 9.11 (1) provide a set $\mathcal{G}_{\varepsilon} \subset \mathcal{R}''_{\varepsilon}$ such that

- (a) $m(\mathcal{G}_{\varepsilon}) \ge 1 2\varepsilon^{1/4}$
- (b) for every $x \in \mathcal{G}_{\varepsilon}$, $W^s_{loc}(x)$ has size $2r(\varepsilon)$;
- (c) for every $x \in \mathcal{G}_{\varepsilon}$, there exists a measurable set $\mathcal{G}_{\varepsilon,x} \subset \mathcal{F}^{-}(x)$ with $m_{x}^{\mathcal{F}^{-}}(\mathcal{G}_{\varepsilon,x}) \ge 1 2\varepsilon^{1/4}$ such that for every y in $\mathcal{G}_{\varepsilon,x}$, $W^{s}_{\text{loc}}(y)$ has size $\ge r(\varepsilon)$ and, viewed as a subset of X,
 - it is tangent to $W^s_{loc}(x)$ to order k_0 at x,
 - $-\operatorname{osc}_{k_0,x,r(\varepsilon)}(W^s_{\operatorname{loc}}(x),W^s_{\operatorname{loc}}(y)) \ge c(\varepsilon).$

Note that $x \notin \mathcal{G}_{\varepsilon,x}$: indeed, when the local stable manifolds vary continuously, one can think of $A_{\varepsilon,x}$ as the complement of a small neighborhood of the diagonal in $\Omega \times \Omega$.

Step 2.- To make the argument more transparent, we first show that the fiber entropy vanishes.

Let η^s be a Pesin partition subordinate to local stable manifolds in \mathcal{X} . By Corollary 7.14 it is enough to show that for *m*-almost every x, $m(\cdot|\eta^s(x))$ is atomic (hence concentrated at x). Assume by contradiction that this is not the case. Therefore for $\varepsilon > 0$ small enough there exists $x = (\xi, x) \in \mathcal{G}_{\varepsilon}$ such that $m(\cdot|\eta^s(x))|_{\eta^s(x)\cap\mathcal{G}_{\varepsilon}}$ is non-atomic, and there exists an infinite sequence of pairwise distinct points $x_j = (\xi, x_j)$ in $\mathcal{G}_{\varepsilon} \cap \eta^s(x)$ converging to x. Then with $\mathcal{G}_{\varepsilon,\star}$ as in Property (c) of the definition of $\mathcal{G}_{\varepsilon}$, we have $m_{x_i}^{\mathcal{F}^-}(\mathcal{G}_{\varepsilon,x_i}) \ge 1 - 2\varepsilon^{1/4}$ for every j.

Identifying all $\mathcal{F}^{-}(x_{j})$ with $\Sigma_{\text{loc}}^{u}(\xi)$, by Lemma 9.11 (2) we can find $\zeta \in \Sigma_{\text{loc}}^{u}(\xi)$ such that (ζ, x_{j}) belongs to $\mathcal{G}_{\varepsilon,(\zeta,x_{j})}$ for infinitely many j's. Along this subsequence the local stable



FIGURE 1. On the left, a generic point x with the local stable manifolds $W^s_{\text{loc}}(\xi_i, x)$ for distinct $(\xi_i)_{i\geq 0}$ (see Step 1). On the right, the choice of the sequence (ζ, x_j) gives a family of local stable manifolds (see Step 2).

manifolds $W_{\text{loc}}^s(\zeta, x_j)$ form a sequence of disks of uniform size $r = 2r(\varepsilon)$ at x_j . Two such local stable manifolds are either pairwise disjoint or coincide along an open subset because they are associated to the same itinerary ζ .

Let us now use the notation from Corollary 9.9 and Lemma 9.10. We know that $W_{r(\varepsilon)}^{s}(\zeta, x_j)$ is tangent to $W_{r(\varepsilon)}^{s}(\xi, x)$ at x_j to order k_0 , with $\operatorname{osc}_{k_0, x_j, r(\varepsilon)}(W_{r(\varepsilon)}^{s}(x), W_{r(\varepsilon)}^{s}(\zeta, x_j)) \ge c(\varepsilon)$; so, by Lemma 9.10, $W_{r(\varepsilon)}^{s}(\zeta, x_j)$ and $W_{r(\varepsilon)}^{s}(\zeta, x_{j'})$ are disjoint as soon as $\operatorname{dist}_X(x_j, x_{j'}) < \beta r(\varepsilon)$. Finally, if j and j' are large enough, then $\operatorname{dist}_X(x_j, x_{j'}) < \alpha r(\varepsilon)$ and the C^1 distance between $W_{r(\varepsilon)}^{s}(\zeta, x_j)$ and $W_{r(\varepsilon)}^{s}(\zeta, x_{j'})$ is smaller than δ ; thus, Corollary 9.9 asserts that $W_{r(\varepsilon)}^{s}(\zeta, x_j)$ and $W_{r(\varepsilon)}^{s}(\zeta, x_{j'})$ cannot both be tangent to $W_{r(\varepsilon)}^{s}(\xi, x)$. This is a contradiction, and we conclude that the fiber entropy of m vanishes.

Step 3.– We now prove the almost sure invariance.

As in [16, Eq. (11.1)] we consider a measurable partition \mathcal{P} of \mathcal{X} with the property that for *m*-almost every (ξ, x) ,

(9.22)
$$\Sigma^s_{\text{loc}}(\xi) \times W^s_{r(\xi,x)}(\xi,x) \subset \mathcal{P}(\xi,x) \subset \Sigma^s_{\text{loc}}(\xi) \times W^s(\xi,x).$$

The existence of such a partition is guaranteed, for instance, by Lemma 7.11. By [16, Prop 11.1](³), to show that μ is almost surely invariant it is enough to prove that:

(9.23) for *m* almost every
$$\xi$$
, $m(\cdot | \mathcal{P}(\xi, x))$ is concentrated on $\sum_{loc}^{s}(\xi) \times \{x\}$.

By contradiction, assume that (9.23) fails. By contraction along the stable leaves, it follows that almost surely $\sum_{loc}^{s}(\xi) \times \{x\}$ is contained in

(9.24)
$$\operatorname{Supp}\left(m(\cdot|\mathcal{P}(\xi,x))|_{\mathcal{P}(\xi,x)\setminus\Sigma_{\operatorname{loc}}^{s}(\xi)\times\{x\}}\right)$$

(this is identical to the argument of Corollary 7.14). In particular for small ε we can find $x = (\xi, x) \in \mathcal{G}_{\varepsilon}$ and a sequence of points $x_j = (\xi_j, x_j) \in \mathcal{G}_{\varepsilon}$ such that x_j belongs to $\mathcal{P}(x) \cap \mathcal{G}_{\varepsilon}$, $x_j \neq x$ and (x_j) converges to x in X. We can also assume that the x_j are all distinct. By definition of $\mathcal{G}_{\varepsilon}$, $m_{x_j}^{\mathcal{F}^-}(\mathcal{G}_{\varepsilon,x_j}) \ge 1 - 2\varepsilon^{1/4}$ for every j. For $(\xi, \zeta) \in \Sigma^2$, set

(9.25)
$$[\xi, \zeta] = \Sigma_{\text{loc}}^u(\xi) \cap \Sigma_{\text{loc}}^s(\zeta);$$

³Brown and Rodriguez-Hertz make it clear that this result holds for an arbitrary smooth random dynamical system on a compact manifold.

that is, $[\xi, \zeta]$ is the itinerary with the same past as ξ and the same future as ζ . As above, identifying the atoms of the partition \mathcal{F}^- with Ω , Lemma 9.11 (2) provides an infinite subsequence (j_ℓ) and for every ℓ an itinerary $\zeta_{j_\ell} \in \Sigma^u_{\text{loc}}(\xi_{j_\ell})$ such that $y_{j_\ell} := (\zeta_{j_\ell}, x_{j_\ell})$ belongs to $\mathcal{G}_{\varepsilon, x_{j_\ell}}$ and all the ζ_{j_ℓ} have the same future, that is ζ_{j_ℓ} is of the form $[\xi_{j_\ell}, \zeta]$ for a fixed ζ . By definition,

(9.26)
$$\operatorname{inter}_{x_{j_{\ell}}}(W^s_{\operatorname{loc}}(x_{j_{\ell}}), W^s_{\operatorname{loc}}(y_{j_{\ell}})) = k_0$$

(9.27)
$$\operatorname{osc}_{k_0, x_{j_\ell}, r(\varepsilon)}(W^s_{\operatorname{loc}}(x_{j_\ell}), W^s_{\operatorname{loc}}(y_{j_\ell})) \ge c(\varepsilon).$$

In addition the disks $\pi_X(W^s_{\text{loc}}(y_{j_\ell}))$ are pairwise disjoint or locally coincide because the x_{j_ℓ} are distinct and the ζ_{j_ℓ} have the same future. Moreover, since x_{j_ℓ} belongs to $\mathcal{P}(x)$, $W^s(x_{j_\ell})$ coincides with $W^s(x)$. Therefore, the $\pi_X(W^s_{\text{loc}}(y_{j_\ell}))$ form a sequence of disjoint disks of size $2r(\varepsilon)$ at x_j , all tangent to $\pi_X(W^s_{\text{loc}}(x))$ to order k_0 , with osculation bounded from below by $c(\varepsilon)$. Since this sequence of disks is continuous and (x_j) converges towards x, Lemma 9.10 and Corollary 9.9 provide a contradiction, exactly as in Step 2. This completes the proof of the theorem.

10. STIFFNESS

Here we study Furstenberg's stiffness property for automorphisms of compact Kähler surfaces, thereby proving Theorem A. Our first results in §10.3 deal with elementary subgroups of Aut(X). The argument relies on the classification of such elementary groups together with general group-theoretic criteria for stiffness; these criteria are recalled in § 10.1 and 10.2. Theorem 10.7 concerns the much more interesting case of non-elementary subgroups; its proof combines all results of the previous sections with the work of Brown and Rodriguez-Hertz [16].

10.1. **Stiffness.** As said in Section 4.2, a random dynamical system (X, ν) is stiff if any ν -stationary measure is almost surely invariant; equivalently, every ergodic stationary measure is almost surely invariant. This property can be expressed in terms of ν -harmonic functions on Γ . Indeed if $\xi \colon X \to \mathbf{R}$ is a continuous function and μ is ν -stationary, then $\Gamma \ni g \mapsto \int_X \xi(gx) d\mu(x)$ is a bounded, continuous, right ν -harmonic function on Γ ; thus proving that μ is invariant amounts to proving that such harmonic functions are constant. Stiffness can also be defined for group actions: a group Γ **acts stiffly** on X if and only if (X, ν) is stiff for every probability measure ν on Γ whose support generates Γ ; in this definition, the measures ν can also be restricted to specific families, for instance symmetric finitely supported measures, or measures satisfying some moment condition. There are some general criteria ensuring stiffness directly from the properties of Γ . For instance, if $G \times X \to X$ is a continuous action of a topological group and $\Gamma \subset G$ is relatively compact, then Γ acts stiffly on X (this follows from the maximum principle for harmonic functions on $\overline{\Gamma}$, see also [51, Thm 3.5]). Another important case is that of Abelian and nilpotent groups:

Theorem 10.1. Let G be a locally compact, second countable, topological group. Let ν be a probability measure on G. If G is nilpotent of class ≤ 2 , then any measurable, ν -harmonic, and bounded function $\varphi: G \to \mathbf{R}$ is constant; thus, every measurable action of such a group is stiff.

This is a special case of results of Dynkin-Malyutov and Guivarc'h, see [57, 74] for the proof (⁴). When applying Theorem 10.1 to subgroups $A \subset Aut(X)$, we implicitly first replace A by its closure in Aut(X) then apply the theorem to this locally compact group.

10.2. Subgroups and hitting measures. A basic tool is the hitting measure on a subgroup, which we briefly introduce now (see [8, Chap. 5] for details). Let G be a locally compact second countable topological group. A notion of length can be defined in this context as follows: given a neighborhood U of the unit element, for any $g \in G$, $\text{length}_U(g)$ is the least integer $n \ge 1$ such that $g \in U^n$. By definition a probability measure ν on G has a finite first moment (resp. a finite exponential moment) if $\int \text{length}_U(g) d\nu(g) < \infty$ (resp. if $\int \exp(\alpha \text{length}_U(g)) d\nu(g) < \infty$ for some $\alpha > 0$). This condition does not depend on the choice of U.

Let ν be a probability measure on G, and consider the left random walk on G governed by ν . Given a subgroup $H \subset G$, for $\omega = (g_i) \in G^{\mathbf{N}}$, define the hitting time

(10.1)
$$T(\omega) = T_H(\omega) := \min\{n \ge 1 \; ; \; g_n \cdots g_1 \in H\}$$

If T is almost surely finite we say that H is **recurrent** and the distribution of $g_{T(\omega)} \cdots g_1$ is by definition the **hitting measure** of ν on H, which will be denoted by ν_H . The key property of ν_H is that if $\varphi : G \to \mathbf{R}$ is a ν -harmonic function, then $\varphi|_H$ is also ν_H -harmonic. Therefore, if μ is a ν -stationary measure, then it is also ν_H -stationary. Conversely, any bounded ν_H -harmonic function h on H admits a unique extension \tilde{h} to a bounded ν -harmonic function on G; this follows from Doob's optional stopping theorem.

If $[G:H] < \infty$ then H is recurrent and its stopping time admits an exponential moment. It follows that ν_H has a finite first (resp. exponential) moment if and only if ν does. Likewise, assume that H is a normal subgroup of G with G/H isomorphic to Z, and that ν is symmetric with a finite first moment. Then, the projection $\overline{\nu}$ of ν on G/H is symmetric with a finite first moment, so the random walk governed by $\overline{\nu}$ on $G/H \simeq Z$ is recurrent (see the Chung-Fuchs Theorem in [43, §5.4] or [35]) and H is recurrent.

Lemma 10.2. Let ν be a probability measure on Aut(X) and Γ' be a closed subgroup which is recurrent for the random walk induced by ν . Let ν' be the induced measure on Γ' . If (X, ν') is stiff then (X, ν) is stiff as well. This holds in particular if:

- (i) either $[\Gamma_{\nu}:\Gamma'] < \infty$
- (ii) or Γ' is a normal subgroup of Γ_{ν} with Γ_{ν}/Γ' isomorphic to **Z**, and ν is symmetric with a *finite first moment.*

Proof. Let μ be a ν -stationary measure on X. Then μ is ν' -stationary, hence by stiffness it is Γ' -invariant. Therefore for every Borel set $B \subset X$, the function $\Gamma \ni g \mapsto \mu(g^{-1}B)$ is a bounded ν -harmonic function which is constant on Γ' so by the uniqueness of harmonic extension it is constant, and ν is Γ -invariant.

10.3. Elementary groups. Recall that Aut(X) is a topological group for the topology of uniform convergence and is in fact a complex Lie group (with possibly infinitely many connected components). Let $Aut(X)^{\circ}$ be the connected component of the identity in Aut(X) and

(10.2)
$$\operatorname{Aut}(X)^{\#} = \operatorname{Aut}(X) / \operatorname{Aut}(X)^{\circ}.$$

⁴The proof in [74] is not correct (Lemma 2.5 there is false) but it works perfectly, and is quite short, if the support of ν is countable or if the nilpotency class is ≤ 2 . See the introduction of [74] for a summary of previous results.

Let ρ : Aut $(X) \to GL(H^*(X; \mathbb{Z}))$ be the natural homomorphism; its image is Aut $(X)^* = \rho(Aut(X))$ (see § 2.1.1); is kernel contains Aut $(X)^\circ$ and a theorem of Lieberman [68] shows that Aut $(X)^\circ$ has finite index in ker (ρ) . If Γ is a subgroup of Aut(X), we set $\Gamma^* = \rho(\Gamma)$.

Theorem 10.3. Let X be a compact Kähler surface. Let ν be a symmetric probability measure on Aut(X) satisfying the moment condition (4.1). If Γ_{ν} is elementary and Γ_{ν}^{*} is infinite, then (X, ν) is stiff.

Note that stiffness can fail when Γ_{ν}^* is finite: see Example 10.4 below. The proof relies on the classification of elementary subgroups of Aut(X) (see [21, Thm 3.2], [46]): if Γ_{ν} is elementary and Γ_{ν}^* is infinite there exists a finite index subgroup $A^* \subset \Gamma_{\nu}^*$ which is

- (a) either cyclic and generated by a loxodromic map;
- (b) or a free Abelian group of parabolic transformations possessing a common isotropic line; in that case, there is a genus 1 fibration $\tau \colon X \to S$, onto a compact Riemann surface S, such that Γ_{ν} permutes the fibers of τ .

Denote by $\rho_{\Gamma_{\nu}} \colon \Gamma_{\nu} \to \Gamma_{\nu}^*$ the restriction of ρ to Γ_{ν} . We distinguish two cases.

Proof when the kernel of $\rho_{\Gamma_{\nu}}$ is finite. Let A be the pre-image of A^* in Γ_{ν} ; it fits into an exact sequence $1 \to F \to A \to A^* \to 0$ with F finite, so a classical group theoretic lemma (see Corollary 4.8 in [31]) asserts that A contains a finite index, free Abelian subgroup A_0 , such that $\rho_{\Gamma_{\nu}}(A_0)$ has finite index in A^* . Since A_0 is Abelian, Theorem 10.1 shows that the action of (A_0, ν_{A_0}) on X is stiff. The index of A_0 in Γ being finite, Lemma 10.2 concludes the proof. \Box

Proof when the kernel of $\rho_{\Gamma_{\nu}}$ is infinite. In case (a), X is a torus \mathbb{C}^2/Λ and ker $(\rho_{\Gamma_{\nu}})$ is a group of translations of X (see Proposition 3.3). Let $A \subset \Gamma_{\nu}$ be the pre-image of A^* ; setting K =ker $(\rho_{\Gamma_{\nu}})$, we obtain an exact sequence $0 \to K \to A \to A^* \to 0$, with $A \subset \Gamma_{\nu}$ of finite index, $A^* \simeq \mathbb{Z}$ generated by a loxodromic element, and $K \subset X$ an infinite group of translations. Since ν is symmetric, the measure ν_A is also symmetric; since ν_A satisfies the moment condition (4.1), its projection on A^* has a first moment (note that if f is loxodromic, then $\log(||(f^*)^n||) \simeq |n|)$. Since K is Abelian, its action on X is stiff; thus, as in Lemma 10.2.(ii), the action of A on X is stiff. Since A has finite index in Γ , the action of Γ on X is stiff too by Lemma 10.2.(i).

In case (b), we apply Proposition 2.15. So, either X is a torus, or the action of Γ_{ν} on the base S of its invariant fibration $\tau: X \to S$ has finite order. In the latter case, a finite index subgroup Γ_0 of Γ preserves each fiber of τ ; then, Γ_0 contains a subgroup of index dividing 12 acting by translations on these fibers. This shows that Γ is virtually Abelian; in particular, Γ is stiff. The last case is when the image of Γ in Aut(S) is infinite and X is a torus \mathbf{C}^2/Λ_X . Then, $S = \mathbf{C}/\Lambda_S$ is an elliptic curve and τ is induced by a linear projection $\mathbf{C}^2 \to \mathbf{C}$, say the projection $(x, y) \mapsto x$. Lifting Γ to \mathbf{C}^2 , and replacing Γ by a finite index subgroup if necessary, its action is by affine transformations of the form $\tilde{f}: (x, y) \mapsto (x + a, y + mx + b)$ with m in \mathbf{C}^* , and (a, b) in \mathbf{C}^2 . This implies that Γ is a nilpotent group of length ≤ 2 ; by Theorem 10.1 it also acts stiffy and we are done.

Example 10.4. If $X = \mathbb{P}^2(\mathbf{C})$, its group of automorphisms is $\mathsf{PGL}_3(\mathbf{C})$ and for most choices of ν there is a unique stationary measure, which is not invariant; the dynamics is proximal, and this is opposite to stiffness (see [51]). If $X = \mathbb{P}^1(\mathbf{C}) \times C$, for some algebraic curve C, then $\mathsf{Aut}(X)$ contains $\mathsf{PGL}_2(\mathbf{C}) \times \mathsf{Aut}(C)$; if ν is a probability measure on $\mathsf{PGL}_2(\mathbf{C}) \times \{\mathrm{id}_C\}$, then in most cases the stationary measures are again non invariant.

Proposition 10.5. Let X be a compact Kähler surface and Γ be a subgroup of Aut(X) such that Γ^* is finite. If Γ preserves a probability measure whose support is Zariski dense in X, then the action of Γ on X is stiff.

Proof. Let μ be the invariant measure. Replacing Γ by a finite index subgroup we may assume that $\Gamma \subset \operatorname{Aut}(X)^\circ$. Let G be its closure (for the euclidean topology) in the Lie group $\operatorname{Aut}(X)^\circ$; it is a real Lie group preserving μ . We can assume that G is not compact, since otherwise stiffness is automatic. According to [33, Lem. 5.7], X is ruled, hence projective (since X is a compact Kähler surface). Pick an ample line bundle L on X, denote by $\mathbb{P}^N(\mathbb{C})$ the projective space $\mathbb{P}(H^0(X, L)^{\vee})$, with $N + 1 = h^0(X, L)$, and by $\Psi_L \colon X \to \mathbb{P}^N(\mathbb{C})$ the Kodaira-Iitaka embedding of X given by L. By hypothesis, $(\Psi_L)_*\mu$ is not supported by a hyperplane of $\mathbb{P}^N(\mathbb{C})$.

Step 1.— Suppose G acts trivially on $\operatorname{Pic}^{0}(X)$. Then L is G-invariant and there is a homomorphism $\beta: G \to \operatorname{PGL}_{N+1}(\mathbb{C})$ such that $\Psi_{L} \circ g = \beta(g) \circ \Psi_{L}$ for every $g \in L$. If G is not compact, there is a sequence of elements $g_n \in G$ going to infinity in $\operatorname{PGL}_{N+1}(\mathbb{C})$: in the KAK decomposition $g_n = k_n a_n k'_n$, the diagonal part a_n goes to ∞ . Then, any probability measure on $\mathbb{P}^N(\mathbb{C})$ which is invariant under all g_n is supported in a proper projective subspace of $\mathbb{P}^N(\mathbb{C})$, and this contradicts our preliminary remark. So, G is compact and the action is stiff.

Step 2.— Suppose the action of G on $\operatorname{Pic}(X)^0$ is non-trivial. Then, the base of the ruling $\alpha \colon X \to B$ has genus ≥ 1 , and the homomorphism $\operatorname{Aut}(X)^0 \to \operatorname{Aut}(B)^0$ has positive dimensional image. So, B is an elliptic curve on which $\operatorname{Aut}(X)^0$ acts transitively. According to [70, Thm 3] and [72, §3], there are two cases: either $X = B \times \mathbb{P}^1(\mathbb{C})$, $\operatorname{Aut}(X) = \operatorname{Aut}(B) \times \operatorname{PGL}_2(\mathbb{C})$ and we deduce, as in the first step, that G is a compact group; or $\operatorname{Aut}(X)^\circ$ is Abelian. In all cases stiffness follows, and we are done.

10.4. Invariant algebraic curves II. If Γ_{ν} admits a smooth invariant rational curve C such that the induced action on $C \simeq \mathbb{P}^1(\mathbb{C})$ comes from a non-elementary subgroup of $\mathsf{PGL}_2(\mathbb{C})$, then there is a unique, non-invariant, stationary measure on C. The next result shows that when ν is symmetric, every non-invariant stationary measure is essentially of this kind.

Proposition 10.6. Let (X, ν) be a random holomorphic dynamical system, with ν symmetric. Let μ be an ergodic ν -stationary measure giving positive mass to some proper Zariski closed subset of X. Then μ is supported on a Γ_{ν} -invariant proper Zariski closed subset and

- (a) either μ is invariant;
- (b) or the Zariski closure of Supp(µ) is a finite, disjoint union of smooth rational curves C_i, the stabilizer of C_i in Γ induces a strongly irreducible and proximal subgroup of Aut(C_i) ≃ PGL₂(**C**), and µ(C_i)⁻¹µ|_{C_i} is the unique stationary measure of this group of Möbius transformations.

Moreover, if (X, ν) is non-elementary, the curves C_i have negative self-intersection and can be contracted on cyclic quotient singularities.

Note that no moment assumption is assumed here. Before giving the proof, let us briefly discuss the question of stiffness for Möbius actions on $\mathbb{P}^1(\mathbf{C})$. Let ν be a symmetric measure on $\mathsf{PGL}_2(\mathbf{C})$. As already said, by Furstenberg's theory, if Γ_{ν} is strongly irreducible and unbounded it admits a unique stationary measure, and this measure is not invariant. Otherwise, any ν -stationary measure is invariant because

– either Γ_{ν} is relatively compact and stiffness follows from [51, Thm. 3.5];

- or Γ_{ν} admits an invariant set made of two points, then Γ_{ν} is virtually Abelian and stiffness follows from Theorem 10.1;
- or Γ_{ν} is conjugate to a subgroup of the affine group Aff(C) with no fixed point.

In the latter case after conjugating Γ_{ν} to a subgroup of Aff(**C**) we can write any $g \in \Gamma_{\nu}$ as g(z) = a(g)z + b(g). If $a(g) \equiv 1$ then Γ_{ν} is Abelian and we are done. Otherwise Γ_{ν} is merely solvable and we apply the following lemma which follows from a result of Bougerol and Picard: *let* ν *be a symmetric probability measure on* Aff(**C**). *If no point of* **C** *is fixed by* ν *-almost every g, then the only* ν *-stationary probability on* $\mathbb{P}^1(\mathbf{C})$ *is the point mass at* ∞ (see [14, Thm. 2.4]; a self-contained proof is provided in [24]).

Proof of Proposition 10.6. If μ has an atom then, by ergodicity, μ is supported on a finite orbit and it is invariant. So we now assume that μ is atomless. By ergodicity, μ gives full mass to a Γ_{ν} -invariant curve D; let C_1, \ldots, C_n be its irreducible components. Let Γ' be the finite index subgroup of Γ_{ν} stabilizing each C_i and ν' be the hitting measure induced by ν on Γ' ; it is symmetric, μ is ν' -stationary, and so are its restrictions $\mu|_{C_i}$, for each C_i .

If the genus of (the normalization of) C_1 is positive, then $\Gamma'|_{C_1} \subset \operatorname{Aut}(C_1)$ is virtually Abelian, hence $\mu|_{C_1}$ is Γ' -invariant. Since μ is ergodic, Γ_{ν} permutes transitively the C_i , and arguing as in Lemma 10.2, we see that μ is ν -invariant as well. Now, assume that the normalization \hat{C}_1 is isomorphic to $\mathbb{P}^1(\mathbb{C})$. If C_1 is not smooth, or if it intersects another Γ_{ν} -periodic curve, then the image of Γ' in $\operatorname{Aut}(\hat{C}_1) \simeq \operatorname{PGL}_2(\mathbb{C})$ is not strongly irreducible, and the discussion preceding this proof shows that μ is Γ' -invariant. Again, this implies that μ is Γ_{ν} -invariant. The same holds if Γ' is a bounded subgroup of $\operatorname{Aut}(\hat{C}_1)$. The only possibility left is that C_1 is smooth, disjoint from the other periodic curves, and Γ' induces a strongly irreducible subgroup of $\operatorname{Aut}(C_1)$. Since Γ_{ν} permutes transitively the C_i , conjugating the dynamics of the groups $\Gamma'|_{C_i}$, the same property holds for each C_i .

If Γ_{ν} is non-elementary, Lemma 2.12 shows that $C_i^2 = -m$ for some m > 0, which does not depend on *i* because Γ_{ν} permutes the C_i transitively. Then, the C_i being disjoint, one can contract them simultaneously, each of the contractions leading to a quotient singularity $(\mathbf{C}^2, 0)/\langle \eta \rangle$ with $\eta(x, y) = (\alpha x, \alpha y)$ for some root of unity α of order *m* (see [4, §III.5]).

10.5. Non-elementary groups: real dynamics. We now consider the action of general nonelementary subgroup of Aut(X) on an invariant, totally real surface Y; as in Theorem A, we further assume the existence of an invariant volume form on Y; this is automatic when X is an Abelian, K3, or Enriques surface (see Remark 3.2 and [26]).

Theorem 10.7. Let (X, ν) be a non-elementary random holomorphic dynamical system on a compact Kähler surface, satisfying the moment condition (4.1). Assume that $Y \subset X$ is a Γ_{ν} -invariant totally real 2-dimensional smooth submanifold such that the action of Γ_{ν} on Y preserves a probability measure vol_Y equivalent to the Riemannian volume on Y. Then, every ergodic stationary measure μ on Y is:

(a) either almost surely invariant,

(b) or supported on a Γ_{ν} -invariant algebraic curve.

In particular if there is no Γ_{ν} -invariant curve then (Y, ν) is stiff. Moreover, if the fiber entropy of μ is positive, then μ is the restriction of vol_Y to a subset of positive volume.

Recall from Lemma 2.12 that Γ_{ν} -invariant curves can be contracted. For the induced random dynamical system on the resulting singular surface, stiffness holds unconditionally. If furthermore ν is symmetric then the result can be made more precise by applying Proposition 10.6.

Proof of Theorem 10.7. We split the proof in two steps.

Step 1.– Let μ be an ergodic stationary measure supported on Y. We assume that μ is not invariant, and we want to prove that it is supported on a Γ_{ν} -invariant curve. Since the action is volume preserving, its Lyapunov exponents satisfy $\lambda^{-} + \lambda^{+} = 0$ (see § 7.2.3). The invariance principle (Theorem 7.5) shows that μ is hyperbolic: indeed μ is almost surely invariant when $\lambda^{-} \ge 0$. We can therefore apply Theorem 3.4 of [16] to obtain the following trichotomy:

- (1) either μ has finite support, so it is invariant;
- (2) or the distribution of Oseledets stable directions is non-random;
- (3) or μ is almost surely invariant and absolutely continuous with respect to vol_Y: even more, it is the restriction of vol_Y to a subset of positive volume.

Since μ is not invariant, we are in case (2). Theorem 9.1 then implies that μ is supported on an invariant algebraic curve. This concludes the proof of the first assertions in Theorem 10.7, including the stiffness property when Γ has no periodic curve.

Step 2.– It remains to prove the last assertion. Let then μ be an ergodic stationary measure with $h_{\mu}(X, \nu) > 0$. In the above trichotomy, (1) is now excluded. To exclude the alternative (2), by Theorem 9.1, it suffices to show that μ is not supported on an invariant curve. By Proposition 7.10 (i.e. the fibered Margulis-Ruelle inequality), μ is hyperbolic. If μ is supported on an algebraic curve, the proof of Corollary 8.3 leads to the following alternative: either μ is atomic or the Lyapunov exponent along that curve is negative. In the latter case μ is proximal along that curve and its stable conditionals are points. In both cases the fiber entropy would vanish, in contradiction with our hypothesis, so μ is not supported on an algebraic curve, as desired.

11. MEASURE RIGIDITY

Invariant measures are classified in [26] when Γ is non-elementary and contains a parabolic element. Thus, in view of the results of Section 10, it is natural to ask for such a classification when Γ does not contain parabolic elements. If μ is a probability measure on X, we denote by Aut_{μ}(X) the group of automorphisms of X preserving μ .

Theorem 11.1. Let f be an automorphism of a complex projective surface X, preserving a totally real and real analytic surface $Y \subset X$. Let μ be an ergodic f-invariant measure on Y with positive entropy. Then

- (a) either μ is absolutely continuous with respect to the Lebesgue measure on Y;
- (b) or $Aut_{\mu}(X)$ is virtually cyclic.

If in addition the Lyapunov exponents of f with respect to μ satisfy $\lambda^{-}(f,\mu) + \lambda^{+}(f,\mu) \neq 0$, then case (a) does not occur, so Aut_{μ}(X) is virtually cyclic.

This result, and its proof, may be viewed as a counterpart, in our setting, to Theorems 5.1 and 5.3 of [16]; again the possibility of invariant line fields is ruled out by using the complex structure. As before the typical case to keep in mind is when X is a projective surface defined over **R** and $Y = X(\mathbf{R})$. Observe that by ergodicity, if f preserves a smooth volume vol_Y, then in case (a) μ will be the restriction of vol_Y to an Aut_{μ}(X)-invariant Borel set of positive volume.

Remark 11.2. Theorem 11.1 holds more generally for compact Kähler surfaces (see [24]).

Proof of Theorem 11.1. Since it admits a measure of positive entropy, f is a loxodromic transformation. By the Ruelle-Margulis inequality μ is hyperbolic with respect to f and it does not charge any point, nor any piecewise smooth curve: indeed, the entropy of a homeomorphism of the circle or the interval is equal to zero.

For μ -almost every $x \in X$, the stable manifold $W^s(f, x)$ is an entire curve in X which is either transcendental or contained in a periodic rational curve (see [21, Thm. 6.2]). Since f has only finitely many invariant algebraic curves (see [21, Prop. 4.1]) and μ gives no mass to curves, $W^s(f, x)$ is μ -almost surely transcendental; then, the only Ahlfors-Nevanlinna current associated to $W^s(f, x)$ is T_f^+ ; similarly, the Ahlfors-Nevanlinna currents of the unstable manifolds give T_f^- . (This is the analogue in deterministic dynamics of Theorem 8.2.) Fix $g \in Aut_{\mu}(X)$ and set $\Gamma := \langle f, g \rangle$. Our first goal is to prove the following:

Alternative: either Γ^* is virtually cyclic and preserves $\{\mathbb{P}[T_f^+], \mathbb{P}[T_f^-]\} \subset \partial \mathbb{H}_X$; or μ is absolutely continuous with respect to the Lebesgue measure on Y.

Let $Y' \subset Y$ be the union of the connected components of Y of positive μ -measure. The measure μ does not charge any analytic subset of Y of dimension ≤ 1 ; thus, by analytic continuation, any $h \in \Gamma$ preserves Y'. So, without loss of generality we can replace Y by Y'.

We divide the argument into several cases according to the existence or non-existence of certain Γ -invariant line fields. In the first two cases we will conclude that Γ is elementary. In the third case, μ will be absolutely continuous with respect to the Lebesgue measure on Y; then by the Pesin formula its Lyapunov exponents satisfy $\lambda^+(f,\mu) = -\lambda^-(f,\mu) = h_{\mu}(f)$ so when $\lambda^+(f,\mu) + \lambda^-(f,\mu) \neq 0$, Case 3 is actually impossible.

Case 1.– There exists a Γ -invariant measurable line field. Specifically, we mean a measurable field of complex lines $x \mapsto E(x) \in \mathbb{P}(T_xX)$, defined on a set of full μ -measure, such that $D_xh(E(x)) = E(h(x))$ for every $h \in \Gamma$ and almost every $x \in X$; since μ is supported on the totally real surface Y, the field of real lines $E(x) \cap T_xY \subset T_xY$ is also invariant, and determines E(x). Now, μ being ergodic and hyperbolic for f, the Oseledets theorem shows that either $E(x) = E_f^s(x) \mu$ -almost everywhere or $E(x) = E_f^u(x) \mu$ -almost everywhere. Changing f into f^{-1} if necessary, we may assume that $E(x) = E_f^s(x)$.

Consider the automorphism $h = g^{-1}fg \in \operatorname{Aut}_{\mu}(X)$. Since h is conjugate to f, μ is also ergodic and hyperbolic for h. Thus, either $E_h^s(x) = E_f^s(x)$ for μ -almost every x or $E_h^u(x) = E_f^s(x)$ for μ -almost every x.

Lemma 11.3. If there is a measurable set A of positive measure along which $E_h^s(x) = E_f^s(x)$ (resp. $E_h^u(x) = E_f^s(x)$), then $W^s(f, x) = W^s(h, x)$ for almost every x in A (resp. $W^u(h, x) = W^s(f, x)$).

Let us postpone the proof of this lemma and conclude the argument. Suppose first that $E_h^s(x) = E_f^s(x)$ on a subset A with $\mu(A) > 0$. Then $T_f^+ = T_h^+$ because for μ -almost every x, the unique Ahlfors-Nevanlinna current associated to the (complex) stable manifold $W^s(f, x)$ (resp. $W^s(h, x)$) is T_f^+ (resp. T_h^+). Since $T_h^+ = \mathbf{M}(g^*T_f^+)^{-1}g^*T_f^+$, we see that g, and therefore Γ itself, preserve the line $\mathbf{R}[T_f^+] \subset H^{1,1}(X)$. Since $[T_f^+]^2 = 0$, Γ fixes a point $\mathbb{P}[T_f^+]$ of the boundary $\partial \mathbb{H}_X$, so it is elementary. Since in addition Γ contains a loxodromic element, Theorem 3.2 of [21] shows that Γ^* is virtually cyclic.

Now, suppose that $E_h^u(x) = E_f^s(x)$ on A. Then, $T_h^- = T_f^+$ and the group generated by f and h is elementary. Since it contains a loxodromic element [21, Thm 3.2] says that $\langle f^*, h^* \rangle$ is virtually cyclic and fixes also $\mathbb{P}[T_f^-] \in \partial \mathbb{H}_X$. This implies that g, hence Γ , preserves the pair of boundary points $\{\mathbb{P}[T_f^+], \mathbb{P}[T_f^-]\} \subset \partial \mathbb{H}_X$. Thus, in both cases Γ^* is virtually cyclic and preserves $\{\mathbb{P}[T_f^+], \mathbb{P}[T_f^-]\} \subset \partial \mathbb{H}_X.$

Proof of Lemma 11.3. The argument is similar to that of Theorem 9.1, in a simplified setting, so we only sketch it. For μ -almost every x, $W^{s}(f, x)$ and $W^{s}(h, x)$ are tangent at x. Assume by contradiction that there exists a measurable subset A' of A of positive measure such that $W^{s}(f,x) \neq W^{s}(h,x)$ for every $x \in A'$. Then for small $\varepsilon > 0$ there exists two positive constants $r = r(\varepsilon)$ and $c = c(\varepsilon)$, an integer $k \ge 2$, and a measurable subset $\mathcal{G}_{\varepsilon} \subset A'$ such that $\mu(\mathcal{G}_{\varepsilon}) > 0$ and

- $W^s_{\text{loc}}(f, x)$ and $W^s_{\text{loc}}(h, x)$ are well defined and of size r for every $x \in \mathcal{G}_{\varepsilon}$, $W^s_{\text{loc}}(f, x)$ and $W^s_{\text{loc}}(h, x)$ depend continuously on x on $\mathcal{G}_{\varepsilon} \subset X$,
- inter_x($W_{loc}^{s}(f, x), W_{loc}^{u}(f, x)$) = k for every $x \in \mathcal{G}_{\varepsilon}$,
- and $\operatorname{osc}_{(k,x,r)}(W_r^s(f,x), W_r^s(h,x)) \ge c$ for every $x \in \mathcal{G}_{\varepsilon}$.

Indeed, to get the first and second properties, one intersects A' with a large Pesin set $\mathcal{R}_{\varepsilon}$. On $A' \cap \mathcal{R}_{\varepsilon}$ the multiplicity of intersection $x \mapsto \operatorname{inter}_{x}(W^{s}_{\operatorname{loc}}(f, x), W^{u}_{\operatorname{loc}}(f, x))$ is semi-continuous, so we can find $k \ge 2$ and a subset $\mathcal{R}'_{\varepsilon} \subset (A' \cap \mathcal{R}_{\varepsilon})$ of positive measure such that

(11.1)
$$\operatorname{inter}_{x}(W^{s}_{\operatorname{loc}}(f, x), W^{u}_{\operatorname{loc}}(f, x)) = k$$

for every $x \in \mathcal{R}'_{\varepsilon}$. Thus, the k-th osculation number is well defined, and the last property holds on a subset $\mathcal{G}_{\varepsilon} \subset \mathcal{R}'_{\varepsilon}$ of positive measure if c is small.

Let η^s be a Pesin partition subordinate to the local stable manifolds of f. Since $h_{\mu}(f) > 0$ 0 the conditional measures $\mu(\cdot|\eta^s)$ are non-atomic. Thus there exists $x \in \mathcal{G}_{\varepsilon}$ such that x is an accumulation point of Supp $(\mu(\cdot|\eta^s(x))|_{\mathcal{G}_{\varepsilon}\cap\eta^s(x)})$. Fix a neighborhood N of x such that $W_r^s(f,x) \cap W_r^s(h,x) \cap N = \{x\}$, and then pick a sequence (x_i) of points in $\mathcal{G}_{\varepsilon} \cap \eta^s(x) \cap N$ converging to x. The local stable manifolds $W_r^s(h, x_j)$ form a sequence of disks of size r at x_j , each of them tangent to $W_r^s(f, x)$ (at x_j), and all of them disjoint from $W_r^s(h, x)$ (because x_j does not belong to $W_r^s(h, x)$). This contradicts Corollary 9.9, and the proof is complete.

Case 2.– There is a pair of distinct measurable line fields $\{E_1(x), E_2(x)\}$ invariant under Γ . Again by the Oseledets theorem applied to f, necessarily $\{E_1(x), E_2(x)\} = \{E_f^s(x), E_f^u(x)\}$. For μ -almost every $x, g(\{E_f^s(x), E_f^u(x)\}) = \{E_f^s(g(x)), E_f^u(g(x))\}$. As before, consider h = $g^{-1}fg \in \operatorname{Aut}_{\mu}(X)$. Since h is conjugate to f, it is hyperbolic and ergodic with respect to μ , and $\{E_f^s(x), E_f^u(x)\} = \{E_h^s(x), E_h^u(x)\}$ for almost every x. Replacing h by h^{-1} if necessary, there exists a set A of positive measure for which $E_h^s(x) = E_f^s(x)$, and we conclude as in Case 1.

Case 3.– There is no Γ -invariant line field or pair of line fields. In other words, Cases 1 or 2 are now excluded. This part of the argument is identical to the proof of [16, Thm 5.1.a].

First, we claim that there exists $g_1, g_2 \in \Gamma$ and a subset A of positive measure such that $D_x g_1(E_f^s(x)) \notin \{E_f^s(g_1(x)), E_f^u(g_1(x))\}$ and $D_x g_2(E_f^u(x)) \notin \{E_f^s(g_2(x)), E_f^u(g_2(x)\}\}$ for evaluation of the second se ery x in A. Indeed since we are not in Case 2 (possibly switching E_f^u and E_f^s) there exists $g_1 \in \Gamma$ and a set A of positive measure such that for $x \in A$, $D_x g_1(E_f^s(x)) \Leftrightarrow E_f^s(g_1(x)) \cup E_f^u(g_1(x))$. Since we are not in Case 1, there exists $g \in \Gamma$ and a set B of positive measure such that for $x \in B$, $D_x g(E_f^u(x)) \neq E_f^u(g(x))$. If $D_x g(E_f^s(x)) \in \{E_f^s(g(x)), E_f^u(g(x))\}$ on a subset B'of B of positive measure, then choose k > 0 and $\ell > 0$ such that $\mu(f^\ell(A) \cap B') > 0$ and $\mu(f^k(g(f^\ell(A))) \cap A) > 0$ and define $g_2 = g_1 f^k g f^\ell$; otherwise, set $g_2 = g f^\ell$ with ℓ such that $\mu(f^\ell(A) \cap B) > 0$. Then change A into $A = A \cap f^{-\ell}(B')$ (resp. $A \cap f^{-\ell}(B)$).

Denote by Δ the simplex $\{(a, b, c, d) \in (\mathbf{R}^*_+)^4 ; a + b + c + d = 1\}$. For $\alpha = (a, b, c, d)$ in Δ , let ν_{α} be the probability measure $\nu_{\alpha} = a\delta_f + b\delta_{f^{-1}} + c\delta_g + d\delta_{g^{-1}}$. Then μ is ν_{α} -stationary and since μ is *f*-ergodic and $\nu_{\alpha}(\{f\}) > 0$, it is also ergodic as a ν_{α} -stationary measure (see [8, §2.1.3]). Since we are not in Cases 1 or 2 and μ is hyperbolic for *f*, the invariance principle of Ledrappier [65] implies that the Lyapunov exponents of μ , viewed as a ν_{α} -stationary measure, satisfy $\lambda_{\alpha}^{-}(\mu) < \lambda_{\alpha}^{+}(\mu)$ (see Section 13.2.2 of [16]; more precise statements and proofs can be found in [23, §7]).

Lemma 11.4. There exists a choice of $\alpha \in \Delta$ such that μ is a hyperbolic ν_{α} -stationary measure, *i.e.* $\lambda_{\alpha}^{-}(\mu) < 0 < \lambda_{\alpha}^{+}(\mu)$

Proof. This is automatic when f and g are volume preserving because $\lambda_{\alpha}^{-}(\mu) = -\lambda_{\alpha}^{+}(\mu)$ in that case. For completeness, let us copy the proof given in [16, §13.2.4]. The assumptions of Case 3 and the strict inequality $\lambda^{-}(\mu) < \lambda^{+}(\mu)$ imply that

(11.2)
$$\alpha \in \Delta \mapsto (\lambda_{\alpha}^{-}(\mu), \lambda_{\alpha}^{+}(\mu)) \in \mathbf{R}^{2}$$

is continuous (see [16, Prop. 13.7] or [77, Chap. 9]). Since $\lambda_{\alpha}^{-}(\mu) < \lambda_{\alpha}^{+}(\mu)$ for every $\alpha \in \Delta$, one of λ_{α}^{-} and λ_{α}^{+} is non zero. Furthermore, μ being invariant, the involution $(a, b, c, d) \mapsto$ (b, a, d, c) interchanges the Lyapunov exponents. It follows that $P = \{\alpha \in \Delta, \lambda_{\alpha}^{+} > 0\}$ and N = $\{\alpha \in \Delta, \lambda_{\alpha}^{-} < 0\}$ are non-empty open subsets of Δ such that $P \cup N = \Delta$. The connectedness of Δ implies $P \cap N \neq \emptyset$, as was to be shown. \Box

Fix $\alpha \in \Delta$ such that μ is hyperbolic as a ν_{α} -stationary measure. The assumptions of Case 3 imply that the stable directions depend on the itinerary so the main result of [16] shows that μ is fiberwise SRB (on the surface Y), that is, the unstable conditionals of the measures μ_x (here $\mu_x = \mu$) are given by the Lebesgue measure (in some natural affine parametrizations of the unstable manifolds by the real line **R**). Since μ is invariant, we can revert the stable and unstable directions by applying the argument to F^{-1} , and we conclude that the stable conditionals are given by the Lebesgue measure as well. The absolute continuity property of the stable and unstable laminations then implies that μ is absolutely continuous with respect to the Lebesgue measure on Y.

Conclusion.– Assume that μ is not absolutely continuous with respect to the Lebesgue measure on Y. The above alternative holds for all subgroups $\Gamma = \langle f, g \rangle$, with $g \in \operatorname{Aut}_{\mu}(X)$ arbitrary. Therefore, $\operatorname{Aut}_{\mu}(X)^*$ preserves $\{\mathbb{P}[T_f^+], \mathbb{P}[T_f^-]\} \subset \partial \mathbb{H}_X$, which implies that $\operatorname{Aut}_{\mu}(X)^*$ is virtually cyclic. It remains to prove that $\operatorname{Aut}_{\mu}(X)$ itself is virtually cyclic. If not, then $\operatorname{Aut}(X)^\circ$ is infinite, X is a torus \mathbb{C}^2/Λ (see Proposition 3.3), and $\operatorname{Aut}_{\mu}(X) \cap \operatorname{Aut}(X)^\circ$ is a normal subgroup of $\operatorname{Aut}_{\mu}(X)$ containing infinitely many translations. This group is a closed subgroup of the compact Lie group $\operatorname{Aut}(X)^\circ = \mathbb{C}^2/\Lambda$; thus, its connected component of the identity is a (real) torus $H \subset \mathbb{C}^2/\Lambda$ of positive dimension. This torus H is invariant under the action of f by conjugacy. Since $X = \mathbb{C}^2/\Lambda$, f is a complex linear Anosov diffeomorphism of X, and it follows that $\dim_{\mathbb{R}}(H) \ge 2$. Being H-invariant, μ is then absolutely continuous with respect to the Lebesgue measure of Y; this contradiction completes the proof. It is natural to expect that the positive entropy assumption in Theorem 11.1 could be replaced by a much weaker assumption, namely, " μ gives no mass to proper Zariski closed subsets". The case when $\text{Supp}(\nu)$ contains a Kummer example is successfully treated in [24]. Also, a version of Theorem 11.1 can be established for polynomial automorphisms of \mathbb{R}^2 , with essentially the same proof (see [24]).

APPENDIX A. STRONG LAMINARITY OF AHLFORS CURRENTS

In this appendix, we sketch the proof of Lemma 8.8, by explaining how to adapt the arguments of [5, 40, 41], written for $X = \mathbb{P}^2(\mathbf{C})$, to our context.

Proof of Lemma 8.8. Let (Δ_n) be a sequence of unions of disks, as in the definition of injective Ahlfors currents, such that $\frac{1}{\mathbf{M}(\Delta_n)} \{\Delta_n\}$ converges to T. Since X is projective we can choose a finite family of meromorphic fibrations $\varpi_i : X \dashrightarrow \mathbb{P}^1$ such that

- the general fibers of ϖ_i are smooth curves of genus ≥ 2 ;
- for every x ∈ X, there are at least two of the fibrations \(\varpi_i\), denoted for simplicity by \(\varpi_1\) and \(\varpi_2\), which are well defined in some neighborhood \(U_x\) of x (x is not a base point of the corresponding pencils), satisfy \((d\varpi_1 \wedge d\varpi_2)(x) \neq 0\) (the fibrations are transverse), and for which the fibers \(\varpi_k^{-1}(\varpi_k(x))\) containing x are smooth.

If we blow-up the base points of ϖ_k , k = 1, 2, we obtain a new surface $X' \to X$ on which each ϖ_k lifts to a regular fibration ϖ'_k ; the open neighborhood U_x is isomorphic to its preimage in X' so, when working on U_x , we can do as if the two fibrations ϖ_k were local submersions with smooth fibers of genus ≥ 2 .

To construct T_r , we follow the proof of [41, Proposition 4.4] (see also [40, Proposition 3.4]). The construction works as follows: we fix a sequence (r_j) converging to zero, and for every j we extract from $\frac{1}{\mathbf{M}(\Delta_n)} \{\Delta_n\}$ a current T_{n,r_j} made of disks of size $\approx r_j$ which are obtained from Δ_n by only keeping graphs of size r_j over one of the projections ϖ_i .

By a covering argument, it is enough to work locally near a point x, with two projections ϖ_1 and ϖ_2 as above. Let $S \subset \mathbf{C}$ be the unit square $\{x + iy ; 0 \le x \le 1, 0 \le y \le 1\} \simeq [0, 1]^2$. To simplify the exposition, we may assume that

(A.1)
$$\varpi_k(U_x) = S \subset \mathbf{C} \subset \mathbb{P}^1(\mathbf{C})$$
 (for $k = 1, 2$).

Set $r_j = 2^{-j}$ and consider the subdivision \mathcal{Q}_j of $S \simeq [0,1]^2$ into 4^j squares Q of size r_j . A connected component of $\Delta_n \cap \varpi_k^{-1}(Q)$, for such a small square Q, is called a graph (with respect to ϖ_k) if it lifts to a local section of the fibration $\varpi'_k \colon X' \to \mathbb{P}^1(\mathbb{C})$ above Q. Then, we fix j, intersect Δ_n with $\varpi_k^{-1}(Q)$, and keep only the components of $\varpi_k^{-1}(Q \cap \Delta_n), Q \in \mathcal{Q}_j$ which are graphs with respect to ϖ_k . Such a family of graphs is normal because the fibers of ϖ'_k have genus ≥ 2 (compare to Lemma 3.5 of [40]).

This being done, we can copy the proof of [41, Proposition 4.4]. Letting n go to $+\infty$ and extracting a converging subsequence, we obtain a uniformly laminar current $T_{Q_{j},k} \leq T$. Away from the base points of ϖ_k , $T_{Q_{j},k}$ is made of disks of size $\approx r_j$ which are limits of disks contained in the Δ_n . Combining the two currents $T_{Q_{j},k}$, we get a current $T_{r_j} \leq T$ which is uniformly laminar in every cube $\varpi_1^{-1}(Q) \cap \varpi_2^{-1}(Q'), Q, Q' \in Q_j$, and such that

(A.2)
$$\langle T - T_{r_j}, \varpi_1^* \kappa_{\mathbb{P}^1} + \varpi_1^* \kappa_{\mathbb{P}^1} \rangle \leq \langle T - T_{\mathcal{Q}_j,1}, \varpi_1^* \kappa_{\mathbb{P}^1} \rangle + \langle T - T_{\mathcal{Q}_j,2}, \varpi_2^* \kappa_{\mathbb{P}^1} \rangle,$$

where $\kappa_{\mathbb{P}^1}$ is the Fubini-Study form. By definition, T will be strongly approximable if locally $\mathbf{M}(T - T_{r_j}) \leq O(r_j^2)$. Using the fact that $\varpi_1^* \kappa_{\mathbb{P}^1} + \varpi_1^* \kappa_{\mathbb{P}^1} \geq C \kappa_0$ and the Inequality (A.2), it will be enough to show that $\langle T - T_{\mathcal{Q}_j,k}, \varpi_k^* \kappa_{\mathbb{P}^1} \rangle = O(r_j^2)$ for k = 1, 2. This itself reduces to counting (with multiplicity) the number of "good components" of Δ_n for the projections $\varpi_k : \Delta_n \to \mathcal{Q}_j$ that is, the components above the squares Q of Q_j that are kept in the above contruction of $T_{\mathcal{Q}_j,k}$ (the graphs relative to ϖ_k).

The counting argument is identical to [5, §7], except that we apply the Ahlfors theory of covering surfaces to a union of disks, not just one. For notational ease, set $\varpi = \varpi_k$, $r = r_j$ and $\mathcal{Q} = \mathcal{Q}_j$; \mathcal{Q} is a subdivision of $S \simeq [0, 1]^2$ by squares of size 2^{-j} . We decompose \mathcal{Q} as a union of four non-overlapping subdivisions \mathcal{Q}^{ℓ} , $\ell = 1, 2, 3, 4$; by this we mean that for each ℓ , the squares $Q \in \mathcal{Q}^{\ell}$ have disjoint closures \overline{Q} . Fix such an ℓ and let $q = \#\mathcal{Q}^{\ell} = 4^{j-1}$. Applying Ahlfors' theorem to each of the disks constituting Δ_n and summing over these disks, we deduce that the number of good components $N(\mathcal{Q}^{\ell})$ satisfies (⁵)

(A.3)
$$N(\mathcal{Q}^{\ell}) \ge (q-4) \operatorname{area}_{\mathbb{P}^1}(\Delta_n) - h \operatorname{length}_{\mathbb{P}^1}(\partial \Delta_n),$$

where $\operatorname{area}_{\mathbb{P}^1}$ (resp. $\operatorname{length}_{\mathbb{P}^1}$) is the area of the projection $\varpi(\Delta_n)$ (resp. length of $\varpi(\partial\Delta_n)$), counted with multiplicity, and h is a constant that depends only on the geometry of \mathcal{Q}^{ℓ} . Dividing by $\operatorname{area}_{\mathbb{P}^1}(\Delta_n)$, using $\operatorname{length}_{\mathbb{P}^1}(\partial\Delta_n) = o(\operatorname{area}_{\mathbb{P}^1}(\Delta_n))$, which is guaranteed by Ahlfors' construction, and letting n go to $+\infty$, we obtain

(A.4)
$$\langle T_{\mathcal{Q}}|_{\mathcal{Q}^{\ell}}, \varpi^* \kappa_{\mathbb{P}^1} \rangle \ge (q-4)r^2 = \operatorname{area}_{\mathbb{P}^1} \left(\bigcup_{S \in \mathcal{Q}^{\ell}} S \right) - 4r^2.$$

Finally, summing from $\ell = 1$ to 4, we see that, relative to $\varpi^* \kappa_{\mathbb{P}^1}$, the mass lost by discarding the bad components of size r in T is of order $O(r^2)$: this is precisely the required estimate.

Let us now justify the geometric intersection statement, following step by step the proof of [41, Thm. 4.2]: let S be a current with continuous normalized potential on X; we have to show that $S \wedge T_r$ increases to $S \wedge T$ as r decreases to 0. Again the result is local so we work near x, use the projections ϖ_1 and ϖ_2 , and keep notation as above. Given squares $Q, Q' \in Q$ and a real number $\lambda < 1$, we denote by λQ the homothetic of Q of factor λ with respect to its center, and by C(Q, Q') the cube $\varpi_1^{-1}(Q) \cap \varpi_2^{-1}(Q')$. Fix $\varepsilon > 0$. We want to show that for $r \leq r(\varepsilon)$, the mass of $(T - T_r) \wedge S$ is smaller than ε . The first observation is that there exists $\lambda(\varepsilon) \in (0, 1)$, independent of r, such that translating Q if necessary, the mass of $T \wedge S$ concentrated in $\bigcup_{Q,Q'} C(Q, Q') \setminus C(\lambda Q, \lambda Q')$ is smaller than $\varepsilon/2$ (see [41, Lem. 4.5]). Fix such a λ . It only remains to estimate the mass of $(T - T_r) \wedge S$ in $\bigcup_{Q,Q'} C(\lambda Q, \lambda Q')$. In such a cube $C(\lambda Q, \lambda Q')$ the argument presented in [41, pp. 123-124], based on an integration by parts, gives the estimate

(A.5)
$$\int_{C(\lambda Q, \lambda Q')} (T - T_r) \wedge S \leq C(\lambda) \operatorname{modc}(u_S, r) \frac{1}{r^2} \mathbf{M} \left((T - T_r)|_{C(Q, Q')} \right),$$

where $modc(u_S, r)$ is the modulus of continuity of the potential u_S of S. To conclude, we sum over all squares Q, Q' and use the estimate $M(T - T_r) = O(r^2)$ to get that

(A.6)
$$\mathbf{M}\left((T-T_r)\big|_{\bigcup_{Q,Q'}C(\lambda Q,\lambda Q')}\right) \leq C\omega(u_S,r).$$

This is smaller than $\varepsilon/2$ if $r \leq r(\varepsilon)$.

⁵The term (q-4) instead of (q-2) in [5] is due to the fact that we are projecting on \mathbb{P}^1 and not on **C**.

SERGE CANTAT AND ROMAIN DUJARDIN

REFERENCES

- Artur Avila and Marcelo Viana. Extremal Lyapunov exponents: an invariance principle and applications. *Invent.* Math., 181(1):115–189, 2010.
- [2] Jörg Bahnmüller and Thomas Bogenschütz. A Margulis-Ruelle inequality for random dynamical systems. *Arch. Math.* (*Basel*), 64(3):246–253, 1995.
- [3] Jörg Bahnmüller and Pei-Dong Liu. Characterization of measures satisfying the Pesin entropy formula for random dynamical systems. *J. Dynam. Differential Equations*, 10(3):425–448, 1998.
- [4] Wolf P. Barth, Klaus Hulek, Chris A. M. Peters, and Antonius Van de Ven. Compact complex surfaces, volume 4 of Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys in Mathematics [Results in Mathematics and Related Areas. 3rd Series. A Series of Modern Surveys in Mathematics]. Springer-Verlag, Berlin, second edition, 2004.
- [5] Eric Bedford, Mikhail Lyubich, and John Smillie. Polynomial diffeomorphisms of C². IV. The measure of maximal entropy and laminar currents. *Invent. Math.*, 112(1):77–125, 1993.
- [6] Yves Benoist and Pierre de la Harpe. Adhérence de Zariski des groupes de Coxeter. *Compos. Math.*, 140(5):1357–1366, 2004.
- [7] Yves Benoist and Jean-François Quint. Mesures stationnaires et fermés invariants des espaces homogènes. *Ann.* of Math. (2), 174(2):1111–1162, 2011.
- [8] Yves Benoist and Jean-François Quint. Random walks on reductive groups, volume 62 of Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys in Mathematics [Results in Mathematics and Related Areas. 3rd Series. A Series of Modern Surveys in Mathematics]. Springer, Cham, 2016.
- [9] V. I. Bogachev. Measure theory. Vol. I, II. Springer-Verlag, Berlin, 2007.
- [10] Thomas Bogenschütz and Hans Crauel. The Abramov-Rokhlin formula. In *Ergodic theory and related topics, III (Güstrow, 1990)*, volume 1514 of *Lecture Notes in Math.*, pages 32–35. Springer, Berlin, 1992.
- [11] Jean-Benoit. Bost, Henri Gillet, and Christophe Soulé. Heights of projective varieties and positive Green forms. J. Amer. Math. Soc., 7(4):903–1027, 1994.
- [12] Sébastien Boucksom. Divisorial Zariski decompositions on compact complex manifolds. Ann. Sci. École Norm. Sup. (4), 37(1):45–76, 2004.
- [13] Philippe Bougerol and Jean Lacroix. *Products of random matrices with applications to Schrödinger operators*, volume 8 of *Progress in Probability and Statistics*. Birkhäuser Boston, Inc., Boston, MA, 1985.
- [14] Philippe Bougerol and Nico Picard. Strict stationarity of generalized autoregressive processes. Ann. Probab., 20(4):1714–1730, 1992.
- [15] Jean Bourgain, Alex Furman, Elon Lindenstrauss, and Shahar Mozes. Stationary measures and equidistribution for orbits of nonabelian semigroups on the torus. J. Amer. Math. Soc., 24(1):231–280, 2011.
- [16] Aaron Brown and Federico Rodriguez Hertz. Measure rigidity for random dynamics on surfaces and related skew products. J. Amer. Math. Soc., 30(4):1055–1132, 2017.
- [17] Marco Brunella. Courbes entières et feuilletages holomorphes. Enseign. Math. (2), 45(1-2):195–216, 1999.
- [18] Serge Cantat. Dynamique des automorphismes des surfaces K3. Acta Math., 187(1):1–57, 2001.
- [19] Serge Cantat. Sur la dynamique du groupe d'automorphismes des surfaces K3. Transform. Groups, 6(3):201– 214, 2001.
- [20] Serge Cantat. Bers and Hénon, Painlevé and Schrödinger. Duke Math. J., 149(3):411-460, 2009.
- [21] Serge Cantat. Dynamics of automorphisms of compact complex surfaces. In *Frontiers in complex dynamics*, volume 51 of *Princeton Math. Ser.*, pages 463–514. Princeton Univ. Press, Princeton, NJ, 2014.
- [22] Serge Cantat. Automorphisms and dynamics: a list of open problems. In Proceedings of the International Congress of Mathematicians—Rio de Janeiro 2018. Vol. II. Invited lectures, pages 619–634. World Sci. Publ., Hackensack, NJ, 2018.
- [23] Serge Cantat and Romain Dujardin. Hyperbolicity for large automorphism groups of projective surfaces. arXiv:2211.02418.
- [24] Serge Cantat and Romain Dujardin. Random dynamics on real and complex projective surfaces. Long version of this paper.
- [25] Serge Cantat and Romain Dujardin. Finite orbits for large groups of automorphisms of projective surfaces. arxiv:2012.01762, 2020.
- [26] Serge Cantat and Romain Dujardin. Invariant measures for large automorphism groups of projective surfaces. *Transform. Groups*, to appear, 2023.

- [27] Serge Cantat and Romain Dujardin. Dynamics of automorphism groups of compact complex surfaces: examples, classification, and outlook. to appear.
- [28] Serge Cantat and Christophe Dupont. Automorphisms of surfaces: Kummer rigidity and measure of maximal entropy. J. Eur. Math. Soc. (JEMS), 22(4):1289–1351, 2020.
- [29] Serge Cantat and Charles Favre. Symétries birationnelles des surfaces feuilletées. J. Reine Angew. Math., 561:199–235, 2003.
- [30] Serge Cantat, Ziyang Gao, Philipp Habegger, and Junyi Xie. The geometric Bogomolov conjecture. Duke Math. J., 170(2):247–277, 2021.
- [31] Serge Cantat, Vincent Guirardel, and Anne Lonjou. Elements generating a proper normal subgroup of the Cremona group. Int. Math. Res. Not. IMRN, (10):7339–7371, 2021.
- [32] Serge Cantat and Keiji Oguiso. Birational automorphism groups and the movable cone theorem for Calabi-Yau manifolds of Wehler type via universal Coxeter groups. Amer. J. Math., 137(4):1013–1044, 2015.
- [33] Serge Cantat and Olga Paris-Romaskevich. Automorphisms of compact Kähler manifolds with slow dynamics. *Trans. Amer. Math. Soc.*, 374(2):1351–1389, 2021.
- [34] E. M. Chirka. Complex analytic sets, volume 46 of Mathematics and its Applications (Soviet Series). Kluwer Academic Publishers Group, Dordrecht, 1989. Translated from the Russian by R. A. M. Hoksbergen.
- [35] K. L. Chung and W. H. J. Fuchs. On the distribution of values of sums of random variables. *Mem. Amer. Math. Soc.*, 6:12, 1951.
- [36] Ping Ngai (Brian) Chung. Stationary measures and orbit closures of uniformly expanding random dynamical systems on surfaces. Preprint arxiv 2006.03166, 2020.
- [37] Hans Crauel. Non-Markovian invariant measures are hyperbolic. Stochastic Process. Appl., 45(1):13–28, 1993.
- [38] Jean-Pierre Demailly. Analytic methods in algebraic geometry, volume 1 of Surveys of Modern Mathematics. International Press, Somerville, MA; Higher Education Press, Beijing, 2012.
- [39] Tien-Cuong Dinh and Nessim Sibony. Green currents for holomorphic automorphisms of compact Kähler manifolds. J. Amer. Math. Soc., 18(2):291–312, 2005.
- [40] Romain Dujardin. Laminar currents in \mathbb{P}^2 . *Math. Ann.*, 325(4):745–765, 2003.
- [41] Romain Dujardin. Sur l'intersection des courants laminaires. Publ. Mat., 48(1):107-125, 2004.
- [42] Romain Dujardin. Fatou directions along the Julia set for endomorphisms of \mathbb{CP}^k . J. Math. Pures Appl. (9), 98(6):591–615, 2012.
- [43] Rick Durrett. Probability—theory and examples, volume 49 of Cambridge Series in Statistical and Probabilistic Mathematics. Cambridge University Press, Cambridge, 2019. Fifth edition of [MR1068527].
- [44] Julien Duval. Singularités des courants d'Ahlfors. Ann. Sci. École Norm. Sup. (4), 39(3):527–533, 2006.
- [45] Alex Eskin and Maryam Mirzakhani. Invariant and stationary measures for the $SL(2, \mathbb{R})$ action on moduli space. *Publ. Math. Inst. Hautes Études Sci.*, 127:95–324, 2018.
- [46] Charles Favre. Le groupe de Cremona et ses sous-groupes de type fini. Number 332, pages Exp. No. 998, vii, 11–43. 2010. Séminaire Bourbaki. Volume 2008/2009. Exposés 997–1011.
- [47] Simion Filip and Valentino Tosatti. Canonical currents and heights for k3 surfaces. Preprint arxiv 2103.02095, 2021.
- [48] Simion Filip and Valentino Tosatti. Kummer rigidity for K3 surface automorphisms via ricci flat metrics. *Camb. J. Math.*, to appear.
- [49] Jacques Franchi and Yves Le Jan. *Hyperbolic dynamics and Brownian motion*. Oxford Mathematical Monographs. Oxford University Press, Oxford, 2012.
- [50] Harry Furstenberg. Noncommuting random products. Trans. Amer. Math. Soc., 108:377–428, 1963.
- [51] Hillel Furstenberg. Stiffness of group actions. In *Lie groups and ergodic theory (Mumbai, 1996)*, volume 14 of *Tata Inst. Fund. Res. Stud. Math.*, pages 105–117. Tata Inst. Fund. Res., Bombay, 1998.
- [52] William M. Goldman. The modular group action on real SL(2)-characters of a one-holed torus. *Geom. Topol.*, 7:443–486, 2003.
- [53] Sébastien Gouëzel and Anders Karlsson. Subadditive and multiplicative ergodic theorems. J. Eur. Math. Soc. (JEMS), 22(6):1893–1915, 2020.
- [54] Phillip Griffiths and Joseph Harris. *Principles of algebraic geometry*. Wiley-Interscience [John Wiley & Sons], New York, 1978. Pure and Applied Mathematics.
- [55] Vincent Guedj. Decay of volumes under iteration of meromorphic mappings. Ann. Inst. Fourier (Grenoble), 54(7):2369–2386 (2005), 2004.

SERGE CANTAT AND ROMAIN DUJARDIN

- [56] Vincent Guedj and Ahmed Zeriahi. *Degenerate complex Monge-Ampère equations*, volume 26 of *EMS Tracts in Mathematics*. European Mathematical Society (EMS), Zürich, 2017.
- [57] Yves Guivarc'h. Croissance polynomiale et périodes des fonctions harmoniques. *Bull. Soc. Math. France*, 101:333–379, 1973.
- [58] Michael Kapovich. *Hyperbolic manifolds and discrete groups*, volume 183 of *Progress in Mathematics*. Birkhäuser Boston, Inc., Boston, MA, 2001.
- [59] Anatole Katok and Boris Hasselblatt. Introduction to the modern theory of dynamical systems, volume 54 of Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, 1995. With a supplementary chapter by Katok and Leonardo Mendoza.
- [60] Shu Kawaguchi. Projective surface automorphisms of positive topological entropy from an arithmetic viewpoint. Amer. J. Math., 130(1):159–186, 2008.
- [61] Yuri Kifer. *Ergodic theory of random transformations*, volume 10 of *Progress in Probability and Statistics*. Birkhäuser Boston, Inc., Boston, MA, 1986.
- [62] Robert Lazarsfeld. Positivity in algebraic geometry. I, volume 48 of Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys in Mathematics [Results in Mathematics and Related Areas. 3rd Series. A Series of Modern Surveys in Mathematics]. Springer-Verlag, Berlin, 2004. Classical setting: line bundles and linear series.
- [63] Yves Le Jan. Équilibre statistique pour les produits de difféomorphismes aléatoires indépendants. Ann. Inst. H. Poincaré Probab. Statist., 23(1):111–120, 1987.
- [64] F. Ledrappier. Quelques propriétés des exposants caractéristiques. In École d'été de probabilités de Saint-Flour, XII—1982, volume 1097 of Lecture Notes in Math., pages 305–396. Springer, Berlin, 1984.
- [65] F. Ledrappier. Positivity of the exponent for stationary sequences of matrices. In Lyapunov exponents (Bremen, 1984), volume 1186 of Lecture Notes in Math., pages 56–73. Springer, Berlin, 1986.
- [66] François Ledrappier and Jean-Marie Strelcyn. A proof of the estimation from below in Pesin's entropy formula. Ergodic Theory Dynam. Systems, 2(2):203–219 (1983), 1982.
- [67] François Ledrappier and Lai-Sang Young. The metric entropy of diffeomorphisms. I. Characterization of measures satisfying Pesin's entropy formula. Ann. of Math. (2), 122(3):509–539, 1985.
- [68] David I. Lieberman. Compactness of the Chow scheme: applications to automorphisms and deformations of Kähler manifolds. In *Fonctions de plusieurs variables complexes, III (Sém. François Norguet, 1975–1977)*, volume 670 of *Lecture Notes in Math.*, pages 140–186. Springer, Berlin, 1978.
- [69] Pei-Dong Liu and Min Qian. Smooth ergodic theory of random dynamical systems, volume 1606 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 1995.
- [70] Masaki Maruyama. On automorphism groups of ruled surfaces. J. Math. Kyoto Univ., 11:89–112, 1971.
- [71] Curtis T. McMullen. Dynamics on K3 surfaces: Salem numbers and Siegel disks. J. Reine Angew. Math., 545:201–233, 2002.
- [72] Joseph Potters. On almost homogeneous compact complex analytic surfaces. Invent. Math., 8:244–266, 1969.
- [73] John G. Ratcliffe. *Foundations of hyperbolic manifolds*, volume 149 of *Graduate Texts in Mathematics*. Springer, New York, second edition, 2006.
- [74] Albert Raugi. A general Choquet-Deny theorem for nilpotent groups. Ann. Inst. H. Poincaré Probab. Statist., 40(6):677–683, 2004.
- [75] Julio Rebelo and Roland Roeder. Dynamics of groups of birational automorphisms of cubic surfaces and Fatou/Julia decomposition for Painlevé 6. arXiv:2104.09256.
- [76] Paul Reschke and Bar Roytman. Lower semi-continuity of entropy in a family of K3 surface automorphisms. *Rocky Mountain J. Math.*, 47(7):2323–2349, 2017.
- [77] Marcelo Viana. *Lectures on Lyapunov exponents*, volume 145 of *Cambridge Studies in Advanced Mathematics*. Cambridge University Press, Cambridge, 2014.
- [78] Marcelo Viana and Jiagang Yang. Measure-theoretical properties of center foliations. In *Modern theory of dynamical systems*, volume 692 of *Contemp. Math.*, pages 291–320. Amer. Math. Soc., Providence, RI, 2017.
- [79] Lan Wang. Rational points and canonical heights on K3-surfaces in $\mathbf{P}^1 \times \mathbf{P}^1 \times \mathbf{P}^1$. In *Recent developments in the inverse Galois problem (Seattle, WA, 1993)*, volume 186 of *Contemp. Math.*, pages 273–289. Amer. Math. Soc., Providence, RI, 1995.
- [80] Joachim Wehler. K3-surfaces with Picard number 2. Arch. Math. (Basel), 50(1):73–82, 1988.

RANDOM DYNAMICS ON COMPLEX SURFACES

Serge Cantat, IRMAR, campus de Beaulieu, bâtiments 22-23 263 avenue du Général Leclerc, CS 74205 35042 RENNES Cédex

Email address: serge.cantat@univ-rennes1.fr

Sorbonne Université, CNRS, Laboratoire de Probabilités, Statistique et Modélisation (LPSM), F-75005 Paris, France

Email address: romain.dujardin@sorbonne-universite.fr